SUBCHAPTER B—PROCEDURAL RULES

PART 11—GENERAL RULEMAKING **PROCEDURES**

Subpart A—Rulemaking Procedures

Sec.

11.1 To what does this part apply?

DEFINITION OF TERMS

- 11.3 What is an advance notice of proposed rulemaking?
- 11.5 What is a notice of proposed rulemaking?
- 11.7 What is a supplemental notice of proposed rulemaking?
- What is a final rule?
- 11.11 What is a final rule with request for comments?
- 11.13 What is a direct final rule?
- 11.15 What is a petition for exemption?
- What is a petition for rulemaking? 11.17
- What is a special condition?

GENERAL

- 11.21 What are the most common kinds of rulemaking actions for which FAA follows the Administrative Procedure Act?
- 11.23 Does FAA follow the same procedures in issuing all types of rules?
- 11.25 How does FAA issue rules?
- 11.27 Are there other ways FAA collects specific rulemaking recommendations before we issue an NPRM?
- 11.29 May FAA change its regulations without first issuing an ANPRM or NPRM?
- 11.31 How does FAA process direct final rules?
- 11.33 How can I track FAA's rulemaking activities?
- 11.35 Does FAA include sensitive security information and proprietary information in the Federal Docket Management System (FDMS)?
- 11.37 Where can I find information about an Airworthiness Directive, an airspace designation, or a petition handled in a re-
- 11.38 What public comment procedures does FAA follow for Special Conditions?
- 11.39 How may I participate in FAA's rulemaking process?
- 11.40 Can I get more information about a rulemaking?

WRITTEN COMMENTS

- 11.41 Who may file comments?
- 11.43 What information must I put in my written comments?
- 11.45 Where and when do I file my comments?
- 11.47 May I ask for more time to file my comments?

PUBLIC MEETINGS AND OTHER PROCEEDINGS

- 11.51 May I request that FAA hold a public meeting on a rulemaking action?
- 11.53 What takes place at a public meeting?

PETITIONS FOR RULEMAKING AND FOR EXEMPTIONS

- 11.61 May I ask FAA to adopt, amend, or repeal a regulation, or grant relief from the requirements of a current regula-
- 11.63 How and to whom do I submit my petition for rulemaking or petition for exemption?
- 11.71 What information must I include in my petition for rulemaking?
- 11.73 How does FAA process petitions for rulemaking?
- 11.75 Does FAA invite public comment on petitions for rulemaking?
- 11.77 Is there any additional information I must include in my petition for designating airspace?
- 11.81 What information must I include in my petition for an exemption?
- 11.83 How can I operate under an exemption outside the United States?
- 11.85 Does FAA invite public comment on petitions for exemption?
- 11.87 Are there circumstances in which FAA may decide not to publish a summary of my petition for exemption?
- 11.89 How much time do I have to submit comments to FAA on a petition for exemption?
- 11.91 How does FAA inform me of its decision on my petition for exemption?
- 11.101 May I ask FAA to reconsider my petition for rulemaking or petition for exemption if it is denied?
- 11.103 What exemption relief may be available to federal, state, and local governments when operating aircraft that are not public aircraft?

Subpart B—Paperwork Reduction Act **Control Numbers**

- 11.201 Office of Management and Budget (OMB) control numbers assigned under the Paperwork Reduction Act.
- APPENDIX 1 TO PART 11—ORAL COMMUNICA-TIONS WITH THE PUBLIC DURING RULE-MAKING
- AUTHORITY: 49 U.S.C. 106(f), 106(g), 40101, 40103, 40105, 40109, 40113, 44110, 44502, 44701-44702, 44711, and 46102.
- Source: Docket No. FAA-1999-6622, 65 FR 50863, Aug. 21, 2000, unless otherwise noted.
- EDITORIAL NOTE: Nomenclature changes to part 11 appear at 61 FR 18052, April 24, 1996.

§ 11.1

Subpart A—Rulemaking Procedures

§11.1 To what does this part apply?

This part applies to the issuance, amendment, and repeal of any regulation for which FAA ("we") follows public rulemaking procedures under the Administrative Procedure Act ("APA") (5 U.S.C. 553).

DEFINITION OF TERMS

§ 11.3 What is an advance notice of proposed rulemaking?

An advance notice of proposed rule-making (ANPRM) tells the public that FAA is considering an area for rule-making and requests written comments on the appropriate scope of the rulemaking or on specific topics. An advance notice of proposed rulemaking may or may not include the text of potential changes to a regulation.

§11.5 What is a notice of proposed rulemaking?

A notice of proposed rulemaking (NPRM) proposes FAA's specific regulatory changes for public comment and contains supporting information. It includes proposed regulatory text.

§ 11.7 What is a supplemental notice of proposed rulemaking?

On occasion, FAA may decide that it needs more information on an issue, or that we should take a different approach than we proposed. Also, we may want to follow a commenter's suggestion that goes beyond the scope of the original proposed rule. In these cases, FAA may issue a supplemental notice of proposed rulemaking (SNPRM) to give the public an opportunity to comment further or to give us more information.

§ 11.9 What is a final rule?

A final rule sets out new or revised requirements and their effective date. It also may remove requirements. When preceded by an NPRM, a final rule will also identify significant substantive issues raised by commenters in response to the NPRM and will give the agency's response.

§ 11.11 What is a final rule with request for comments?

A final rule with request for comment is a rule that the FAA issues in final (with an effective date) that invites public comment on the rule. We usually do this when we have not first issued an ANPRM or NPRM, because we have found that doing so would be impracticable, unnecessary, or contrary to the public interest. We give our reasons for our determination in the preamble. The comment period often ends after the effective date of the rule. A final rule not preceded by an ANPRM or NPRM is commonly called an "immediately adopted final rule." We invite comments on these rules only if we think that we will receive useful information. For example, we would not invite comments when we are just making an editorial clarification or correction.

§11.13 What is a direct final rule?

A direct final rule is a type of final rule with request for comments. Our reason for issuing a direct final rule without an NPRM is that we would not expect to receive any adverse comments, and so an NPRM is unnecessary. However, to be certain that we are correct, we set the comment period to end before the effective date. If we receive an adverse comment or notice of intent to file an adverse comment, we then withdraw the final rule before it becomes effective and may issue an NPRM.

§ 11.15 What is a petition for exemption?

A petition for exemption is a request to FAA by an individual or entity asking for relief from the requirements of a current regulation.

§11.17 What is a petition for rule-making?

A petition for rulemaking is a request to FAA by an individual or entity asking the FAA to adopt, amend, or repeal a regulation.

§11.19 What is a special condition?

A special condition is a regulation that applies to a particular aircraft design. The FAA issues special conditions when we find that the airworthiness

regulations for an aircraft, aircraft engine, or propeller design do not contain adequate or appropriate safety standards, because of a novel or unusual design feature.

GENERAL

§ 11.21 What are the most common kinds of rulemaking actions for which FAA follows the Administrative Procedure Act?

FAA follows the Administrative Procedure Act (APA) procedures for these common types of rules:

- (a) Rules found in the Code of Federal Regulations;
- (b) Airworthiness directives issued under part 39 of this chapter; and
- (c) Airspace Designations issued under various parts of this chapter.

§11.23 Does FAA follow the same procedures in issuing all types of rules?

Yes, in general, FAA follows the same procedures for all rule types. There are some differences as to which FAA official has authority to issue each type, and where you send petitions for FAA to adopt, amend, or repeal each type. Assume that the procedures in this subpart apply to all rules, except where we specify otherwise.

§11.25 How does FAA issue rules?

- (a) The FAA uses APA rulemaking procedures to adopt, amend, or repeal regulations. To propose or adopt a new regulation, or to change a current regulation, FAA will issue one or more of the following documents. We publish these rulemaking documents in the FEDERAL REGISTER unless we name and personally serve a copy of a rule on every person subject to it. We also make all documents available to the public by posting them in the Federal Docket Management System at http://www.regulations.gov.
- (1) An advance notice of proposed rulemaking (ANPRM).
- (2) A notice of proposed rulemaking (NPRM).
- (3) A supplemental notice of proposed rulemaking (SNPRM).
 - (4) A final rule.
- (5) A final rule with request for comments.
 - (6) A direct final rule.

- (b) Each of the rulemaking documents in paragraph (a) of this section generally contains the following information:
- (1) The topic involved in the rule-making document.
- (2) FAA's legal authority for issuing the rulemaking document.
- (3) How interested persons may participate in the rulemaking proceeding (for example, by filing written comments or making oral presentations at a public meeting).
- (4) Whom to call if you have questions about the rulemaking document.
- (5) The date, time, and place of any public meetings FAA will hold to discuss the rulemaking document.
- (6) The docket number and regulation identifier number (RIN) for the rule-making proceeding.

[Doc. No. 1999-6622, 65 FR 50863, Aug. 21, 2000, as amended at 72 FR 68474, Dec. 5, 2007]

§11.27 Are there other ways FAA collects specific rulemaking recommendations before we issue an NPRM?

Yes, the FAA obtains advice and recommendations from rulemaking advisory committees. One of these committees is the Aviation Rulemaking Advisory Committee (ARAC), which is a formal standing committee comprised of representatives of aviation associations and industry, consumer groups, and interested individuals. In conducting its activities, ARAC complies with the Federal Advisory Committee Act and the direction of FAA. We task ARAC with providing us with recommended rulemaking actions dealing with specific areas and problems. If we accept an ARAC recommendation to change an FAA rule, we ordinarily publish an NPRM using the procedures in this part. The FAA may establish other rulemaking advisory committees as needed to focus on specific issues for a limited period of time.

§ 11.29 May FAA change its regulations without first issuing an ANPRM or NPRM?

The FAA normally adds or changes a regulation by issuing a final rule after an NPRM. However, FAA may adopt, amend, or repeal regulations without

§ 11.31

first issuing an ANPRM or NPRM in the following situations:

(a) We may issue a final rule without first requesting public comment if, for good cause, we find that an NPRM is impracticable, unnecessary, or contrary to the public interest. We place that finding and a brief statement of the reasons for it in the final rule. For example, we may issue a final rule in response to a safety emergency.

(b) If an NPRM would be unnecessary because we do not expect to receive adverse comment, we may issue a direct final rule.

§11.31 How does FAA process direct final rules?

(a) A direct final rule will take effect on a specified date unless FAA receives an adverse comment or notice of intent to file an adverse comment within the comment period—generally 60 days after the direct final rule is published in the FEDERAL REGISTER. An adverse comment explains why a rule would be inappropriate, or would be ineffective or unacceptable without a change. It may challenge the rule's underlying premise or approach. Under the direct final rule process, we do not consider the following types of comments to be adverse:

- (1) A comment recommending another rule change, in addition to the change in the direct final rule at issue. We consider the comment adverse, however, if the commenter states why the direct final rule would be ineffective without the change.
- (2) A frivolous or insubstantial comment.
- (b) If FAA has not received an adverse comment or notice of intent to file an adverse comment, we will publish a confirmation document in the FEDERAL REGISTER, generally within 15 days after the comment period closes. The confirmation document tells the public the effective date of the rule.
- (c) If we receive an adverse comment or notice of intent to file an adverse comment, we will advise the public by publishing a document in the Federal Register before the effective date of the direct final rule. This document may withdraw the direct final rule in whole or in part. If we withdraw a direct final rule because of an adverse

comment, we may incorporate the commenter's recommendation into another direct final rule or may publish a notice of proposed rulemaking.

§11.33 How can I track FAA's rule-making activities?

The best ways to track FAA's rule-making activities are with the docket number or the regulation identifier number.

(a) Docket ID. We assign a docket ID to each rulemaking document proceeding. Each rulemaking document FAA issues in a particular rulemaking proceeding, as well as public comments on the proceeding, will display the same docket ID. This ID allows you to search the Federal Docket Management System (FDMS) for information on most rulemaking proceedings. You can view and copy docket materials during regular business hours at the U.S. Department of Transportation, Docket Operations, West Building Ground Floor, Room W12-140, 1200 New Jersey Avenue, SE., Washington, DC 20590. Or you can view and download docketed materials through the Internet at http://www.regulations.gov. If you can't find the material in the electronic docket, contact the person listed under FOR FURTHER INFORMATION CONTACT in the document you are interested in.

(b) Regulation identifier number. DOT publishes a semiannual agenda of all projected current and DOT rulemakings, reviews of existing regulations, and completed actions. This semiannual agenda appears in the Unified Agenda of Federal Regulations, published in the FEDERAL REGISTER in April and October of each year. The semiannual agenda tells the public about DOT's-including FAA's-regulatory activities. DOT assigns a regulation identifier number (RIN) to each individual rulemaking proceeding in the semiannual agenda. This number appears on all rulemaking documents published in the Federal Register and makes it easy for you to track those rulemaking proceedings in both the FEDERAL REGISTER and the semiannual regulatory agenda.

[Doc. No. 1999–6622, 65 FR 50863, Aug. 21, 2000, as amended at 72 FR 68474, Dec. 5, 2007]

§11.35 Does FAA include sensitive security information and proprietary information in the Federal Docket Management System (FDMS)?

(a) Sensitive security information. You should not submit sensitive security information to the rulemaking docket, unless you are invited to do so in our request for comments. If we ask for this information, we will tell you in the specific document how to submit this information, and we will provide a separate non-public docket for it. For all proposed rule changes involving civil aviation security, we review comments as we receive them, before they are placed in the docket. If we find that a comment contains sensitive security information, we remove that information before placing the comment in the general docket.

(b) Proprietary information. When we are aware of proprietary information filed with a comment, we do not place it in the docket. We hold it in a separate file to which the public does not have access, and place a note in the docket that we have received it. If we receive a request to examine or copy this information, we treat it as any other request under the Freedom of Information Act (5 U.S.C. 552). We process such a request under the DOT procedures found in 49 CFR part 7.

[Doc. No. 1999–6622, 65 FR 50863, Aug. 21, 2000, as amended at 72 FR 68474, Dec. 5, 2007]

§11.37 Where can I find information about an Airworthiness Directive, an airspace designation, or a petition handled in a region?

The FAA includes most documents concerning Airworthiness Directives, airspace designations, or petitions handled in a region in the electronic docket. If the information isn't in the docket, contact the person listed under FOR FURTHER INFORMATION CONTACT in the FEDERAL REGISTER document about the action.

§11.38 What public comment procedures does the FAA follow for Special Conditions?

Even though the Administrative Procedure Act does not require notice and comment for rules of particular applicability, FAA does publish proposed special conditions for comment. In the

following circumstances we may not invite comment before we issue a special condition. If we don't, we will invite comment when we publish the final special condition.

- (a) The FAA considers prior notice to be impracticable if issuing a design approval would significantly delay delivery of the affected aircraft. We consider such a delay to be contrary to the public interest.
- (b) The FAA considers prior notice to be unnecessary if we have provided previous opportunities to comment on substantially identical proposed special conditions, and we are satisfied that new comments are unlikely.

§11.39 How may I participate in FAA's rulemaking process?

You may participate in FAA's rulemaking process by doing any of the following:

- (a) File written comments on any rulemaking document that asks for comments, including an ANPRM, NPRM, SNPRM, a final rule with request for comments, or a direct final rule. Follow the directions for commenting found in each rulemaking document.
- (b) Ask that we hold a public meeting on any rulemaking, and participate in any public meeting that we hold.
- (c) File a petition for rulemaking that asks us to adopt, amend, or repeal a regulation.

§11.40 Can I get more information about a rulemaking?

You can contact the person listed under FOR FURTHER INFORMATION CONTACT in the preamble of a rule. That person can explain the meaning and intent of a proposed rule, the technical aspects of a document, the terminology in a document, and can tell you our published schedule for the rule-making process. We cannot give you information that is not already available to other members of the public. Department of Transportation policy on oral communications with the public during rulemaking appears in appendix 1 of this part.

§ 11.41

WRITTEN COMMENTS

§11.41 Who may file comments?

Anyone may file written comments about proposals and final rules that request public comments.

§ 11.43 What information must I put in my written comments?

- (a) Your written comments must be in English and must contain the following:
- (1) The docket number of the rulemaking document you are commenting on, clearly set out at the beginning of your comments.
- (2) Your name and mailing address, and, if you wish, other contact information, such as a fax number, telephone number, or e-mail address.
- (3) Your information, views, or arguments, following the instructions for participation in the rulemaking document on which you are commenting.
- (b) You should also include all material relevant to any statement of fact or argument in your comments, to the extent that the material is available to you and reasonable for you to submit. Include a copy of the title page of the document. Whether or not you submit a copy of the material to which you refer, you should indicate specific places in the material that support your position.

§ 11.45 Where and when do I file my comments?

- (a) Send your comments to the location specified in the rulemaking document on which you are commenting. If you are asked to send your comments to the Federal Document Management System, you may send them in either of the following ways:
- (1) By mail to: U.S. Department of Transportation, Docket Operations, West Building Ground Floor, Room W12–140, 1200 New Jersey Avenue, SE., Washington, DC 20590.
- (2) Through the Internet to http://www.regulations.gov.
- (3) In any other manner designated by FAA.
- (b) Make sure that your comments reach us by the deadline set out in the rulemaking document on which you are commenting. We will consider latefiled comments to the extent possible

only if they do not significantly delay the rulemaking process.

(c) We may reject your paper or electronic comments if they are frivolous, abusive, or repetitious. We may reject comments you file electronically if you do not follow the electronic filing instructions at the Federal Docket Management System Web site.

[Doc. No. 1999–6622, 65 FR 50863, Aug. 21, 2000, as amended at 72 FR 68474, Dec. 5, 2007]

§ 11.47 May I ask for more time to file my comments?

Yes, if FAA grants your request for more time to file comments, we grant all persons the same amount of time. We will notify the public of the extension by a document in the FEDERAL REGISTER. If FAA denies your request, we will notify you of the denial. To ask for more time, you must file a written or electronic request for extension at least 10 days before the end of the comment period. Your letter or message must.—

- (a) Show the docket number of the rule at the top of the first page;
- (b) State, at the beginning, that you are requesting an extension of the comment period;
- (c) Show that you have good cause for the extension and that an extension is in the public interest;
- (d) Be sent to the address specified for comments in the rulemaking document on which you are commenting.

PUBLIC MEETINGS AND OTHER PROCEEDINGS

§11.51 May I request that FAA hold a public meeting on a rulemaking action?

Yes, you may request that we hold a public meeting. FAA holds a public meeting when we need more than written comments to make a fully informed decision. Submit your written request to the address specified in the rulemaking document on which you are commenting. Specify at the top of your letter or message that you are requesting that the agency hold a public meeting. Submit your request no later than 30 days after our rulemaking notice. If we find good cause for a meeting, we will notify you and publish a

notice of the meeting in the FEDERAL REGISTER.

§11.53 What takes place at a public meeting?

A public meeting is a non-adversarial, fact-finding proceeding conducted by an FAA representative. Public meetings are announced in the FEDERAL REGISTER. We invite interested persons to attend and to present their views to the agency on specific issues. There are no formal pleadings and no adverse parties, and any regulation issued afterward is not necessarily based exclusively on the record of the meeting.

PETITIONS FOR RULEMAKING AND FOR EXEMPTION

§ 11.61 May I ask FAA to adopt, amend, or repeal a regulation, or grant relief from the requirements of a current regulation?

- (a) Using a petition for rulemaking, you may ask FAA to add a new regulation to title 14 of the Code of Federal Regulations (14 CFR) or ask FAA to amend or repeal a current regulation in 14 CFR.
- (b) Using a petition for exemption, you may ask FAA to grant you relief from current regulations in 14 CFR.

§ 11.63 How and to whom do I submit my petition for rulemaking or petition for exemption?

- (a) To submit a petition for rule-making or exemption—
- (1) By electronic submission, submit your petition for rulemaking or exemption to FAA through the Internet at http://www.regulations.gov, the Federal Docket Management System Web site. For additional instructions, you may visit http://www.faa.gov/regulations.
- (2) By paper submission, send the original signed copy of your petition for rulemaking or exemption to this address: U.S. Department of Transportation, Docket Operations, West Building Ground Floor, Room W12–140, 1200 New Jersey Avenue, SE., Washington, DC 20590.
- (b) Submit a petition for rulemaking or exemption from part 139 of this chapter—

- (1) To the appropriate FAA airport field office in whose area your airport is, or will be, established; and
- (2) To the U.S. Department of Transportation, Docket Operations, West Building Ground Floor, Room W12–140, 1200 New Jersey Avenue, SE., Washington, DC 20590 or by electronic submission to this Internet address: http://www.regulations.gov.
- (c) The FAA may designate other means by which you can submit petitions in the future.
- (d) Submit your petition for exemption 120 days before you need the exemption to take effect.

[Amdt. 11-50, 69 FR 22386, Apr. 26, 2004, as amended at 72 FR 68474, Dec. 5, 2007; Amdt. 11-55, 74 FR 202, Jan. 5, 2009]

§11.71 What information must I include in my petition for rule-making?

- (a) You must include the following information in your petition for rule-making:
- (1) Your name and mailing address and, if you wish, other contact information such as a fax number, telephone number, or e-mail address.
- (2) An explanation of your proposed action and its purpose.
- (3) The language you propose for a new or amended rule, or the language you would remove from a current rule.
- (4) An explanation of why your proposed action would be in the public interest.
- (5) Information and arguments that support your proposed action, including relevant technical and scientific data available to you.
- (6) Any specific facts or circumstances that support or demonstrate the need for the action you propose.
- (b) In the process of considering your petition, we may ask that you provide information or data available to you about the following:
- (1) The costs and benefits of your proposed action to society in general, and identifiable groups within society in particular.
- (2) The regulatory burden of your proposed action on small businesses, small organizations, small governmental jurisdictions, and Indian tribes.

§11.73

- (3) The recordkeeping and reporting burdens of your proposed action and whom the burdens would affect.
- (4) The effect of your proposed action on the quality of the natural and social environments.

§11.73 How does FAA process petitions for rulemaking?

After we have determined the disposition of your petition, we will contact you in writing about our decision. The FAA may respond to your petition for rulemaking in one of the following ways:

- (a) If we determine that your petition justifies our taking the action you suggest, we may issue an NPRM or ANPRM. We will do so no later than 6 months after the date we receive your petition. In making our decision, we consider:
- (1) The immediacy of the safety or security concerns you raise;
- (2) The priority of other issues the FAA must deal with; and
- (3) The resources we have available to address these issues.
- (b) If we have issued an ANPRM or NPRM on the subject matter of your petition, we will consider your arguments for a rule change as a comment in connection with the rulemaking proceeding. We will not treat your petition as a separate action.
- (c) If we have begun a rulemaking project in the subject area of your petition, we will consider your comments and arguments for a rule change as part of that project. We will not treat your petition as a separate action.
- (d) If we have tasked ARAC to study the general subject area of your petition, we will ask ARAC to review and evaluate your proposed action. We will not treat your petition as a separate action.
- (e) If we determine that the issues you identify in your petition may have merit, but do not address an immediate safety concern or cannot be addressed because of other priorities and resource constraints, we may dismiss your petition. Your comments and arguments for a rule change will be placed in a database, which we will examine when we consider future rulemaking.

§11.75 Does FAA invite public comment on petitions for rulemaking?

Generally, FAA does not invite public comment on petitions for rule-making.

§ 11.77 Is there any additional information I must include in my petition for designating airspace?

In petitions asking FAA to establish, amend, or repeal a designation of air-space, including special use airspace, you must include all the information specified by §11.71 and also:

- (a) The location and a description of the airspace you want assigned or designated:
- (b) A complete description of the activity or use to be made of that airspace, including a detailed description of the type, volume, duration, time, and place of the operations to be conducted in the area:
- (c) A description of the air navigation, air traffic control, surveillance, and communication facilities available and to be provided if we grant the designation; and
- (d) The name and location of the agency, office, facility, or person who would have authority to permit the use of the airspace when it was not in use for the purpose to which you want it assigned.

§11.81 What information must I include in my petition for an exemption?

You must include the following information in your petition for an exemption and submit it to FAA as soon as you know you need an exemption.

- (a) Your name and mailing address and, if you wish, other contact information such as a fax number, telephone number, or e-mail address:
- (b) The specific section or sections of 14 CFR from which you seek an exemption:
- (c) The extent of relief you seek, and the reason you seek the relief;
- (d) The reasons why granting your request would be in the public interest; that is, how it would benefit the public as a whole:
- (e) The reasons why granting the exemption would not adversely affect safety, or how the exemption would provide a level of safety at least equal

to that provided by the rule from which you seek the exemption;

- (f) A summary we can publish in the FEDERAL REGISTER, stating:
- (1) The rule from which you seek the exemption; and
- (2) A brief description of the nature of the exemption you seek;
- (g) Any additional information, views or arguments available to support your request; and
- (h) If you want to exercise the privileges of your exemption outside the United States, the reason why you need to do so.

§ 11.83 How can I operate under an exemption outside the United States?

If you want to be able to operate under your exemption outside the United States, you must request this when you petition for relief and give us the reason for this use. If you do not provide your reason or we determine that it does not justify this relief, we will limit your exemption to use within the United States. Before we extend your exemption for use outside the United States, we will verify that the exemption would be in compliance with the Standards of the International Civil Aviation Organization (ICAO). If it would not, but we still believe it would be in the public interest to allow you to do so, we will file a difference with ICAO. However, a foreign country still may not allow you to operate in that country without meeting the ICAO standard.

§11.85 Does FAA invite public comment on petitions for exemption?

Yes, FAA publishes information about petitions for exemption in the FEDERAL REGISTER. The information includes—

- (a) The docket number of the petition:
- (b) The citation to the rule or rules from which the petitioner requested relief:
 - (c) The name of the petitioner;
- (d) The petitioner's summary of the action requested and the reasons for requesting it; and
- (e) A request for comments to assist FAA in evaluating the petition.

§11.87 Are there circumstances in which FAA may decide not to publish a summary of my petition for exemption?

The FAA may not publish a summary of your petition for exemption and request comments if you present or we find good cause why we should not delay action on your petition. The factors we consider in deciding not to request comment include:

- (a) Whether granting your petition would set a precedent.
- (b) Whether the relief requested is identical to exemptions granted previously.
- (c) Whether our delaying action on your petition would affect you adversely.
- (d) Whether you filed your petition in a timely manner.

§ 11.89 How much time do I have to submit comments to FAA on a petition for exemption?

The FAA states the specific time allowed for comments in the FEDERAL REGISTER notice about the petition. We usually allow 20 days to comment on a petition for exemption.

§ 11.91 How does FAA inform me of its decision on my petition for exemption?

The FAA will notify you in writing about its decision on your petition. A copy of this decision is also placed in the public docket. We will include the docket number associated with your petition in our letter to you.

[Doc. No. FAA–2005–22982, 71 FR 1485, Jan. 10, 2006]

§11.101 May I ask FAA to reconsider my petition for rulemaking or petition for exemption if it is denied?

Yes, you may petition FAA to reconsider your petition denial. You must submit your request to the address to which you sent your original petition, and FAA must receive it within 60 days after we issued the denial. For us to accept your petition, show the following:

- (a) That you have a significant additional fact and why you did not present it in your original petition;
- (b) That we made an important factual error in our denial of your original petition; or

§ 11.103

(c) That we did not correctly interpret a law, regulation, or precedent.

§11.103 What exemption relief may be available to federal, state, and local governments when operating aircraft that are not public aircraft?

The Federal Aviation Administration may grant a federal, state, or local government an exemption from part A of subtitle VII of title 49 United States Code, and any regulation issued under that authority that is applicable to an aircraft as a result of the Independent Safety Board Act Amendments of 1994, Public Law 103–411, if—

- (a) The Administrator finds that granting the exemption is necessary to prevent an undue economic burden on the unit of government: and
- (b) The Administrator certifies that the aviation safety program of the unit of government is effective and appropriate to ensure safe operations of the type of aircraft operated by the unit of government.

[68 FR 25488, May 13, 2003]

Subpart B—Paperwork Reduction Act Control Numbers

§ 11.201 Office of Management and Budget (OMB) control numbers assigned under the Paperwork Reduction Act.

- (a) The Paperwork Reduction Act of 1995 (44 U.S.C. 3501–3520) requires FAA to get approval from OMB for our information collection activities, and to list a record of those approvals in the FEDERAL REGISTER. This subpart lists the control numbers OMB assigned to FAA's information collection activities.
- (b) The table listing OMB control numbers assigned to FAA's information collection activities follows:

14 CFR part or section identified and described	Current OMB control number
Part 14	2120-0539
Part 17	2120-0632
Part 21	2120-0018, 2120-0552
Part 34	2120-0508
Part 39	2120-0056
Part 43	2120-0020
Part 45	2120-0508
Part 47	2120-0024, 2120-0042
Part 48	2120-0765
Part 49	2120-0043
Part 60	2120-0680

14 CFR part or section identified and described	Current OMB control number
Part 61	2120–0021, 2120–0034, 2120–0543, 2120–0571
Part 63	2120-0007
Part 65	2120-0022, 2120-0535, 2120-0571, 2120-0648
Part 67	2120-0034, 2120-0543
Part 77	2120-0001
Part 91	2120-0005, 2120-0026, 2120-0027, 2120-0573, 2120-0606, 2120-0620, 2120-0631, 2120-0651
Part 93	2120-0524, 2120-0606, 2120-0639
Part 101	2120-0027
Part 105	2120-0027, 2120-0641
Part 107	2120-0005, 2120-0021, 2120-0027,
	2120-0767, 2120-0768.
Part 119	2120-0593
Part 121	2120-0008, 2120-0028, 2120-0535, 2120-0571, 2120-0600, 2120-0606,
	2120-0571, 2120-0600, 2120-0606,
	2120-0614, 2120-0616, 2120-0631,
	2120–0651, 2120–0653, 2120–0691, 2120–0739, 2120–0760, 2120–0766.
Part 125	2120–0739, 2120–0700, 2120–0700. 2120–0028, 2120–0085, 2120–0616, 2120–0651
Part 129	2120–0028, 2120–0536, 2120–0616, 2120–0638
Part 133	2120-0044
Part 135	2120-0003, 2120-0028, 2120-0039,
	2120-0003, 2120-0028, 2120-0039, 2120-0535, 2120-0571, 2120-0600, 2120-0606, 2120-0614, 2120-0616,
	2120–0606, 2120–0614, 2120–0616, 2120–0620, 2120–0631, 2120–0653, 2120–0766.
Part 137	2120-0049
Part 139	2120-0045, 2120-0063
Part 141	2120-0009
Part 142	2120-0570
Part 145	2120-0003, 2120-0010, 2120-0571
Part 147	2120-0040
Part 150	2120-0517
Part 157	2120-0036
Part 158	2120-0557
Part 161	2120-0563
Part 171	2120-0014
Part 183	2120-0033, 2120-0604
Part 193	2120–0646
Part 198	2120–0514
Part 400	2120–0643, 2120–0644, 0649
Part 401	2120–0608
Part 440	2120-0601
	2120-0507
SFAR 71	2120-0620

[Doc. No. 1999–6622, 65 FR 50863, Aug. 21, 2000, as amended by Amdt. 11–47, 67 FR 9553, Mar. 1, 2002; Amdt. 11–49, 68 FR 61321, Oct. 27, 2003; Amdt. 11–49, 68 FR 70132, Dec. 17, 2003; 70 FR 40163, July 12, 2005; 71 FR 63426, Oct. 30, 2006; 72 FR 59599, Oct. 22, 2007; Amdt. 11–56, 79 FR 12937, Mar. 7, 2014; Amdt. 11–57, 80 FR 58586, Sept. 30, 2015; Doc. FAA–2015–7396, Amdt. 11–58, 80 FR 79255, Dec. 21, 2015; Doc. FAA–2011–1136, Amdt. 11–59, 81 FR 13969, Mar. 16, 2016; Doc. FAA–2014–0554, Amdt. 11–60, 81 FR 33117, May 24, 2016; 81 FR 38573, June 14, 2016; Doc. FAA–2016–9064, Amdt. 11–61, 81 FR 59129, Aug. 29, 2016]

APPENDIX 1 TO PART 11—ORAL COMMU-NICATIONS WITH THE PUBLIC DURING RULEMAKING

1. What is an ex parte contact?

"Ex parte" is a Latin term that means "one sided," and indicates that not all parties to an issue were present when it was discussed. An ex parte contact involving rulemaking is any communication between FAA and someone outside the government regarding a specific rulemaking proceeding, before that proceeding closes. A rulemaking proceeding does not close until we publish the final rule or withdraw the NPRM. Because an ex parte contact excludes other interested persons, including the rest of the public, from the communication, it may give an unfair advantage to one party, or appear to do so.

2. Are written comments to the docket ex parte contacts?

Written comments submitted to the docket are not ex parte contacts because they are available for inspection by all members of the public.

3. What is DOT policy on ex parte contacts?

It is DOT policy to provide for open development of rules and to encourage full public participation in rulemaking actions. In addition to providing opportunity to respond in writing to an NPRM and to appear and be heard at a hearing, DOT policy encourages agencies to contact the public directly when we need factual information to resolve questions of substance. It also encourages DOT agencies to be receptive to appropriate contacts from persons affected by or interested in a proposed action. But under some circumstances an ex parte contact could affect the basic openness and fairness of the rulemaking process. Even the appearance of impropriety can affect public confidence in the process. For this reason, DOT policy sets careful guidelines for these contacts. The kind of ex parte contacts permitted and the procedures we follow depend on when the contact occurs in the rulemaking process.

4. What kinds of ex parte contacts does DOT policy permit before we issue an ANPRM, NPRM, Supplemental NPRM, or immediately adopted final rule?

The DOT policy authorizes ex parte contacts that we need to obtain technical and economic information. We need this information to decide whether to issue a regulation and what it should say. Each contact that influences our development of the regulation is noted in the preamble. For multiple contacts that are similar, we may provide only a general discussion. For contacts not discussed in the preamble, we place a report discussing

each contact or group of related contacts in the rulemaking docket when it is opened.

5. Does DOT policy permit ex parte contacts during the comment period?

No, during the comment period, the public docket is available for written comments from any member of the public. These comments can be examined and responded to by any interested person. Because this public forum is available, DOT policy discourages ex parte contacts during the comment period. They are not necessary to collect the information the agency needs to make its decision

6. What if the FAA believes it needs to meet with members of the public to discuss the proposal?

If the FAA determines that it would be helpful to invite members of the public to make oral presentations to it regarding the proposal, we will announce a public meeting in the FEDERAL REGISTER.

7. Are any oral contacts concerning the proposal permitted during the comment period?

If you contact the agency with questions regarding the proposal during the comment period, we can only provide you with information that has already been made available to the general public. If you contact the agency to discuss the proposal, you will be told that the proper avenue of communication during the comment period is a written communication to the docket.

8. If a substantive ex parte contact does occur during the comment period, what does FAA do?

While FAA tries to ensure that FAA personnel and the public are aware of DOT policy, substantive ex parte contacts do occasionally occur, for example, at meetings not intended for that purpose. In such a case, we place a summary of the contact and a copy of any materials provided at the meeting in the rulemaking docket. We encourage participants in such a meeting to file written comments in the docket.

9. Does DOT policy permit ex parte contacts the comment period has closed?

DOT policy strongly discourages ex parte contacts initiated by commenters to discuss their position on the proposal once the comment period has closed. Such a contact at this time would be improper, since other interested persons would not have an opportunity to respond. If we need further information regarding a comment in the docket, we may request this from a commenter. A record of this contact and the information provided is placed in the docket. If we need to make other contacts to update factual information, such as economic data, we will disclose this information in the final rule

Pt. 13

docket or in the economic studies accompanying it, which are available in the dock-

10. What if FAA needs to meet with interested persons to discuss the proposal after the comment period has closed?

If FAA determines that it would be helpful to meet with a person or group after the close of the comment period to discuss a course of action to be taken, we will announce the meeting in the FEDERAL REG-ISTER. We will also consider reopening the comment period. If an inappropriate ex parte contact does occur after the comment period closes, a summary of the contact and a copy of any material distributed during meeting will be placed in the docket if it could be seen as influencing the rulemaking process.

11. Under what circumstances will FAA reopen the comment period?

If we receive an ex parte communication after the comment period has closed that could substantially influence the rulemaking, we may reopen the comment period. DOT policy requires the agency to carefully consider whether the substance of the contact will give the commenter an unfair advantage, since the rest of the public may not see the record of the contact in the docket. When the substance of a proposed rule is significantly changed as a result of such an oral communication, DOT policy and practice requires that the comment period be reopened by issuing a supplemental NPRM in which the reasons for the change are discussed.

12. What if I have important information for FAA and the comment period is closed?

You may always provide FAA with written information after the close of the comment period and it will be considered if time permits. Because contacts after the close of the comment may not be seen by other interested persons, if they substantially and specifically influence the FAA's decision, we may need to reopen the comment period.

PART 13—INVESTIGATIVE AND **ENFORCEMENT PROCEDURES**

Subpart A—Investigative Procedures

Sec.

- 13.1 Reports of violations.
- 13.3 Investigations (general).
- 13.5 Formal complaints.
- 13.7 Records, documents and reports.

Subpart B—Administrative Actions

13.11 Administrative disposition of certain violations.

Subpart C—Legal Enforcement Actions

- 13.13 Consent orders.
- 13.14 Civil penalties: General.
- 13.15 Civil penalties: Other than by administrative assessment.
- 13.16 Civil penalties: Administrative assessment against a person other than an individual acting as a pilot, flight engineer, mechanic, or repairman. Administrative assessment against all persons for hazardous materials violations.
- 13.17 Seizure of aircraft.
- 13.18 Civil penalties: Administrative assessment against an individual acting as a pilot, flight engineer, mechanic, or repairman.
- 13.19 Certificate action.
- 13.20 Orders of compliance, cease and desist orders, orders of denial, and other orders.
- 13.21 Military personnel.
- 13.23 Criminal penalties.
- 13.25 Injunctions.
- Final order of Hearing Officer in cer-13.27tificate of aircraft registration proceedings.
- 13.29 Civil penalties: Streamlined enforcement procedures for certain security vio-

Subpart D-Rules of Practice for FAA **Hearings**

- 13.31 Applicability.
- 13.33 Appearances.
- 13.35 Request for hearing.
- Hearing Officer's powers. 13.39 Disqualification of Hearing Officer.
- 13.41 [Reserved]
- 13.43 Service and filing of pleadings, motions, and documents.
- 13.44 Computation of time and extension of time.
- 13.45 Amendment of notice and answer.13.47 Withdrawal of notice or request for hearing.
- 13.49 Motions.
- 13.51 Intervention.
- 13.53 Depositions.
- Notice of hearing. 13.55
- Subpoenas and witness fees. 13.57
- 13.59 Evidence.
- 13.61 Argument and submittals.
- 13.63 Record.

Subpart E—Orders of Compliance Under the Hazardous Materials Transportation Act

- Applicability.
- 13.73 Notice of proposed order of compli-
- 13.75 Reply or request for hearing.13.77 Consent order of compliance.
- 13.79 Hearing.
- 13.81 Order of immediate compliance.
- 13 83 Appeal.
- Filing, service and computation of 13.85time.

13.87 Extension of time.

Subpart F-Formal Fact-Finding Investigation Under an Order of Investigation

- Applicability.
- Order of investigation.
- 13.105 Notification.
- Designation of additional parties.
- Convening the investigation. 13.109
- 13.111 Subpoenas
- 13.113 Noncompliance with the investigative process.
- 13.115 Public proceedings.13.117 Conduct of investigative proceeding or deposition.
- 13.119 Rights of persons against self-incrimination.
- 13.121 Witness fees.
- 13.123 Submission by party to the investigation.
- 13.125 Depositions.
- Reports, decisions and orders. 13.127
- 13.129 Post-investigation action.
- 13.131 Other procedures.

Subpart G-Rules of Practice in FAA Civil **Penalty Actions**

- 13.201 Applicability.
- 13.202 Definitions.
- Separation of functions. 13.203
- 13.204 Appearances and rights of parties.
- 13.205 Administrative law judges.
- 13.206 Intervention.
- 13.207Certification of documents.
- 13.208 Complaint.
- 13.209 Answer.
- Filing of documents. 13.210
- Service of documents. 13.211
- 13.212 Computation of time.
- 13.213 Extension of time
- Amendment of pleadings. 13.214 13.215
- Withdrawal of complaint or request for hearing. 13.216 Waivers.
- 13.217 Joint procedural or discovery schedule.
- 13.218 Motions.
- 13.219 Interlocutory appeals.
- 13.220 Discovery
- Notice of hearing. 13.221
- 13.222 Evidence.
- Standard of proof. 13.223
- 13.224 Burden of proof.
- 13.225 Offer of proof.
- 13.226Public disclosure of evidence.
- 13.227 Expert or opinion witnesses.
- 13.228 Subpoenas.
- 13.229 Witness fees.
- 13.230 Record.
- 13.231 Argument before the administrative law judge.
- 13.232 Initial decision.
- 13.233 Appeal from initial decision.
- Petition to reconsider or modify a 13.234 final decision and order of the FAA decisionmaker on appeal.

13.235 Judicial review of a final decision and

Subpart H—Civil Monetary Penalty Inflation **Adjustment**

13.301 Scope and purpose.

Subpart I—Flight Operational Quality **Assurance Programs**

13.401 Flight Operational Quality Assurance program: Prohibition against use of data for enforcement purposes.

AUTHORITY: 18 U.S.C. 6002; 28 U.S.C. 2461 (note); 49 U.S.C. 106(g), 5121-5128, 40113-40114, 44103-44106, 44701-44703, 44709-44710, 44713, 46101-46111, 46301, 46302 (for a violation of 49 U.S.C. 46504), 46304-46316, 46318, 46501-46502, 46504-46507, 47106, 47107, 47111, 47122, 47306, 47531-47532; 49 CFR 1.47.

Source: Docket No. 18884, 44 FR 63723, Nov. 5, 1979, unless otherwise noted.

Subpart A—Investigative **Procedures**

§ 13.1 Reports of violations.

(a) Any person who knows of a violation of the Federal Aviation Act of 1958, as amended, the Hazardous Materials Transportation Act relating to the transportation or shipment by air of hazardous materials, the Airport and Airway Development Act of 1970, the Airport and Airway Improvement Act of 1982, the Airport and Airway Improvement Act of 1982 as amended by the Airport and Airway Safety and Capacity Expansion Act of 1987, or any rule, regulation, or order issued thereunder, should report it to appropriate personnel of any FAA regional or district office.

(b) Each report made under this section, together with any other information the FAA may have that is relevant to the matter reported, will be reviewed by FAA personnel to determine the nature and type of any additional investigation or enforcement action the FAA will take.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13-17, 53 FR 33783, Aug. 31,

§13.3 Investigations (general).

(a) Under the Federal Aviation Act of 1958, as amended, (49 U.S.C. 1301 et seq.),

the Hazardous Materials Transportation Act (49 U.S.C. 1801 et seq.), the Airport and Airway Development Act of 1970 (49 U.S.C. 1701 et seq.), the Airport and Airway Improvement Act of 1982 (49 U.S.C. 2201 et seq.), the Airport and Airway Improvement Act of 1982 (as amended, 49 U.S.C. App. 2201 et seq., Airport and Airway Safety and Capacity Expansion Act of 1987), and the Regulations of the Office of the Secretary of Transportation (49 CFR 1 et seq.), the Administrator may conduct investigations, hold hearings, issue subpoenas, require the production of relevant documents, records, and property, and take evidence and depositions.

- (b) For the purpose of investigating alleged violations of the Federal Aviation Act of 1958, as amended the Hazardous Materials Transportation Act, the Airport and Airway Development Act of 1970, the Airport and Airway Improvement Act of 1982, the Airport and Airway Improvement Act of 1982 as amended by the Airport and Airway Safety and Capacity Expansion Act of 1987, or any rule, regulation, or order issued thereunder, the Administrator's authority has been delegated to the various services and or offices for matters within their respective areas for all routine investigations. When the compulsory processes of sections 313 and 1004 (49 U.S.C. 1354 and 1484) of the Federal Aviation Act, or section 109 of the Hazardous Materials Transportation Act (49 U.S.C. 1808) are invoked, the Administrator's authority has been delegated to the Chief Counsel, the Deputy Chief Counsel, each Assistant Chief Counsel, each Regional Counsel, the Aeronautical Center Counsel, and the Technical Center Counsel.
- (c) In conducting formal investigations, the Chief Counsel, the Deputy Chief Counsel, each Assistant Chief Counsel, each Regional Counsel, the Aeronautical Center Counsel, and the Technical Center Counsel may issue an order of investigation in accordance with subpart F of this part.
- (d) A complaint against the sponsor, proprietor, or operator of a Federally-assisted airport involving violations of the legal authorities listed in §16.1 of this chapter shall be filed in accordance with the provisions of part 16 of

this chapter, except in the case of complaints, investigations, and proceedings initiated before December 16, 1996, the effective date of part 16 of this chapter.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–17, 53 FR 33783, Aug. 31, 1988; 53 FR 35255, Sept. 12, 1988; Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–27, 61 FR 54004, Oct. 16, 1996; Amdt. 13–29, 62 FR 46865, Sept. 4, 1997]

§13.5 Formal complaints.

- (a) Any person may file a complaint with the Administrator with respect to anything done or omitted to be done by any person in contravention of any provision of any Act or of any regulation or order issued under it, as to matters within the jurisdiction of the Administrator. This section does not apply to complaints against the Administrator or employees of the FAA acting within the scope of their employment.
- (b) Complaints filed under this section must—
- (1) Be submitted in writing and identified as a complaint filed for the purpose of seeking an appropriate order or other enforcement action;
- (2) Be submitted to the Federal Aviation Administration, Office of the Chief Counsel, Attention: Enforcement Docket (AGC-10), 800 Independence Avenue, S.W., Washington, DC 20591;
- (3) Set forth the name and address, if known, of each person who is the subject of the complaint and, with respect to each person, the specific provisions of the Act or regulation or order that the complainant believes were violated;
- (4) Contain a concise but complete statement of the facts relied upon to substantiate each allegation;
- (5) State the name, address and telephone number of the person filing the complaint; and
- (6) Be signed by the person filing the complaint or a duly authorized representative.
- (c) Complaints which do not meet the requirements of paragraph (b) of this section will be considered reports under §13.1.
- (d) Complaints which meet the requirements of paragraph (b) of this section will be docketed and a copy

mailed to each person named in the complaint.

- (e) Any complaint filed against a member of the Armed Forces of the United States acting in the performance of official duties shall be referred to the Secretary of the Department concerned for action in accordance with the procedures set forth in §13.21 of this part.
- (f) The person named in the complaint shall file an answer within 20 days after service of a copy of the complaint.
- (g) After the complaint has been answered or after the allotted time in which to file an answer has expired, the Administrator shall determine if there are reasonable grounds for investigating the complaint.
- (h) If the Administrator determines that a complaint does not state facts which warrant an investigation or action, the complaint may be dismissed without a hearing and the reason for the dismissal shall be given, in writing, to the person who filed the complaint and the person named in the complaint.
- (i) If the Administrator determines that reasonable grounds exist, an informal investigation may be initiated or an order of investigation may be issued in accordance with subpart F of this part, or both. Each person named in the complaint shall be advised which official has been delegated the responsibility under §13.3(b) or (c) for conducting the investigation.
- (j) If the investigation substantiates the allegations set forth in the complaint, a notice of proposed order may be issued or other enforcement action taken in accordance with this part.
- (k) The complaint and other pleadings and official FAA records relating to the disposition of the complaint are maintained in current docket form in the Enforcement Docket (AGC-10), Office of the Chief Counsel, Federal Aviation Administration, 800 Independence Avenue, S.W., Washington, D. C. 20591. Any interested person may examine any docketed material at that office, at any time after the docket is established, except material that is ordered withheld from the public under applicable law or regulations, and may ob-

tain a photostatic or duplicate copy upon paying the cost of the copy.

(Secs. 313(a), 314(a), 601 through 610, and 1102 of the Federal Aviation Act of 1958 (49 U.S.C. 1354(a), 1421 through 1430, 1502); sec. 6(c), Dept. of Transportation Act (49 U.S.C. 1655(c)))

[Doc. No. 13–14, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–16, 45 FR 35307, May 27, 1980; Amdt. 13–19, 54 FR 39290, Sept. 25, 1989]

§ 13.7 Records, documents and reports.

Each record, document and report that the Federal Aviation Regulations require to be maintained, exhibited or submitted to the Administrator may be used in any investigation conducted by the Administrator; and, except to the extent the use may be specifically limited or prohibited by the section which imposes the requirement, the records, documents and reports may be used in any civil penalty action, certificate action, or other legal proceeding.

Subpart B—Administrative Actions

§ 13.11 Administrative disposition of certain violations.

- (a) If it is determined that a violation or an alleged violation of the Federal Aviation Act of 1958, or an order or regulation issued under it, or of the Hazardous Materials Transportation Act, or an order or regulation issued under it, does not require legal enforcement action, an appropriate official of the FAA field office responsible for processing the enforcement case or other appropriate FAA official may take administrative action in disposition of the case.
- (b) An administrative action under this section does not constitute a formal adjudication of the matter, and may be taken by issuing the alleged violator—
- (1) A "Warning Notice" which recites available facts and information about the incident or condition and indicates that it may have been a violation; or
- (2) A "Letter of Correction" which confirms the FAA decision in the matter and states the necessary corrective action the alleged violator has taken or agrees to take. If the agreed corrective action is not fully completed, legal enforcement action may be taken.

Subpart C—Legal Enforcement Actions

§13.13 Consent orders.

- (a) At any time before the issuance of an order under this subpart, the official who issued the notice and the person subject to the notice may agree to dispose of the case by the issuance of a consent order by the official.
- (b) A proposal for a consent order, submitted to the official who issued the notice, under this section must include—
 - (1) A proposed order;
- (2) An admission of all jurisdictional facts:
- (3) An express waiver of the right to further procedural steps and of all rights to judicial review; and
- (4) An incorporation by reference of the notice and an acknowledgment that the notice may be used to constructhe terms of the order.
- (c) If the issuance of a consent order has been agreed upon after the filing of a request for hearing in accordance with subpart D of this part, the proposal for a consent order shall include a request to be filed with the Hearing Officer withdrawing the request for a hearing and requesting that the case be dismissed.

§13.14 Civil penalties: General.

- (a) Any person who violates any of the following statutory provisions, or any rule, regulation, or order issued thereunder, is subject to a civil penalty of not more than the amount specified in 49 U.S.C. chapter 463 for each violation:
- (1) Chapter 401 (except sections 40103(a) and (d), 40105, 40116, and 40117);
 - (2) Chapter 441 (except section 44109);
 - (3) Section 44502(b) or (c);
- (4) Chapter 447 (except sections 44717 and 44719–44723);
- (5) Chapter 451;
- (6) Sections 46301(b), 46302 (for a violation of 49 U.S.C. 46504), or 46318;
 - (7) Section 47107(b); or
 - (8) Sections 47528 through 47530.
- (b) Any person who knowingly commits an act in violation of 49 U.S.C. chapter 51 or a regulation prescribed or order issued under that chapter, is subject to a civil penalty under 49 U.S.C. 5123.

(c) The minimum and maximum amounts of civil penalties for violations of the statutory provisions specified in paragraphs (a) and (b) of this section, or rules, regulations, or orders issued thereunder, are periodically adjusted for inflation in accordance with the formula established in 28 U.S.C. 2461 note and implemented in 14 CFR part 13, subpart H.

[Amdt. 13–32; 69 FR 59495, Oct. 4, 2004; Amdt. 13–32; 70 FR 1813, Jan. 11, 2005; 71 FR 70464, Dec. 5, 2006]

§ 13.15 Civil penalties: Other than by administrative assessment.

- (a) The FAA uses the procedures in this section when it seeks a civil penalty other than by the administrative assessment procedures in §§ 13.16 or 13.18.
- (b) The authority of the Administrator, under 49 U.S.C. chapter 463, to seek a civil penalty for a violation cited in §13.14(a), and the ability to refer cases to the United States Attorney General, or the delegate of the Attorney General, for prosecution of civil penalty actions sought by the Administrator is delegated to the Chief Counsel; the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel: and the Technical Center Counsel. This delegation applies to cases involving:
- (1) An amount in controversy in excess of:
- (i) \$50,000, if the violation was committed by any person before December 12, 2003;
- (ii) \$400,000, if the violation was committed by a person other than an individual or small business concern on or after December 12, 2003;
- (iii) \$50,000, if the violation was committed by an individual or small business concern on or after December 12, 2003; or
- (2) An in rem action, seizure of aircraft subject to lien, suit for injunctive relief, or for collection of an assessed civil penalty.
- (c) The Administrator may compromise any civil penalty proposed under this section, before referral to the United States Attorney General, or

the delegate of the Attorney General, for prosecution.

- (1) The Administrator, through the Chief Counsel; the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel; or the Technical Center Counsel sends a civil penalty letter to the person charged with a violation cited in §13.14(a). The civil penalty letter contains a statement of the charges, the applicable law, rule, regulation, or order, the amount of civil penalty that the Administrator will accept in full settlement of the action or an offer to compromise the civil penalty.
- (2) Not later than 30 days after receipt of the civil penalty letter, the person charged with a violation may present any material or information in answer to the charges to the agency attorney, either orally or in writing, that may explain, mitigate, or deny the violation or that may show extenuating circumstances. The Administrator will consider any material or information submitted in accordance with this paragraph to determine whether the person is subject to a civil penalty or to determine the amount for which the Administrator will compromise the action.
- (3) If the person charged with the violation offers to compromise for a specific amount, that person must send to the agency attorney a certified check or money order for that amount, payable to the Federal Aviation Administration. The Chief Counsel; the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; Aeronautical Center Counsel; or the Technical Center Counsel may accept the certified check or money order or may refuse and return the certified check or money order.
- (4) If the offer to compromise is accepted by the Administrator, the agency attorney will send a letter to the person charged with the violation stating that the certified check or money order is accepted in full settlement of the civil penalty action.

(5) If the parties cannot agree to compromise the civil penalty action or the offer to compromise is rejected and the certified check or money order submitted in compromise is returned, the Administrator may refer the civil penalty action to the United States Attorney General, or the delegate of the Attorney General, to begin proceedings in a United States district court, pursuant to the authority in 49 U.S.C. 46305, to prosecute and collect the civil penalty.

[Amdt. 13–18, 53 FR 34653, Sept. 7, 1988, as amended by Amdt. 13–20, 55 FR 15128, Apr. 20, 1990; Amdt. 13–29, 62 FR 46865, Sept. 4, 1997; Amdt. 13–32; 69 FR 59495, Oct. 4, 2004]

- § 13.16 Civil Penalties: Administrative assessment against a person other than an individual acting as a pilot, flight engineer, mechanic, or repairman. Administrative assessment against all persons for hazardous materials violations.
- (a) The FAA uses these procedures when it assesses a civil penalty against a person other than an individual acting as a pilot, flight engineer, mechanic, or repairman for a violation cited in the first sentence of 49 U.S.C. 46301(d)(2) or in 49 U.S.C. 47531, or any implementing rule, regulation or order.
- (b) District court jurisdiction. Notwithstanding the provisions of paragraph (a) of this section, the United States district courts have exclusive jurisdiction of any civil penalty action initiated by the FAA for violations described in those paragraphs, under 49 U.S.C. 46301(d)(4), if—
- (1) The amount in controversy is more than \$50,000 for a violation committed by any person before December 12, 2003;
- (2) The amount in controversy is more than \$400,000 for a violation committed by a person other than an individual or small business concern on or after December 12, 2003;
- (3) The amount in controversy is more than \$50,000 for a violation committed by an individual or a small business concern on or after December 12, 2003;
- (4) The action is in rem or another action in rem based on the same violation has been brought;

- (5) The action involves an aircraft subject to a lien that has been seized by the Government; or
- (6) Another action has been brought for an injunction based on the same violation.
- (c) Hazardous materials violations. The FAA may assess a civil penalty against any person who knowingly commits an act in violation of 49 U.S.C. chapter 51 or a regulation prescribed or order issued under that chapter, under 49 U.S.C. 5123 and 49 CFR 1.47(k). An order assessing a civil penalty for a violation under 49 U.S.C. chapter 51, or a rule, regulation, or order issued thereunder, is issued only after the following factors have been considered:
- (1) The nature, circumstances, extent, and gravity of the violation;
- (2) With respect to the violator, the degree of culpability, any history of prior violations, the ability to pay, and any effect on the ability to continue to do business; and
- (3) Such other matters as justice may require.
- (d) Order assessing civil penalty. An order assessing civil penalty may be issued for a violation described in paragraphs (a) or (c) of this section, or as otherwise provided by statute, after notice and opportunity for a hearing. A person charged with a violation may be subject to an order assessing civil penalty in the following circumstances:
- (1) An order assessing civil penalty may be issued if a person charged with a violation submits or agrees to submit a civil penalty for a violation.
- (2) An order assessing civil penalty may be issued if a person charged with a violation does not request a hearing under paragraph (g)(2)(ii) of this section within 15 days after receipt of a final notice of proposed civil penalty.
- (3) Unless an appeal is filed with the FAA decisionmaker in a timely manner, an initial decision or order of an administrative law judge shall be considered an order assessing civil penalty if an administrative law judge finds that an alleged violation occurred and determines that a civil penalty, in an amount found appropriate by the administrative law judge, is warranted.
- (4) Unless a petition for review is filed with a U.S. Court of Appeals in a timely manner, a final decision and

- order of the Administrator shall be considered an order assessing civil penalty if the FAA decisionmaker finds that an alleged violation occurred and a civil penalty is warranted.
- (e) Delegation of authority. (1) The authority of the Administrator under 49 U.S.C. 46301(d), 47531, and 5123, and 49 CFR 1.47(k) to initiate and assess civil penalties for a violation of those statutes or a rule, regulation, or order issued thereunder, is delegated to the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel; and the Technical Center Counsel.
- (2) The authority of the Administrator under 49 U.S.C. 5123, 49 CFR 1.47(k), 49 U.S.C. 46301(d), and 49 U.S.C. 46305 to refer cases to the Attorney General of the United States, or the delegate of the Attorney General, for collection of civil penalties is delegated to the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel; and the Technical Center Counsel.
- (3) The authority of the Administrator under 49 U.S.C. 46301(f) to compromise the amount of a civil penalty imposed is delegated to the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel; and the Technical Center Counsel.
- (4) The authority of the Administrator under 49 U.S.C. 5123 (e) and (f) and 49 CFR 1.47(k) to compromise the amount of a civil penalty imposed is delegated to the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel; and the Technical Center Counsel.
- (f) Notice of proposed civil penalty. A civil penalty action is initiated by

sending a notice of proposed civil penalty to the person charged with a violation or to the agent for services for the person under 49 U.S.C. 46103. A notice of proposed civil penalty will be sent to the individual charged with a violation or to the president of the corporation or company charged with a violation. In response to a notice of proposed civil penalty, a corporation or company may designate in writing another person to receive documents in that civil penalty action. The notice of proposed civil penalty contains a statement of the charges and the amount of the proposed civil penalty. Not later than 30 days after receipt of the notice of proposed civil penalty, the person charged with a violation shall-

- (1) Submit the amount of the proposed civil penalty or an agreed-upon amount, in which case either an order assessing civil penalty or compromise order shall be issued in that amount;
- (2) Submit to the agency attorney one of the following:
- (i) Written information, including documents and witness statements, demonstrating that a violation of the regulations did not occur or that a penalty or the amount of the penalty is not warranted by the circumstances.
- (ii) A written request to reduce the proposed civil penalty, the amount of reduction, and the reasons and any documents supporting a reduction of the proposed civil penalty, including records indicating a financial inability to pay or records showing that payment of the proposed civil penalty would prevent the person from continuing in business.
- (iii) A written request for an informal conference to discuss the matter with the agency attorney and to submit relevant information or documents; or
- (3) Request a hearing, in which case a complaint shall be filed with the hearing docket clerk.
- (g) Final notice of proposed civil penalty. A final notice of proposed civil penalty may be issued after participation in informal procedures provided in paragraph (f)(2) of this section or failure to respond in a timely manner to a notice of proposed civil penalty. A final notice of proposed civil penalty will be sent to the individual charged with a

violation, to the president of the corporation or company charged with a violation, or a person previously designated in writing by the individual, corporation, or company to receive documents in that civil penalty action. If not previously done in response to a notice of proposed civil penalty, a corporation or company may designate in writing another person to receive documents in that civil penalty action. The final notice of proposed civil penalty contains a statement of the charges and the amount of the proposed civil penalty and, as a result of information submitted to the agency attorney during informal procedures, may modify an allegation or a proposed civil penalty contained in a notice of proposed civil penalty.

- (1) A final notice of proposed civil penalty may be issued—
- (i) If the person charged with a violation fails to respond to the notice of proposed civil penalty within 30 days after receipt of that notice; or
- (ii) If the parties participated in any informal procedures under paragraph (f)(2) of this section and the parties have not agreed to compromise the action or the agency attorney has not agreed to withdraw the notice of proposed civil penalty.
- (2) Not later than 15 days after receipt of the final notice of proposed civil penalty, the person charged with a violation shall do one of the following—
- (i) Submit the amount of the proposed civil penalty or an agreed-upon amount, in which case either an order assessing civil penalty or a compromise order shall be issued in that amount; or
- (ii) Request a hearing, in which case a complaint shall be filed with the hearing docket clerk.
- (h) Request for a hearing. Any person charged with a violation may request a hearing, pursuant to paragraph (f)(3) or paragraph (g)(2)(ii) of this section, to be conducted in accordance with the procedures in subpart G of this part. A person requesting a hearing shall file a written request for a hearing with the hearing docket clerk, using the appropriate address set forth in §13.210(a) of this part, and shall mail a copy of the

request to the agency attorney. The request for a hearing may be in the form of a letter but must be dated and signed by the person requesting a hearing. The request for a hearing may be typewritten or may be legibly handwritten.

- (i) Hearing. If the person charged with a violation requests a hearing pursuant to paragraph (f)(3) or paragraph (g)(2)(ii) of this section, the original complaint shall be filed with the hearing docket clerk and a copy shall be sent to the person requesting the hearing. The procedural rules in subpart G of this part apply to the hearing and any appeal. At the close of the hearing, the administrative law judge shall issue, either orally on the record or in writing, an initial decision, including the reasons for the decision, that contains findings or conclusions on the allegations contained, and the civil penalty sought, in the com-
- (j) Appeal. Either party may appeal the administrative law judge's initial decision to the FAA decisionmaker pursuant to the procedures in subpart G of this part. If a party files a notice of appeal pursuant to §13.233 of subpart G. the effectiveness of the initial decision is stayed until a final decision and order of the Administrator have been entered on the record. The FAA decisionmaker shall review the record and issue a final decision and order of the Administrator that affirm, modify, or reverse the initial decision. The FAA decisionmaker may assess a civil penalty but shall not assess a civil penalty in an amount greater than that sought in the complaint.
- (k) Payment. A person shall pay a civil penalty by sending a certified check or money order, payable to the Federal Aviation Administration, to the agency attorney.
- (1) Collection of civil penalties. If an individual does not pay a civil penalty imposed by an order assessing civil penalty or other final order, the Administrator may take action provided under the law to collect the penalty.
- (m) A party may seek review only of a final decision and order of the FAA decisionmaker involving a violation of the Federal aviation statute or the Federal hazardous materials transpor-

tation law. Judicial review is in the United States Court of Appeals for the District of Columbia Circuit or the United States court of appeals for the circuit in which the party resides or has the party's principal place of business as provided in §13.235 of this part. Neither an initial decision or an order issued by an administrative law judge that has not been appealed to the FAA decisionmaker, nor an order compromising a civil penalty action, may be appealed under any of those sections.

- (n) Compromise. The FAA may compromise the amount of any civil penalty imposed under this section, under 49 U.S.C. 5123(e), 46031(f), 46303(b), or 46318 at any time before referring the action to the United States Attorney General, or the delegate of the Attorney General, for collection.
- (1) An agency attorney may compromise any civil penalty action where a person charged with a violation agrees to pay a civil penalty and the FAA agrees not to make a finding of violation. Under such agreement, a compromise order is issued following the payment of the agreed-on amount or the signing of a promissory note. The compromise order states the following:
- (i) The person has paid a civil penalty or has signed a promissory note providing for installment payments.
- (ii) The FAA makes no finding of a violation.
- (iii) The compromise order shall not be used as evidence of a prior violation in any subsequent civil penalty proceeding or certificate action proceeding.
- (2) An agency attorney may compromise the amount of a civil penalty proposed in a notice, assessed in an order, or imposed in a compromise order.

[Amdt. 13–32; 70 FR 1813, Jan. 11, 2005; 70 FR 2925, Jan. 18, 2005, as amended at 70 FR 8238, Feb. 18, 2005; 71 FR 70464, Dec. 5, 2006]

§13.17 Seizure of aircraft.

(a) Under section 903 of the Federal Aviation Act of 1958 (49 U.S.C. 1473), a State or Federal law enforcement officer, or a Federal Aviation Administration safety inspector, authorized in an order of seizure issued by the Regional Administrator of the region, or by the

Chief Counsel, may summarily seize an aircraft that is involved in a violation for which a civil penalty may be imposed on its owner or operator.

- (b) Each person seizing an aircraft under this section shall place it in the nearest available and adequate public storage facility in the judicial district in which it was seized.
- (c) The Regional Administrator or Chief Counsel, without delay, sends a written notice and a copy of this section, to the registered owner of the seized aircraft, and to each other persons shown by FAA records to have an interest in it, stating the—
 - (1) Time, date, and place of seizure;
- (2) Name and address of the custodian of the aircraft;
- (3) Reasons for the seizure, including the violations believed, or judicially determined, to have been committed; and
- (4) Amount that may be tendered as—
- (i) A compromise of a civil penalty for the alleged violation; or
- (ii) Payment for a civil penalty imposed by a Federal court for a proven violation.
- (d) The Chief Counsel, or the Regional Counsel or Assistant Chief Counsel for the region or area in which an aircraft is seized under this section, immediately sends a report to the United States District Attorney for the judicial district in which it was seized, requesting the District Attorney to institute proceedings to enforce a lien against the aircraft.
- (e) The Regional Administrator or Chief Counsel directs the release of a seized aircraft whenever—
- (1) The alleged violator pays a civil penalty or an amount agreed upon in compromise, and the costs of seizing, storing, and maintaining the aircraft;
- (2) The aircraft is seized under an order of a Federal Court in proceedings in rem to enforce a lien against the aircraft, or the United States District Attorney for the judicial district concerned notifies the FAA that the District Attorney refuses to institute those proceedings; or
- (3) A bond in the amount and with the sureties prescribed by the Chief Counsel, the Regional Counsel, or the Assistant Chief Counsel is deposited,

conditioned on payment of the penalty, or the compromise amount, and the costs of seizing, storing, and maintaining the aircraft.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46865, Sept. 4, 1997]

§13.18 Civil penalties: Administrative assessment against an individual acting as a pilot, flight engineer, mechanic, or repairman.

- (a) General. (1) This section applies to each action in which the FAA seeks to assess a civil penalty by administrative procedures against an individual acting as a pilot, flight engineer, mechanic, or repairman, under 49 U.S.C. 46301(d)(5), for a violation listed in 49 U.S.C. 46301(d)(2). This section does not apply to a civil penalty assessed for violation of 49 U.S.C. chapter 51, or a rule, regulation, or order issued thereunder.
- (2) District court jurisdiction. Notwithstanding the provisions of paragraph (a)(1) of this section, the United States district courts have exclusive jurisdiction of any civil penalty action involving an individual acting as a pilot, flight engineer, mechanic, or repairman for violations described in that paragraph, under 49 U.S.C. 46301(d)(4), if:
- (i) The amount in controversy is more than \$50,000.
- (ii) The action involves an aircraft subject to a lien that has been seized by the Government; or
- (iii) Another action has been brought for an injunction based on the same violation.
- (b) *Definitions*. As used in this part, the following definitions apply:
- (1) Flight engineer means an individual who holds a flight engineer certificate issued under part 63 of this chapter.
- (2) Individual acting as a pilot, flight engineer, mechanic, or repairman means an individual acting in such capacity, whether or not that individual holds the respective airman certificate issued by the FAA.
- (3) *Mechanic* means an individual who holds a mechanic certificate issued under part 65 of this chapter.

- (4) *Pilot* means an individual who holds a pilot certificate issued under part 61 of this chapter.
- (5) Repairman means an individual who holds a repairman certificate issued under part 65 of this chapter.
- (c) Delegation of authority. (1) The authority of the Administrator under 49 U.S.C. 46301(d)(5), to initiate and assess civil penalties is delegated to the Chief Counsel; the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel; and the Technical Center Counsel.
- (2) The authority of the Administrator to refer cases to the Attorney General of the United States, or the delegate of the Attorney General, for collection of civil penalties is delegated to the Chief Counsel; the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel; and the Technical Center Counsel.
- (3) The authority of the Administrator to compromise the amount of a civil penalty under 49 U.S.C. 46301(f) is delegated to the Chief Counsel; the Deputy Chief Counsel for Operations; the Assistant Chief Counsel for Enforcement; Assistant Chief Counsel, Europe, Africa, and Middle East Area Office; the Regional Counsel; the Aeronautical Center Counsel; and the Technical Center Counsel.
- (d) Notice of proposed assessment. A civil penalty action is initiated by sending a notice of proposed assessment to the individual charged with a violation specified in paragraph (a) of this section. The notice of proposed assessment contains a statement of the charges and the amount of the proposed civil penalty. The individual charged with a violation may do the following:
- (1) Submit the amount of the proposed civil penalty or an agreed-on amount, in which case either an order of assessment or a compromise order will be issued in that amount.
 - (2) Answer the charges in writing.

- (3) Submit a written request for an informal conference to discuss the matter with an agency attorney and submit relevant information or documents
- (4) Request that an order be issued in accordance with the notice of proposed assessment so that the individual charged may appeal to the National Transportation Safety Board.
- (e) Failure to respond to notice of proposed assessment. An order of assessment may be issued if the individual charged with a violation fails to respond to the notice of proposed assessment within 15 days after receipt of that notice.
- (f) Order of assessment. An order of assessment, which assesses a civil penalty, may be issued for a violation described in paragraph (a) of this section after notice and an opportunity to answer any charges and be heard as to why such order should not be issued.
- (g) Appeal. Any individual who receives an order of assessment issued under this section may appeal the order to the National Transportation Safety Board. The appeal stays the effectiveness of the Administrator's order.
- (h) Exhaustion of administrative remedies. An individual substantially affected by an order of the NTSB or the Administrator may petition for review only of a final decision and order of the National Transportation Safety Board to a court of appeals of the United States for the circuit in which the individual charged resides or has his or her principal place of business or the United States Court of Appeals for the District of Columbia Circuit, under 49 U.S.C. 46110 and 46301(d)(6). Neither an order of assessment that has not been appealed to the National Transportation Board, nor an order compromising a civil penalty action, may be appealed under those sections.
- (i) Compromise. The FAA may compromise any civil penalty action initiated under this section, in accordance with 49 U.S.C. 46301(f).
- (1) An agency attorney may compromise any civil penalty action where an individual charged with a violation agrees to pay a civil penalty and the

FAA agrees to make no finding of violation. Under such agreement, a compromise order is issued following the payment of the agreed-on amount or the signing of a promissory note. The compromise order states the following:

- (i) The individual has paid a civil penalty or has signed a promissory note providing for installment payments:
- (ii) The FAA makes no finding of violation; and
- (iii) The compromise order will not be used as evidence of a prior violation in any subsequent civil penalty proceeding or certificate action proceeding.
- (2) An agency attorney may compromise the amount of any civil penalty proposed or assessed in an order.
- (j) Payment. (1) An individual must pay a civil penalty by:
- (i) Sending a certified check or money order, payable to the Federal Aviation Administration, to the FAA office identified in the order of assessment, or
- (ii) Making an electronic funds transfer according to the directions specified in the order of assessment.
- (2) The civil penalty must be paid within 30 days after service of the order of assessment, unless an appeal is filed with the National Transportation Safety Board. The civil penalty must be paid within 30 days after a final order of the Board or the Court of Appeals affirms the order of assessment in whole or in part.
- (k) Collection of civil penalties. If an individual does not pay a civil penalty imposed by an order of assessment or other final order, the Administrator may take action provided under the law to collect the penalty.

[Amdt. 13–32; 69 FR 59497, Oct. 4, 2004]

§13.19 Certificate action.

(a) Under section 609 of the Federal Aviation Act of 1958 (49 U.S.C. 1429), the Administrator may reinspect any civil aircraft, aircraft engine, propeller, appliance, air navigation facility, or air agency, and may re-examine any civil airman. Under section 501(e) of the FA Act, any Certificate of Aircraft Registration may be suspended or revoked by the Administrator for any cause

that renders the aircraft ineligible for registration.

- (b) If, as a result of such a reinspection re-examination, or other investigation made by the Administrator under section 609 of the FA Act, the Administrator determines that the public interest and safety in air commerce requires it, the Administrator may issue an order amending, suspending, or revoking, all or part of any type certificate, production certificate, airworthiness certificate, airman certificate, air carrier operating certificate, air navigation facility certificate, or air agency certificate. This authority may be exercised for remedial purposes in cases involving the Hazardous Materials Transportation Act (49) U.S.C. 1801 et seq.) or regulations issued under that Act. This authority is also exercised by the Chief Counsel, the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Regulations, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, each Regional Counsel, and the Aeronautical Center Counsel. If the Administrator finds that any aircraft registered under Part 47 of this chapter is ineligible for registration, the Administrator issues an order suspending or revoking that certificate. This authority as to aircraft found ineligible for registration is also exercised by each Regional Counsel, the Aeronautical Center Counsel, and the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office.
- (c) Before issuing an order under paragraph (b) of this section, the Chief Counsel, the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Regulations, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, each Regional Counsel, or the Aeronautical Center Counsel advises the certificate holder of the charges or other reasons upon which the Administrator bases the proposed action and, except in an emergency, allows the holder to answer any charges and to be heard as to why the certificate should not be amended, suspended, or revoked. The holder may, by checking the appropriate box on the form that is sent to the holder with the notice of proposed certificate action, elect to-

- (1) Admit the charges and surrender his or her certificate;
 - (2) Answer the charges in writing:
- (3) Request that an order be issued in accordance with the notice of proposed certificate action so that the certificate holder may appeal to the National Transportation Safety Board, if the charges concerning a matter under Title VI of the FA Act;
- (4) Request an opportunity to be heard in an informal conference with the FAA counsel; or
- (5) Request a hearing in accordance with Subpart D of this part if the charges concern a matter under Title V of the FA Act.

Except as provided in §13.35(b), unless the certificate holder returns the form and, where required, an answer or motion, with a postmark of not later than 15 days after the date of receipt of the notice, the order of the Administrator is issued as proposed. If the certificate holder has requested an informal conference with the FAA counsel and the charges concern a matter under Title V of the FA Act, the holder may after that conference also request a formal hearing in writing with a postmark of not later than 10 days after the close of the conference. After considering any information submitted by the certificate holder, the Chief Counsel, the Assistant Chief Counsel for Regulations and Enforcement, the Regional Counsel concerned, or the Aeronautical Center Counsel (as to matters under Title V of the FA Act) issues the order of the Administrator, except that if the holder has made a valid request for a formal hearing on a matter under Title V of the FA Act initially or after an informal conference, Subpart D of this part governs further proceedings.

(d) Any person whose certificate is affected by an order issued under this section may appeal to the National Transportation Safety Board. If the certificate holder files an appeal with the Board, the Administrator's order is stayed unless the Administrator advises the Board that an emergency exists and safety in air commerce requires that the order become effective immediately. If the Board is so advised, the order remains effective and the Board shall finally dispose of the appeal within 60 days after the date of

the advice. This paragraph does not apply to any person whose Certificate of Aircraft Registration is affected by an order issued under this section.

[Doc. No. 13–14, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–15, 45 FR 20773, Mar. 31, 1980; Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46865, Sept. 4, 1997; 75 FR 41979, July 20, 2010]

§ 13.20 Orders of compliance, cease and desist orders, orders of denial, and other orders.

- (a) This section applies to orders of compliance, cease and desist orders, orders of denial, and other orders issued by the Administrator to carry out the provisions of the Federal Aviation Act of 1958, as amended, the Hazardous Materials Transportation Act, the Airport and Airway Development Act of 1970, and the Airport and Airway Improvement Act of 1982, or the Airport and Airway Improvement Act of 1982 as amended by the Airport and Airway Safety and Capacity Expansion Act of 1987. This section does not apply to orders issued pursuant to section 602 or section 609 of the Federal Aviation Act of 1958, as amended.
- (b) Unless the Administrator determines that an emergency exists and safety in air commerce requires the immediate issuance of an order under this section, the person subject to the order shall be provided with notice prior to issuance.
- (c) Within 30 days after service of the notice, the person subject to the order may"
- (1) Request an opportunity to be heard in an informal conference with an FAA attorney;
 - (2) Reply in writing; or
- (3) Request a hearing in accordance with subpart D of this part.
- (d) If an informal conference is held or a reply is filed, as to any charges not withdrawn or not subject to a consent order, the person subject to the order may, within 10 days after receipt of notice that the remaining charges are not withdrawn, request a hearing in accordance with subpart D of this part.
- (e) Failure to request a hearing within the period provided in paragraphs (c) or (d) of this section—

- (1) Constitutes a waiver of the right to appeal and the right to a hearing, and
- (2) Authorizes the official who issued the notice to find the facts to be as alleged in the notice, or as modified as the official may determine necessary based on any written response, and to issue an appropriate order, without further notice or proceedings.
- (f) If a hearing is requested in accordance with paragraph (c) or (d) of this section, the procedure of Subpart D of this part applies. At the close of the hearing, the Hearing Officer, on the record or subsequently in writing, shall set forth findings and conclusions and the reasons therefor, and either—
 - (1) Dismiss the notice; or
 - (2) Issue an order.
- (g) Any party to the hearing may appeal from the order of the Hearing Officer by filing a notice of appeal with the Administrator within 20 days after the date of issuance of the order.
- (h) If a notice of appeal is not filed from the order issued by a Hearing Officer, such order is the final agency order.
- (i) Any person filing an appeal authorized by paragraph (g) of this section shall file an appeal brief with the Administrator within 40 days after the date of issuance of the order, and serve a copy on the other party. A reply brief must be filed within 20 days after service of the appeal brief and a copy served on the appellant.
- (j) On appeal the Administrator reviews the available record of the proceeding, and issues an order dismissing, reversing, modifying or affirming the order. The Administrator's order includes the reasons for the Administrator's action.
- (k) For good cause shown, requests for extensions of time to file any document under this section may be granted by—
- (1) The official who issued the order, if the request is filed prior to the designation of a Hearing Officer; or
- (2) The Hearing Officer, if the request is filed prior to the filing of a notice of appeal; or
- (3) The Administrator, if the request is filed after the filing of a notice of appeal.

- (1) Except in the case of an appeal from the decision of a Hearing Officer, the authority of the Administrator under this section is also exercised by the Chief Counsel, Deputy Chief Counsel, each Assistant Chief Counsel, each Regional Counsel, and the Aeronautical Center Counsel (as to matters under Title V of the Federal Aviation Act of 1958).
- (m) Filing and service of documents under this section shall be accomplished in accordance with §13.43; and the periods of time specified in this section shall be computed in accordance with §13.44.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–17, 53 FR 33783, Aug. 31, 1988; Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46865, Sept. 4, 1997; Amdt. 13–36, 79 FR 46967, Aug. 12, 2014]

§13.21 Military personnel.

If a report made under this part indicates that, while performing official duties, a member of the Armed Forces, or a civilian employee of the Department of Defense who is subject to the Uniform Code of Military Justice (10 U.S.C. Ch. 47), has violated the Federal Aviation Act of 1958, or a regulation or order issued under it, the Chief Counsel, the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Regulations, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, each Regional Counsel, and the Aeronautical Center Counsel send a copy of the report to the appropriate military authority for such disciplinary action as that authority considers appropriate and a report to the Administrator thereon.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46866, Sept. 4, 1997]

§13.23 Criminal penalties.

(a) Sections 902 and 1203 of the Federal Aviation Act of 1958 (49 U.S.C. 1472 and 1523), provide criminal penalties for any person who knowingly and willfully violates specified provisions of that Act, or any regulation or order issued under those provisions. Section 110(b) of the Hazardous Materials Transportation Act (49 U.S.C. 1809(b)) provides for a criminal penalty of a

fine of not more than \$25,000, imprisonment for not more than five years, or both, for any person who willfully violates a provision of that Act or a regulation or order issued under it.

(b) If an inspector or other employee of the FAA becomes aware of a possible violation of any criminal provision of the Federal Aviation Act of 1958 (except a violation of section 902 (i) through (m) which is reported directly to the Federal Bureau of Investigation), or of the Hazardous Materials Transportation Act, relating to the transportation or shipment by air of hazardous materials, he or she shall report it to the Office of the Chief Counsel or the Regional Counsel or Assistant Chief Counsel for the region or area concerned. If appropriate, that office refers the report to the Department of Justice for criminal prosecution of the offender. If such an inspector or other employee becomes aware of a possible violation of a Federal statute that is within the investigatory jurisdiction of another Federal agency, he or she shall immediately report it to that agency according to standard FAA practices.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46866, Sept. 4, 1997]

§ 13.25 Injunctions.

(a) Whenever it is determined that a person has engaged, or is about to engage, in any act or practice constituting a violation of the Federal Aviation Act of 1958, or any regulation or order issued under it for which the FAA exercises enforcement responsibility, or, with respect to the transportation or shipment by air of any hazardous materials, in any act or practice constituting a violation of the Hazardous Materials Transportation Act, or any regulation or order issued under it for which the FAA exercises enforcement responsibility, the Chief Counsel, the Assistant Chief Counsel, Enforcement, the Assistant Counsel, Regulations, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, each Regional Counsel, and the Aeronautical Center Counsel may request the United States Attorney General, or the delegate of the Attorney General, to bring an action in the appropriate United States District Court for such relief as is necessary or appropriate, including mandatory or prohibitive injunctive relief, interim equitable relief, and punitive damages, as provided by section 1007 of the Federal Aviation Act of 1958 (49 U.S.C. 1487) and section 111(a) of the Hazardous Materials Transportation Act (49 U.S.C. 1810).

(b) Whenever it is determined that there is substantial likelihood that death, serious illness, or severe personal injury, will result from the transportation by air of a particular hazardous material before an order of compliance proceeding, or other administrative hearing or formal proceeding to abate the risk of the harm can be completed, the Chief Counsel, the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Regulations, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, each Regional Counsel, and the Aeronautical Center Counsel may bring, or request the United States Attorney General to bring, an action in the appropriate United States District Court for an order suspending or restricting the transportation by air of the hazardous material or for such other order as is necessary to eliminate or ameliorate the imminent hazard, as provided by section 111(b) of the Hazardous Materials Transportation Act (49 U.S.C. 1810)

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46866, Sept. 4, 1997]

§ 13.27 Final order of Hearing Officer in certificate of aircraft registration proceedings.

(a) If, in proceedings under section 501(b) of the Federal Aviation Act of 1958 (49 U.S.C. 1401), the Hearing Officer determines that the aircraft is ineligible for a Certificate of Aircraft Registration, the Hearing Officer shall suspend or revoke the respondent's certificate, as proposed in the notice of proposed certificate action.

(b) If the final order of the Hearing Officer makes a decision on the merits, it shall contain a statement of the findings and conclusions of law on all material issues of fact and law. If the Hearing Officer finds that the allegations of the notice have been proven, but that no sanction is required, the Hearing Officer shall make appropriate findings and issue an order terminating the notice. If the Hearing Officer finds that the allegations of the notice have not been proven, the Hearing Officer shall issue an order dismissing the notice. If the Hearing Officer finds it to be equitable and in the public interest, the Hearing Officer shall issue an order terminating the proceeding upon payment by the respondent of a civil penalty in an amount agreed upon by the parties.

(c) If the order is issued in writing, it shall be served upon the parties.

[Doc. No. 13–14, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–15, 45 FR 20773, Mar. 31, 1980; Amdt. 13–34, 75 FR 41979, July 20, 2010]

§ 13.29 Civil penalties: Streamlined enforcement procedures for certain security violations.

This section may be used, at the agency's discretion, in enforcement actions involving individuals presenting dangerous or deadly weapons for screening at airports or in checked baggage where the amount of the proposed civil penalty is less than \$5,000. In these cases, sections 13.16(a), 13.16(c), and 13.16 (f) through (l) of this chapter are used, as well as paragraphs (a) through (d) of this section:

(a) Delegation of authority. The authority of the Administrator, under 49 U.S.C. 46301, to initiate the assessment of civil penalties for a violation of 49 U.S.C. Subtitle VII, or a rule, regulation, or order issued thereunder, is delegated to the regional Civil Aviation Security Division Manager and the regional Civil Aviation Security Deputy Division Manager for the purpose of issuing notices of violation in cases involving violations of 49 U.S.C. Subtitle VII and the FAA's regulations by individuals presenting dangerous or deadly weapons for screening at airport checkpoints or in checked baggage. This authority may not be delegated below the level of the regional Civil Aviation Security Deputy Division Manager.

(b) *Notice of violation*. A civil penalty action is initiated by sending a notice of violation to the person charged with the violation. The notice of violation

contains a statement of the charges and the amount of the proposed civil penalty. Not later than 30 days after receipt of the notice of violation, the person charged with a violation shall:

- (1) Submit the amount of the proposed civil penalty or an agreed-upon amount, in which case either an order assessing a civil penalty or a compromise order shall be issued in that amount; or
- (2) Submit to the agency attorney identified in the material accompanying the notice any of the following:
- (i) Written information, including documents and witness statements, demonstrating that a violation of the regulations did not occur or that a penalty or the penalty amount is not warranted by the circumstances; or
- (ii) A written request to reduce the proposed civil penalty, the amount of reduction, and the reasons and any documents supporting a reduction of the proposed civil penalty, including records indicating a financial inability to pay or records showing that payment of the proposed civil penalty would prevent the person from continuing in business; or
- (iii) A written request for an informal conference to discuss the matter with an agency attorney and submit relevant information or documents; or
- (3) Request a hearing in which case a complaint shall be filed with the hearing docket clerk.
- (c) Final notice of violation and civil penalty assessment order. A final notice of violation and civil penalty assessment order ("final notice and order") may be issued after participation in any informal proceedings as provided in paragraph (b)(2) of this section, or after failure of the respondent to respond in a timely manner to a notice of violation. A final notice and order will be sent to the individual charged with a violation. The final notice and order will contain a statement of the charges and the amount of the proposed civil penalty and, as a result of information submitted to the agency attorney during any informal procedures, may reflect a modified allegation or proposed civil penalty.

A final notice and order may be issued—

- (1) If the person charged with a violation fails to respond to the notice of violation within 30 days after receipt of that notice: or
- (2) If the parties participated in any informal procedures under paragraph (b)(2) of this section and the parties have not agreed to compromise the action or the agency attorney has not agreed to withdraw the notice of violation.
- (d) Order assessing civil penalty. An order assessing civil penalty may be issued after notice and opportunity for a hearing. A person charged with a violation may be subject to an order assessing civil penalty in the following circumstances:
- (1) An order assessing civil penalty may be issued if a person charged with a violation submits, or agrees to submit, the amount of civil penalty proposed in the notice of violation.
- (2) An order assessing civil penalty may be issued if a person charged with a violation submits, or agrees to submit, an agreed-upon amount of civil penalty that is not reflected in either the notice of violation or the final notice and order.
- (3) The final notice and order becomes (and contains a statement so indicating) an order assessing a civil penalty when the person charged with a violation submits the amount of the proposed civil penalty that is reflected in the final notice and order.
- (4) The final notice and order becomes (and contains a statement so indicating) an order assessing a civil penalty 16 days after receipt of the final notice and order, *unless* not later than 15 days after receipt of the final notice and order, the person charged with a violation does one of the following—
- (i) Submits an agreed-upon amount of civil penalty that is not reflected in the final notice and order, in which case an order assessing civil penalty or a compromise order shall be issued in that amount; or
- (ii) Requests a hearing in which case a complaint shall be filed with the hearing docket clerk.
- (5) Unless an appeal is filed with the FAA decisionmaker in a timely manner, an initial decision or order of an administrative law judge shall be considered an order assessing civil penalty

if an administrative law judge finds that an alleged violation occurred and determines that a civil penalty, in an amount found to be appropriate by the administrative law judge, is warranted.

(6) Unless a petition for review is filed with a U.S. Court of Appeals in a timely manner, a final decision and order of the Administrator shall be considered an order assessing civil penalty if the FAA decisionmaker finds that an alleged violation occurred and a civil penalty is warranted.

[Doc. No. 27873, 61 FR 44155, Aug. 28, 1996]

Subpart D—Rules of Practice for FAA Hearings

§13.31 Applicability.

This subpart applies to proceedings in which a hearing has been requested in accordance with §§ 13.19(c)(5), 13.20(c), 13.20(d), 13.75(a)(2), 13.75(b), or 13.81(e).

[Amdt. 13–18, 53 FR 34655, Sept. 7, 1988]

§13.33 Appearances.

Any party to a proceeding under this subpart may appear and be heard in person or by attorney.

§13.35 Request for hearing.

- (a) A request for hearing must be made in writing and filed in the Hearing Docket.
- (1) If delivery is in person, or by expedited courier service. A person delivering the request for hearing in person or sending the request for hearing by commercial expedited courier (for example, Federal Express or United Parcel Service), should use the following address: Federal Aviation Administration, 600 Independence Avenue, SW., Wilbur Wright Building—Suite 2W1000, Washington, DC 20591; Attention: Hearing Docket Clerk, AGC-430.
- (2) If delivery is by U.S. Mail: If the request for hearing is sent by U.S. Mail, then it should be addressed as follows: Federal Aviation Administration, 800 Independence Avenue, SW., Washington, DC 20591: Attention: Hearing Docket Clerk, AGC-430, Wilbur Wright Building—Suite 2W1000.
- (b) The request for hearing must describe the action proposed by the FAA, and must contain a statement that a

hearing is requested. A copy of the request for hearing and a copy of the answer required by paragraph (c) of this section must be served on the official who issued the notice of proposed action.

- (c) An answer to the notice of proposed action must be filed with the request for hearing. All allegations in the notice not specifically denied in the answer are deemed admitted.
- (d) Within 15 days after service of the copy of the request for hearing, the official who issued the notice of proposed action forwards a copy of that notice, which serves as the complaint, to the Hearing Docket.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; 71 FR 70464, Dec. 5, 2006]

§ 13.37 Hearing Officer's powers.

Any Hearing Officer may—

- (a) Give notice concerning, and hold, prehearing conferences and hearings;
- (b) Administrator oaths and affirmations;
 - (c) Examine witnesses;
- (d) Adopt procedures for the submission of evidence in written form;
- (e) Issue subpoenas and take depositions or cause them to be taken;
 - (f) Rule on offers of proof;
 - (g) Receive evidence;
- (h) Regulate the course of the hearing:
- (i) Hold conferences, before and during the hearing, to settle and simplify issues by consent of the parties;
- (j) Dispose of procedural requests and similar matters; and
- (k) Issue decisions, make findings of fact, make assessments, and issue orders, as appropriate.

§ 13.39 Disqualification of Hearing Officer.

If disqualified for any reason, the Hearing Officer shall withdraw from the case.

§13.41 [Reserved]

§ 13.43 Service and filing of pleadings, motions, and documents.

(a) Copies of all pleadings, motions, and documents filed with the Hearing Docket must be served upon all parties

to the proceedings by the person filing them.

- (b) Service may be made by personal delivery or by mail.
- (c) A certificate of service shall accompany all documents when they are tendered for filing and shall consist of a certificate of personal delivery or a certificate of mailing, executed by the person making the personal delivery or mailing the document.
- (d) Whenever proof of service by mail is made, the date of mailing or the date as shown on the postmark shall be the date of service, and where personal service is made, the date of personal delivery shall be the date of service.
- (e) The date of filing is the date the document is actually received.

§ 13.44 Computation of time and extension of time.

- (a) In computing any period of time prescribed or allowed by this subpart, the date of the act, event, default, notice or order after which the designated period of time begins to run is not to be included in the computation. The last day of the period so computed is to be included unless it is a Saturday, Sunday, or legal holiday for the FAA, in which event the period runs until the end of the next day which is neither a Saturday, Sunday nor a legal holiday.
- (b) Upon written request filed with the Hearing Docket and served upon all parties, and for good cause shown, a Hearing Officer may grant an extension of time to file any documents specified in this subpart.

§ 13.45 Amendment of notice and answer.

At any time more than 10 days before the date of hearing, any party may amend his or her notice, answer, or other pleading, by filing the amendment with the Hearing Officer and serving a copy of it on each other party. After that time, amendments may be allowed only in the discretion of the Hearing Officer. If an amendment to an initial pleading has been allowed, the Hearing Officer shall allow the other parties a reasonable opportunity to answer.

§ 13.47 Withdrawal of notice or request for hearing.

At any time before the hearing, the FAA counsel may withdraw the notice of proposed action, and the party requesting the hearing may withdraw the request for hearing.

§13.49 Motions.

- (a) Motion to dismiss for insufficiency. A respondent who requests a formal hearing may, in place of an answer, file a motion to dismiss for failure of the allegations in the notice of proposed action to state a violation of the FA Act or of this chapter or to show lack of qualification of the respondent. If the Hearing Officer denies the motion, the respondent shall file an answer within 10 days.
 - (b) [Reserved]
- (c) Motion for more definite statement. The certificate holder may, in place of an answer, file a motion that the allegations in the notice be made more definite and certain. If the Hearing Officer grants the motion, the FAA counsel shall comply within 10 days after the date it is granted. If the Hearing Officer denies the motion the certificate holder shall file an answer within 10 days after the date it is denied.
- (d) Motion for judgment on the pleadings. After the pleadings are closed, either party may move for a judgment on the pleadings.
- (e) Motion to strike. Upon motion of either party, the Hearing Officer may order stricken, from any pleadings, any insufficient allegation or defense, or any immaterial, impertinent, or scandalous matter.
- (f) Motion for production of documents. Upon motion of any party showing good cause, the Hearing Officer may, in the manner provided by Rule 34, Federal Rules of Civil Procedure, order any party to produce any designated document, paper, book, account, letter, photograph, object, or other tangible thing, that is not privileged, that constitutes or contains evidence relevant to the subject matter of the hearings, and that is in the party's possession, custody, or control.
- (g) Consolidation of motions. A party who makes a motion under this section shall join with it all other motions that are then available to the party.

Any objection that is not so raised is considered to be waived.

(h) Answers to motions. Any party may file an answer to any motion under this section within 5 days after service of the motion.

§13.51 Intervention.

Any person may move for leave to intervene in a proceeding and may become a party thereto, if the Hearing Officer, after the case is sent to the Hearing Officer for hearing, finds that the person may be bound by the order to be issued in the proceedings or has a property or financial interest that may not be adequately represented by existing parties, and that the intervention will not unduly broaden the issues or delay the proceedings. Except for good cause shown, a motion for leave to intervene may not be considered if it is filed less than 10 days before the hearing.

§13.53 Depositions.

After the respondent has filed a request for hearing and an answer, either party may take testimony by deposition in accordance with section 1004 of the Federal Aviation Act of 1958 (49 U.S.C. 1484) or Rule 26, Federal Rules of Civil Procedure.

§ 13.55 Notice of hearing.

The Hearing Officer shall set a reasonable date, time, and place for the hearing, and shall give the parties adequate notice thereof and of the nature of the hearing. Due regard shall be given to the convenience of the parties with respect to the place of the hearing.

§13.57 Subpoenas and witness fees.

(a) The Hearing Officer to whom a case is assigned may, upon application by any party to the proceeding, issue subpoenas requiring the attendance of witnesses or the production of documentary or tangible evidence at a hearing or for the purpose of taking depositions. However, the application for producing evidence must show its general relevance and reasonable scope. This paragraph does not apply to the attendance of FAA employees or to the production of documentary evidence in

the custody of such an employee at a hearing.

- (b) A person who applies for the production of a document in the custody of an FAA employee must follow the procedure in §13.49(f). A person who applies for the attendance of an FAA employee must send the application, in writing, to the Hearing Officer setting forth the need for that employee's attendance.
- (c) A witness in a proceeding under this subpart is entitled to the same fees and mileage as is paid to a witness in a court of the United States under comparable circumstances. The party at whose instance the witness is subpoenaed or appears shall pay the witness fees.
- (d) Notwithstanding the provisions of paragraph (c) of this section, the FAA pays the witness fees and mileage if the Hearing Officer who issued the subpoena determines, on the basis of a written request and good cause shown, that—
- (1) The presence of the witness will materially advance the proceeding; and
- (2) The party at whose instance the witness is subpoenaed would suffer a serious hardship if required to pay the witness fees and mileage.

§13.59 Evidence.

- (a) Each party to a hearing may present the party's case or defense by oral or documentary evidence, submit evidence in rebuttal, and conduct such cross-examination as may be needed for a full disclosure of the facts.
- (b) Except with respect to affirmative defenses and orders of denial, the burden of proof is upon the FAA counsel.
- (c) The Hearing Officer may order information contained in any report or document filed or in any testimony given pursuant to this subpart withheld from public disclosure when, in the judgment of the Hearing Officer, disclosure would adversely affect the interests of any person and is not required in the public interest or is not otherwise required by statute to be made available to the public. Any person may make written objection to the public disclosure of such information, stating the ground for such objection.

§13.61 Argument and submittals.

The Hearing Officer shall give the parties adequate opportunity to present arguments in support of motions, objections, and the final order. The Hearing Officer may determine whether arguments are to be oral or written. At the end of the hearing the Hearing Officer may, in the discretion of the Hearing Officer, allow each party to submit written proposed findings and conclusions and supporting reasons for them.

§ 13.63 Record.

The testimony and exhibits presented at a hearing, together with all papers, requests, and rulings filed in the proceedings are the exclusive basis for the issuance of an order. Either party may obtain a transcript from the official reporter upon payment of the fees fixed therefor.

Subpart E—Orders of Compliance Under the Hazardous Materials Transportation Act

§13.71 Applicability.

Whenever the Chief Counsel, the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, or a Regional Counsel has reason to believe that a person is engaging in the transportation or shipment by air of hazardous materials in violation of the Hazardous Materials Transportation Act, or any regulation or order issued under it for which the FAA exercises enforcement responsibility, and the circumstances do not require the issuance of an order of immediate compliance, he may conduct proceedings pursuant to section 109 of that Act (49 U.S.C. 1808) to determine the nature and extent of the violation, and may thereafter issue an order directing compliance.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46866, Sept. 4, 1997]

§ 13.73 Notice of proposed order of compliance.

A compliance order proceeding commences when the Chief Counsel, the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, or a Regional Counsel sends the alleged violator a notice of proposed order of compliance advising the alleged violator of the charges and setting forth the remedial action sought in the form of a proposed order of compliance.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46866, Sept. 4, 1997)

§ 13.75 Reply or request for hearing.

- (a) Within 30 days after service upon the alleged violator of a notice of proposed order of compliance, the alleged violator may—
- (1) File a reply in writing with the official who issued the notice; or
- (2) Request a hearing in accordance with subpart D of this part.
- (b) If a reply is filed, as to any charges not dismissed or not subject to a consent order of compliance, the alleged violator may, within 10 days after receipt of notice that the remaining charges are not dismissed, request a hearing in accordance with subpart D of this part.
- (c) Failure of the alleged violator to file a reply or request a hearing within the period provided in paragraph (a) or (b) of this section—
- (1) Constitutes a waiver of the right to a hearing and the right to an appeal, and
- (2) Authorizes the official who issued the notice to find the facts to be as alleged in the notice and to issue an appropriate order directing compliance, without further notice or proceedings.

$\S 13.77$ Consent order of compliance.

- (a) At any time before the issuance of an order of compliance, the official who issued the notice and the alleged violator may agree to dispose of the case by the issuance of a consent order of compliance by the official.
- (b) A proposal for a consent order submitted to the official who issued the notice under this section must include—

- (1) A proposed order of compliance;
- (2) An admission of all jurisdictional facts:
- (3) An express waiver of right to further procedural steps and of all rights to judicial review;
- (4) An incorporation by reference of the notice and an acknowledgement that the notice may be used to construe the terms of the order of compliance; and
- (5) If the issuance of a consent order has been agreed upon after the filing of a request for hearing in accordance with subpart D of this part, the proposal for a consent order shall include a request to be filed with the Hearing Officer withdrawing the request for a hearing and requesting that the case be dismissed.

§13.79 Hearing.

If an alleged violator requests a hearing in accordance with §13.75, the procedure of subpart D of this part applies. At the close of the hearing, the Hearing Officer, on the record or subsequently in writing, sets forth the Hearing Officer's findings and conclusion and the reasons therefor, and either—

- (a) Dismisses the notice of proposed order of compliance; or
 - (b) Issues an order of compliance.

§ 13.81 Order of immediate compliance.

- (a) Notwithstanding §§ 13.73 through 13.79, the Chief Counsel, the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, or a Regional Counsel may issue an order of immediate compliance, which is effective upon issuance, if the person who issues the order finds that—
- (1) There is strong probability that a violation is occurring or is about to occur:
- (2) The violation poses a substantial risk to health or to safety of life or property; and
- (3) The public interest requires the avoidance or amelioration of that risk through immediate compliance and waiver of the procedures afforded under §§ 13.73 through 13.79.
- (b) An order of immediate compliance is served promptly upon the person against whom the order is issued

by telephone or telegram, and a written statement of the relevant facts and the legal basis for the order, including the findings required by paragraph (a) of this section, is served promptly by personal service or by mail.

- (c) The official who issued the order of immediate compliance may rescind or suspend the order if it appears that the criteria set forth in paragraph (a) of this section are no longer satisfied, and, when appropriate, may issue a notice of proposed order of compliance under §13.73 in lieu thereof.
- (d) If at any time in the course of a proceeding commenced in accordance with §13.73 the criteria set forth in paragraph (a) of this section are satisfied, the official who issued the notice may issue an order of immediate compliance, even if the period for filing a reply or requesting a hearing specified in §13.75 has not expired.
- (e) Within three days after receipt of service of an order of immediate compliance, the alleged violator may request a hearing in accordance with Subpart D of this part and the procedure in that subpart will apply except that—
- (1) The case will be heard within fifteen days after the date of the order of immediate compliance unless the alleged violator requests a later date;
- (2) The order will serve as the complaint; and
- (3) The Hearing Officer shall issue his decision and order dismissing, reversing, modifying, or affirming the order of immediate compliance on the record at the close of the hearing.
- (f) The filing of a request for hearing in accordance with paragraph (e) of this section does not stay the effectiveness of an order of immediate compliance.
- (g) At any time after an order of immediate compliance has become effective, the official who issued the order may request the United States Attorney General, or the delegate of the Attorney General, to bring an action for appropriate relief in accordance with §13.25.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 13–19, 54 FR 39290, Sept. 25, 1989; Amdt. 13–29, 62 FR 46866, Sept. 4, 19071

§13.83 Appeal.

- (a) Any party to the hearing may appeal from the order of the Hearing Officer by filing a notice of appeal with the Administrator within 20 days after the date of issuance of the order.
- (b) Any person against whom an order of immediate compliance has been issued in accordance with §13.81 or the official who issued the order of immediate compliance may appeal from the order of the Hearing Officer by filing a notice of appeal with the Administrator within three days after the date of issuance of the order by the Hearing Officer.
- (c) Unless the Administrator expressly so provides, the filing of a notice of appeal does not stay the effectiveness of an order of immediate compliance.
- (d) If a notice of appeal is not filed from the order of compliance issued by a Hearing Officer, such order is the final agency order of compliance.
- (e) Any person filing an appeal authorized by paragraph (a) of this section shall file an appeal brief with the Administrator within 40 days after the date of the issuance of the order, and serve a copy on the other party. Any reply brief must be filed within 20 days after service of the appeal brief. A copy of the reply brief must be served on the appellant.
- (f) Any person filing an appeal authorized by paragraph (b) of this section shall file an appeal brief with the Administrator with the notice of appeal and serve a copy on the other party. Any reply brief must be filed within 3 days after receipt of the appeal brief. A copy of the reply brief must be served on the appellant.
- (g) On appeal the Administrator reviews the available record of the proceeding, and issues an order dismissing, reversing, modifying or affirming the order of compliance or the order of immediate compliance. The Administrator's order includes the reasons for the action.
- (h) In cases involving an order of immediate compliance, the Administrator's order on appeal is issued within ten days after the filing of the notice of appeal.

§ 13.85 Filing, service and computation of time.

Filing and service of documents under this subpart shall be accomplished in accordance with §13.43 except service of orders of immediate compliance under §13.81(b); and the periods of time specified in this subpart shall be computed in accordance with §13.44.

§13.87 Extension of time.

- (a) The official who issued the notice of proposed order of compliance, for good cause shown, may grant an extension of time to file any document specified in this subpart, except documents to be filed with the Administrator.
- (b) Extensions of time to file documents with the Administrator may be granted by the Administrator upon written request, served upon all parties, and for good cause shown.

Subpart F—Formal Fact-Finding Investigation Under an Order of Investigation

§13.101 Applicability.

- (a) This subpart applies to fact-finding investigations in which an order of investigation has been issued under §13.3(c) or §13.5(i) of this part.
- (b) This subpart does not limit the authority of duly designated persons to issue subpoenas, administer oaths, examine witnesses and receive evidence in any informal investigation as provided for in sections 313 and 1004(a) of the Federal Aviation Act (49 U.S.C. 1354 and 1484(a)) and section 109(a) of the Hazardous Materials Transportation Act (49 U.S.C. 1808(a)).

§13.103 Order of investigation.

The order of investigation—

- (a) Defines the scope of the investigation by describing the information sought in terms of its subject matter or its relevancy to specified FAA functions;
- (b) Sets forth the form of the investigation which may be either by individual deposition or investigative proceeding or both; and
- (c) Names the official who is authorized to conduct the investigation and serve as the Presiding Officer.

§13.105 Notification.

Any person under investigation and any person required to testify and produce documentary or physical evidence during the investigation will be advised of the purpose of the investigation, and of the place where the investigative proceeding or deposition will be convened. This may be accomplished by a notice of investigation or by a subpoena. A copy of the order of investigation may be sent to such persons, when appropriate.

$\S 13.107$ Designation of additional parties.

- (a) The Presiding Officer may designate additional persons as parties to the investigation, if in the discretion of the Presiding Officer, it will aid in the conduct of the investigation.
- (b) The Presiding Officer may designate any person as a party to the investigation if that person—
- (1) Petitions the Presiding Officer to participate as a party; and
- (2) Is so situated that the disposition of the investigation may as a practical matter impair the ability to protect that person's interest unless allowed to participate as a party, and
- (3) Is not adequately represented by existing parties.

§13.109 Convening the investigation.

The investigation shall be conducted at such place or places designated by the Presiding Officer, and as convenient to the parties involved as expeditious and efficient handling of the investigation permits.

§13.111 Subpoenas.

- (a) Upon motion of the Presiding Officer, or upon the request of a party to the investigation, the Presiding Officer may issue a subpoena directing any person to appear at a designated time and place to testify or to produce documentary or physical evidence relating to any matter under investigation.
- (b) Subpoenas shall be served by personal service, or upon an agent designated in writing for the purpose, or by registered or certified mail addressed to such person or agent. Whenever service is made by registered or certified mail, the date of mailing shall

be considered as the time when service is made.

(c) Subpoenas shall extend in jurisdiction throughout the United States or any territory or possession thereof.

§13.113 Noncompliance with the investigative process.

If any person fails to comply with the provisions of this subpart or with any subpoena or order issued by the Presiding Officer or the designee of the Presiding Officer, judicial enforcement may be initiated against that person under applicable statutes.

§13.115 Public proceedings.

- (a) All investigative proceedings and depositions shall be public unless the Presiding Officer determines that the public interest requires otherwise.
- (b) The Presiding Officer may order information contained in any report or document filed or in any testimony given pursuant to this subpart withheld from public disclosure when, in the judgment of the Presiding Officer, disclosure would adversely affect the interests of any person and is not required in the public interest or is not otherwise required by statute to be made available to the public. Any person may make written objection to the public disclosure of such information, stating the grounds for such objection.

§13.117 Conduct of investigative proceeding or deposition.

- (a) The Presiding Officer or the designee of the Presiding Officer may question witnesses.
- (b) Any witness may be accompanied by counsel.
- (c) Any party may be accompanied by counsel and either the party or counsel may—
- (1) Question witnesses, provided the questions are relevant and material to the matters under investigation and would not unduly impede the progress of the investigation; and
- (2) Make objections on the record and argue the basis for such objections.
- (d) Copies of all notices or written communications sent to a party or witness shall upon request be sent to that person's attorney of record.

§ 13.119 Rights of persons against selfincrimination.

- (a) Whenever a person refuses, on the basis of a privilege against self-incrimination, to testify or provide other information during the course of any investigation conducted under this subpart, the Presiding Officer may, with the approval of the Attorney General of the United States, issue an order requiring the person to give testimony or provide other information. However, no testimony or other information so compelled (or any information directly or indirectly derived from such testimony or other information) may be used against the person in any criminal case, except in a prosecution for perjury, giving a false statement, or otherwise failing to comply with the order.
- (b) The Presiding Officer may issue an order under this section if—
- (1) The testimony or other information from the witness may be necessary to the public interest; and
- (2) The witness has refused or is likely to refuse to testify or provide other information on the basis of a privilege against self-incrimination.
- (c) Immunity provided by this section will not become effective until the person has refused to testify or provide other information on the basis of a privilege against self-incrimination, and an order under this section has been issued. An order, however, may be issued prospectively to become effective in the event of a claim of the privilege.

§13.121 Witness fees.

All witnesses appearing shall be compensated at the same rate as a witness appearing before a United States District Court.

§ 13.123 Submission by party to the investigation.

- (a) During an investigation conducted under this subpart, a party may submit to the Presiding Officer—
- (1) A list of witnesses to be called, specifying the subject matter of the expected testimony of each witness, and
- (2) A list of exhibits to be considered for inclusion in the record.
- (b) If the Presiding Officer determines that the testimony of a witness

or the receipt of an exhibit in accordance with paragraph (a) of this section will be relevant, competent and material to the investigation, the Presiding Officer may subpoen the witness or use the exhibit during the investigation

§ 13.125 Depositions.

Depositions for investigative purposes may be taken at the discretion of the Presiding Officer with reasonable notice to the party under investigation. Such depositions shall be taken before the Presiding Officer or other person authorized to administer oaths and designated by the Presiding Officer. The testimony shall be reduced to writing by the person taking the deposition, or under the direction of that person, and where possible shall then be subscribed by the deponent. Any person may be compelled to appear and testify and to produce physical and documentary evidence.

§13.127 Reports, decisions and orders.

The Presiding Officer shall issue a written report based on the record developed during the formal investigation, including a summary of principal conclusions. A summary of principal conclusions shall be prepared by the official who issued the order of investigation in every case which results in no action, or no action as to a particular party to the investigation. All such reports shall be furnished to the parties to the investigation and filed in the public docket. Insertion of the report in the Public Docket shall constitute "entering of record" and publication as prescribed by section 313(b) of the Federal Aviation Act.

§13.129 Post-investigation action.

A decision on whether to initiate subsequent action shall be made on the basis of the record developed during the formal investigation and any other information in the possession of the Administrator.

§13.131 Other procedures.

Any question concerning the scope or conduct of a formal investigation not covered in this subpart may be ruled on by the Presiding Officer on motion of the Presiding Officer, or on the motion of a party or a person testifying or producing evidence.

Subpart G—Rules of Practice in FAA Civil Penalty Actions

SOURCE: Amdt. 13-21, 55 FR 27575, July 3, 1990, unless otherwise noted.

§ 13.201 Applicability.

- (a) This subpart applies to all civil penalty actions initiated under §13.16 of this part in which a hearing has been requested.
- (b) This subpart applies only to proceedings initiated after September 7, 1988. All other cases, hearings, or other proceedings pending or in progress before September 7, 1988, are not affected by the rules in this subpart.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended by Amdt. 13–32; 69 FR 59497, Oct. 4, 2004]

§13.202 Definitions.

Administrative law judge means an administrative law judge appointed pursuant to the provisions of 5 U.S.C. 3105.

Agency attorney means the Deputy Chief Counsel for Operations, the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, each Regional Counsel, the Aeronautical Center Counsel, or the Technical Center Counsel, or an attorney on the staff of the Assistant Chief Counsel, Enforcement, the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office, each Regional Counsel, the Aeronautical Center Counsel, or the Technical Center Counsel who prosecutes a civil penalty action. An agency attorney shall not include:

- (1) The Chief Counsel, the Deputy Chief Counsel for Policy and Adjudication, or the Assistant Chief Counsel for Litigation;
- (2) Any attorney on the staff of the Assistant Chief Counsel for Litigation;
- (3) Any attorney who is supervised in a civil penalty action by a person who provides such advice to the FAA decisionmaker in that action or a factually-related action.

Attorney means a person licensed by a state, the District of Columbia, or a

territory of the United States to practice law or appear before the courts of that state or territory.

Complaint means a document issued by an agency attorney alleging a violation of a provision of the Federal aviation statute listed in the first sentence of 49 U.S.C. 46301(d)(2) or in 49 U.S.C. 47531, or of the Federal hazardous materials transportation statute, 49 U.S.C. 5121–5128, or a rule, regulation, or order issued under those statutes, that has been filed with the Hearing Docket after a hearing has been requested under §13.16(f)(3) or (g)(2)(ii) of this part.

FAA decisionmaker means the Administrator of the Federal Aviation Administration, acting in the capacity of the decisionmaker on appeal, or any person to whom the Administrator has delegated the Administrator's decisionmaking authority in a civil penalty action. As used in this subpart, the FAA decisionmaker is the official authorized to issue a final decision and order of the Administrator in a civil penalty action.

Mail includes U.S. certified mail, U.S. registered mail, or use of an overnight express courier service.

Order assessing civil penalty means a document that contains a finding of a violation of a provision of the Federal aviation statute listed in the first sentence of 49 U.S.C. 46301(d)(2) or in 49 U.S.C. 47531, or of the Federal hazardous materials transportation statute, 49 U.S.C. 5121-5128, or a rule, regulation or order issued under those statutes, and may direct payment of a civil penalty. Unless an appeal is filed with the FAA decisionmaker in a timely manner, an initial decision or order of an administrative law judge shall be considered an order assessing civil penalty if an administrative law judge finds that an alleged violation occurred and determines that a civil penalty, in an amount found appropriate by the administrative law judge, is warranted. Unless a petition for review is filed with a U.S. Court of Appeals in a timely manner, a final decision and order of the Administrator shall be considered an order assessing civil penalty if the FAA decisionmaker finds that an alleged violation occurred and a civil penalty is warranted.

Party means the respondent or the Federal Aviation Administration (FAA).

Personal delivery includes hand-delivery or use of a contract or express messenger service. "Personal delivery" does not include the use of Government interoffice mail service.

Pleading means a complaint, an answer, and any amendment of these documents permitted under this subpart.

Properly addressed means a document that shows an address contained in agency records, a residential, business, or other address submitted by a person on any document provided under this subpart, or any other address shown by other reasonable and available means.

Respondent means a person, corporation, or company named in a complaint.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended by Amdt. 13–24, 58 FR 50241, Sept. 24, 1993; Amdt. 13–29, 62 FR 46866, Sept. 4, 1997; 70 FR 8238, Feb. 18, 2005; 71 FR 70464, Dec. 5, 2006]

§13.203 Separation of functions.

- (a) Civil penalty proceedings, including hearings, shall be prosecuted by an agency attorney.
- (b) An agency employee engaged in the performance of investigative or prosecutorial functions in a civil penalty action shall not, in that case or a factually-related case, participate or give advice in a decision by the administrative law judge or by the FAA decisionmaker on appeal, except as counsel or a witness in the public proceedings.
- (c) The Chief Counsel, the Deputy Chief Counsel for Policy and Adjudication, and the Assistant Chief Counsel for Litigation, or an attorney on the staff of the Assistant Chief Counsel for Litigation will advise the FAA decisionmaker regarding an initial decision or any appeal of a civil penalty action to the FAA decisionmaker.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended by Amdt. 13–24, 58 FR 50241, Sept. 24, 1993; 70 FR 8238, Feb. 18, 2005]

§ 13.204 Appearances and rights of parties.

- (a) Any party may appear and be heard in person.
- (b) Any party may be accompanied, represented, or advised by an attorney

or representative designated by the party and may be examined by that attorney or representative in any proceeding governed by this subpart. An attorney or representative who represents a party may file a notice of appearance in the action, in the manner provided in §13.210 of this subpart, and shall serve a copy of the notice of appearance on each party, in the manner provided in §13.211 of this subpart, before participating in any proceeding governed by this subpart. The attorney or representative shall include the name, address, and telephone number of the attorney or representative in the notice of appearance.

(c) Any person may request a copy of a document upon payment of reasonable costs. A person may keep an original document, data, or evidence, with the consent of the administrative law judge, by substituting a legible copy of the document for the record.

$\S 13.205$ Administrative law judges.

- (a) Powers of an administrative law judge. In accordance with the rules of this subpart, an administrative law judge may:
- (1) Give notice of, and hold, prehearing conferences and hearings;
- (2) Administer oaths and affirmations:
- (3) Issue subpoenas authorized by law and issue notices of deposition requested by the parties;
 - (4) Rule on offers of proof;
- (5) Receive relevant and material evidence:
- (6) Regulate the course of the hearing in accordance with the rules of this subpart;
- $(\overline{7})$ Hold conferences to settle or to simplify the issues by consent of the parties;
- (8) Dispose of procedural motions and requests; and
- (9) Make findings of fact and conclusions of law, and issue an initial decision.
- (b) Limitations on the power of the administrative law judge. The administrative law judge shall not issue an order of contempt, award costs to any party, or impose any sanction not specified in this subpart. If the administrative law judge imposes any sanction not specified in this subpart, a party may file an

interlocutory appeal of right with the FAA decisionmaker pursuant to §13.219(c)(4) of this subpart. This section does not preclude an administrative law judge from issuing an order that bars a person from a specific proceeding based on a finding of obstreperous or disruptive behavior in that specific proceeding.

(c) Disqualification. The administrative law judge may disqualify himself or herself at any time. A party may file a motion, pursuant to §13.218(f)(6), requesting that an administrative law judge be disqualified from the proceedings.

[Amdt. 13–21, 55 FR 27575, July 3, 1990; 55 FR 29293, July 18, 1990]

§13.206 Intervention.

- (a) A person may submit a motion for leave to intervene as a party in a civil penalty action. Except for good cause shown, a motion for leave to intervene shall be submitted not later than 10 days before the hearing.
- (b) If the administrative law judge finds that intervention will not unduly broaden the issues or delay the proceedings, the administrative law judge may grant a motion for leave to intervene if the person will be bound by any order or decision entered in the action or the person has a property, financial, or other legitimate interest that may not be addressed adequately by the parties. The administrative law judge may determine the extent to which an intervenor may participate in the proceedings.

§13.207 Certification of documents.

- (a) Signature required. The attorney of record, the party, or the party's representative shall sign each document tendered for filing with the hearing docket clerk, the administrative law judge, the FAA decisionmaker on appeal, or served on each party.
- (b) Effect of signing a document. By signing a document, the attorney of record, the party, or the party's representative certifies that the attorney, the party, or the party's representative has read the document and, based on reasonable inquiry and to the best of that person's knowledge, information, and belief, the document is—
- (1) Consistent with these rules;

- (2) Warranted by existing law or that a good faith argument exists for extension, modification, or reversal of existing law: and
- (3) Not unreasonable or unduly burdensome or expensive, not made to harass any person, not made to cause unnecessary delay, not made to cause needless increase in the cost of the proceedings, or for any other improper purpose.
- (c) Sanctions. If the attorney of record, the party, or the party's representative signs a document in violation of this section, the administrative law judge or the FAA decisionmaker shall:
- (1) Strike the pleading signed in violation of this section;
- (2) Strike the request for discovery or the discovery response signed in violation of this section and preclude further discovery by the party;
- (3) Deny the motion or request signed in violation of this section;
- (4) Exclude the document signed in violation of this section from the record:
- (5) Dismiss the interlocutory appeal and preclude further appeal on that issue by the party who filed the appeal until an initial decision has been entered on the record; or
- (6) Dismiss the appeal of the administrative law judge's initial decision to the FAA decisionmaker.

§13.208 Complaint.

- (a) Filing. The agency attorney shall file the original and one copy of the complaint with the hearing docket clerk, or may file a written motion pursuant to \$13.218(f)(2)(i) of this subpart instead of filing a complaint, not later than 20 days after receipt by the agency attorney of a request for hearing. The agency attorney should suggest a location for the hearing when filing the complaint.
- (b) Service. An agency attorney shall personally deliver or mail a copy of the complaint on the respondent, the president of the corporation or company named as a respondent, or a person designated by the respondent to accept service of documents in the civil penalty action.
- (c) Contents. A complaint shall set forth the facts alleged, any regulation

- allegedly violated by the respondent, and the proposed civil penalty in sufficient detail to provide notice of any factual or legal allegation and proposed civil penalty.
- (d) Motion to dismiss allegations or complaint. Instead of filing an answer to the complaint, a respondent may move to dismiss the complaint, or that part of the complaint, alleging a violation that occurred on or after August 2, 1990, and more than 2 years before an agency attorney issued a notice of proposed civil penalty to the respondent.
- (1) An administrative law judge may not grant the motion and dismiss the complaint or part of the complaint if the administrative law judge finds that the agency has shown good cause for any delay in issuing the notice of proposed civil penalty.
- (2) If the agency fails to show good cause for any delay, an administrative law judge may dismiss the complaint, or that part of the complaint, alleging a violation that occurred more than 2 years before an agency attorney issued the notice of proposed civil penalty to the respondent.
- (3) A party may appeal the administrative law judge's ruling on the motion to dismiss the complaint or any part of the complaint in accordance with §13.219(b) of this subpart.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended by Amdt. 13–22, 55 FR 31176, Aug. 1, 1990]

§13.209 Answer.

- (a) Writing required. A respondent shall file a written answer to the complaint, or may file a written motion pursuant to §13.208(d) or §13.218(f)(1-4) of this subpart instead of filing an answer, not later than 30 days after service of the complaint. The answer may be in the form of a letter but must be dated and signed by the person responding to the complaint. An answer may be typewritten or may be legibly handwritten.
- (b) Filing and address. A person filing an answer shall personally deliver or mail the original and one copy of the answer for filing with the hearing docket clerk, not later than 30 days after service of the complaint to the

Hearing Docket at the appropriate address set forth in §13.210(a) of this subpart. The person filing an answer should suggest a location for the hearing when filing the answer.

- (c) Service. A person filing an answer shall serve a copy of the answer on the agency attorney who filed the complaint.
- (d) Contents. An answer shall specifically state any affirmative defense that the respondent intends to assert at the hearing. A person filing an answer may include a brief statement of any relief requested in the answer.
- (e) Specific denial of allegations required. A person filing an answer shall admit, deny, or state that the person is without sufficient knowledge or information to admit or deny, each numbered paragraph of the complaint. Any statement or allegation contained in the complaint that is not specifically denied in the answer may be deemed an admission of the truth of that allegation. A general denial of the complaint is deemed a failure to file an answer.
- (f) Failure to file answer. A person's failure to file an answer without good cause shall be deemed an admission of the truth of each allegation contained in the complaint.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended at 70 FR 8238, Feb. 18, 2005]

§13.210 Filing of documents.

- (a) Address and method of filing. A person tendering a document for filing shall personally deliver or mail the signed original and one copy of each document to the Hearing Docket using the appropriate address:
- (1) If delivery is in person, or via expedited courier service: Federal Aviation Administration, 600 Independence Avenue, SW., Wilbur Wright Building—Suite 2W1000, Washington, DC 20591; Attention: Hearing Docket Clerk, AGC-430.
- (2) If delivery is via U.S. Mail: Federal Aviation Administration, 800 Independence Avenue, SW., Washington, DC 20591; Attention: Hearing Docket Clerk, AGC-430, Wilbur Wright Building—Suite 2W1000.
- (b) Date of filing. A document shall be considered to be filed on the date of personal delivery; or if mailed, the mailing date shown on the certificate

of service, the date shown on the postmark if there is no certificate of service, or other mailing date shown by other evidence if there is no certificate of service or postmark.

- (c) Form. Each document shall be typewritten or legibly handwritten.
- (d) Contents. Unless otherwise specified in this subpart, each document must contain a short, plain statement of the facts on which the person's case rests and a brief statement of the action requested in the document.
- (e) Internet accessibility of documents filed in the Hearing Docket. (1) Unless protected from public disclosure by an order of the ALJ under 13.226, all documents filed in the Hearing Docket are accessible through the Federal Docket Management System (FDMS): http://www.regulations.gov. To access a particular case file, use the FDMS number assigned to the case.
- (2) Decisions and orders issued by the Administrator in civil penalty cases, indexes of decisions, contact information for the FAA Hearing Docket and the administrative law judges, the rules of practice, and other information are available on the FAA civil penalty adjudication Web site at: http://www.faa.gov/about/office_org/

headquarters offices/agc/

pol adjudication/AGC400/Civil Penalty.

[Amdt. 13–21, 55 FR 27575, July 3, 1990; 55 FR 29293, July 18, 1990, as amended at 70 FR 8238, Feb. 18, 2005; 71 FR 70464, Dec. 5, 2006; 72 FR 14668, Mar. 29, 2007; 72 FR 68474, Dec. 5, 2007]

§13.211 Service of documents.

- (a) General. A person shall serve a copy of any document filed with the Hearing Docket on each party at the time of filing. Service on a party's attorney of record or a party's designated representative may be considered adequate service on the party.
- (b) *Type of service*. A person may serve documents by personal delivery or by mail.
- (c) Certificate of service. A person may attach a certificate of service to a document tendered for filing with the hearing docket clerk. A certificate of service shall consist of a statement, dated and signed by the person filing the document, that the document was personally delivered or mailed to each party on a specific date.

- (d) Date of service. The date of service shall be the date of personal delivery; or if mailed, the mailing date shown on the certificate of service, the date shown on the postmark if there is no certificate of service, or other mailing date shown by other evidence if there is no certificate of service or postmark.
- (e) Additional time after service by mail. Whenever a party has a right or a duty to act or to make any response within a prescribed period after service by mail, or on a date certain after service by mail, 5 days shall be added to the prescribed period.
- (f) Service by the administrative law judge. The administrative law judge shall serve a copy of each document including, but not limited to, notices of prehearing conferences and hearings, rulings on motions, decisions, and orders, upon each party to the proceedings by personal delivery or by mail
- (g) Valid service. A document that was properly addressed, was sent in accordance with this subpart, and that was returned, that was not claimed, or that was refused, is deemed to have been served in accordance with this subpart. The service shall be considered valid as of the date and the time that the document was deposited with a contract or express messenger, the document was mailed, or personal delivery of the document was refused.
- (h) Presumption of service. There shall be a presumption of service where a party or a person, who customarily receives mail, or receives it in the ordinary course of business, at either the person's residence or the person's principal place of business, acknowledges receipt of the document.

§13.212 Computation of time.

- (a) This section applies to any period of time prescribed or allowed by this subpart, by notice or order of the administrative law judge, or by any applicable statute.
- (b) The date of an act, event, or default, after which a designated time period begins to run, is not included in a computation of time under this subpart.
- (c) The last day of a time period is included in a computation of time unless it is a Saturday, Sunday, or a legal hol-

iday. If the last day of the time period is a Saturday, Sunday, or legal holiday, the time period runs until the end of the next day that is not a Saturday, Sunday, or legal holiday.

§13.213 Extension of time.

- (a) Oral requests. The parties may agree to extend for a reasonable period the time for filing a document under this subpart. If the parties agree, the administrative law judge shall grant one extension of time to each party. The party seeking the extension of time shall submit a draft order to the administrative law judge to be signed by the administrative law judge and filed with the hearing docket clerk. The administrative law judge may grant additional oral requests for an extension of time where the parties agree to the extension.
- (b) Written motion. A party shall file a written motion for an extension of time with the administrative law judge not later than 7 days before the document is due unless good cause for the late filing is shown. A party filing a written motion for an extension of time shall serve a copy of the motion on each party. The administrative law judge may grant the extension of time if good cause for the extension is shown.
- (c) Failure to rule. If the administrative law judge fails to rule on a written motion for an extension of time by the date the document was due, the motion for an extension of time is deemed granted for no more than 20 days after the original date the document was to be filed.

§13.214 Amendment of pleadings.

- (a) Filing and service. A party shall file the amendment with the administrative law judge and shall serve a copy of the amendment on all parties to the proceeding.
- (b) *Time*. A party shall file an amendment to a complaint or an answer within the following:
- (1) Not later than 15 days before the scheduled date of a hearing, a party may amend a complaint or an answer without the consent of the administrative law judge.

- (2) Less than 15 days before the scheduled date of a hearing, the administrative law judge may allow amendment of a complaint or an answer only for good cause shown in a motion to amend.
- (c) Responses. The administrative law judge shall allow a reasonable time, but not more than 20 days from the date of filing, for other parties to respond if an amendment to a complaint, answer, or other pleading has been filed with the administrative law judge.

§ 13.215 Withdrawal of complaint or request for hearing.

At any time before or during a hearing, an agency attorney may withdraw a complaint or a party may withdraw a request for a hearing without the consent of the administrative law judge. If an agency attorney withdraws the complaint or a party withdraws the request for a hearing and the answer, the administrative law judge shall dismiss the proceedings under this subpart with prejudice.

§13.216 Waivers.

Waivers of any rights provided by statute or regulation shall be in writing or by stipulation made at a hearing and entered into the record. The parties shall set forth the precise terms of the waiver and any conditions.

§13.217 Joint procedural or discovery schedule.

- (a) General. The parties may agree to submit a schedule for filing all prehearing motions, a schedule for conducting discovery in the proceedings, or a schedule that will govern all prehearing motions and discovery in the proceedings.
- (b) Form and content of schedule. If the parties agree to a joint procedural or discovery schedule, one of the parties shall file the joint schedule with the administrative law judge, setting forth the dates to which the parties have agreed, and shall serve a copy of the joint schedule on each party.
- (1) The joint schedule may include, but need not be limited to, requests for discovery, any objections to discovery requests, responses to discovery requests to which there are no objections, submission of prehearing mo-

- tions, responses to prehearing motions, exchange of exhibits to be introduced at the hearing, and a list of witnesses that may be called at the hearing.
- (2) Each party shall sign the original joint schedule to be filed with the administrative law judge.
- (c) *Time*. The parties may agree to submit all prehearing motions and responses and may agree to close discovery in the proceedings under the joint schedule within a reasonable time before the date of the hearing, but not later than 15 days before the hearing.
- (d) Order establishing joint schedule. The administrative law judge shall approve the joint schedule filed by the parties. One party shall submit a draft order establishing a joint schedule to the administrative law judge to be signed by the administrative law judge and filed with the hearing docket clerk.
- (e) Disputes. The administrative law judge shall resolve disputes regarding discovery or disputes regarding compliance with the joint schedule as soon as possible so that the parties may continue to comply with the joint schedule.
- (f) Sanctions for failure to comply with joint schedule. If a party fails to comply with the administrative law judge's order establishing a joint schedule, the administrative law judge may direct that party to comply with a motion to discovery request or, limited to the extent of the party's failure to comply with a motion or discovery request, the administrative law judge may:
- (1) Strike that portion of a party's pleadings;
- (2) Preclude prehearing or discovery motions by that party;
- (3) Preclude admission of that portion of a party's evidence at the hearing, or
- (4) Preclude that portion of the testimony of that party's witnesses at the hearing.

§ 13.218 Motions.

(a) General. A party applying for an order or ruling not specifically provided in this subpart shall do so by motion. A party shall comply with the requirements of this section when filing a motion with the administrative law

judge. A party shall serve a copy of each motion on each party.

- (b) Form and contents. A party shall state the relief sought by the motion and the particular grounds supporting that relief. If a party has evidence in support of a motion, the party shall attach any supporting evidence, including affidavits, to the motion.
- (c) Filing of motions. A motion made prior to the hearing must be in writing. Unless otherwise agreed by the parties or for good cause shown, a party shall file any prehearing motion, and shall serve a copy on each party, not later than 30 days before the hearing. Motions introduced during a hearing may be made orally on the record unless the administrative law judge directs otherwise.
- (d) Answers to motions. Any party may file an answer, with affidavits or other evidence in support of the answer, not later than 10 days after service of a written motion on that party. When a motion is made during a hearing, the answer may be made at the hearing on the record, orally or in writing, within a reasonable time determined by the administrative law judge.
- (e) Rulings on motions. The administrative law judge shall rule on all motions as follows:
- (1) Discovery motions. The administrative law judge shall resolve all pending discovery motions not later than 10 days before the hearing.
- (2) Prehearing motions. The administrative law judge shall resolve all pending prehearing motions not later than 7 days before the hearing. If the administrative law judge issues a ruling or order orally, the administrative law judge shall serve a written copy of the ruling or order, within 3 days, on each party. In all other cases, the administrative law judge shall issue rulings and orders in writing and shall serve a copy of the ruling or order on each party.
- (3) Motions made during the hearing. The administrative law judge may issue rulings and orders on motions made during the hearing orally. Oral rulings or orders on motions must be made on the record.
- (f) Specific motions. A party may file the following motions with the administrative law judge:

- (1) Motion to dismiss for insufficiency. A respondent may file a motion to dismiss the complaint for insufficiency instead of filing an answer. If the administrative law judge denies the motion to dismiss the complaint for insufficiency, the respondent shall file an answer not later than 10 days after service of the administrative law judge's denial of the motion. A motion to dismiss the complaint for insufficiency must show that the complaint fails to state a violation of a provision of the Federal aviation statute listed in the first sentence in 49 U.S.C. 46301(d)(2) or in 49 U.S.C. 47531, or any implementing rule, regulation, or order, or a violation of the Federal hazardous materials transportation statute, 49 U.S.C. 5121-5128, or any implementing rule, regulation or order.
- (2) Motion to dismiss. A party may file a motion to dismiss, specifying the grounds for dismissal. If an administrative law judge grants a motion to dismiss in part, a party may appeal the administrative law judge's ruling on the motion to dismiss under §13.219(b) of this subpart.
- (i) Motion to dismiss a request for a hearing. An agency attorney may file a motion to dismiss a request for a hearing instead of filing a complaint. If the motion to dismiss is not granted, the agency attorney shall file the complaint and shall serve a copy of the complaint on each party not later than 10 days after service of the administrative law judge's ruling or order on the motion to dismiss. If the motion to dismiss is granted and the proceedings are terminated without a hearing, the respondent may file an appeal pursuant to §13.233 of this subpart. If required by the decision on appeal, the agency attorney shall file a complaint and shall serve a copy of the complaint on each party not later than 10 days after service of the decision on appeal.
- (ii) Motion to dismiss a complaint. A respondent may file a motion to dismiss a complaint instead of filing an answer. If the motion to dismiss is not granted, the respondent shall file an answer and shall serve a copy of the answer on each party not later than 10 days after service of the administrative law judge's ruling or order on the motion to dismiss. If the motion to dismiss is

granted and the proceedings are terminated without a hearing, the agency attorney may file an appeal pursuant to §13.233 of this subpart. If required by the decision on appeal, the respondent shall file an answer and shall serve a copy of the answer on each party not later than 10 days after service of the decision on appeal.

- (3) Motion for more definite statement. A party may file a motion for more definite statement of any pleading which requires a response under this subpart. A party shall set forth, in detail, the indefinite or uncertain allegations contained in a complaint or response to any pleading and shall submit the details that the party believes would make the allegation or response definite and certain.
- (i) Complaint. A respondent may file a motion requesting a more definite statement of the allegations contained in the complaint instead of filing an answer. If the administrative law judge grants the motion, the agency attorney shall supply a more definite statement not later than 15 days after service of the ruling granting the motion. If the agency attorney fails to supply a more definite statement, the administrative law judge shall strike the allegations in the complaint to which the motion is directed. If the administrative law judge denies the motion, the respondent shall file an answer and shall serve a copy of the answer on each party not later than 10 days after service of the order of denial.
- (ii) Answer. An agency attorney may file a motion requesting a more definite statement if an answer fails to respond clearly to the allegations in the complaint. If the administrative law judge grants the motion, the respondent shall supply a more definite statement not later than 15 days after service of the ruling on the motion. If the respondent fails to supply a more definite statement, the administrative law judge shall strike those statements in the answer to which the motion is directed. The respondent's failure to supply a more definite statement may be deemed an admission of unanswered allegations in the complaint.
- (4) Motion to strike. Any party may make a motion to strike any insufficient allegation or defense, or any re-

dundant, immaterial, or irrelevant matter in a pleading. A party shall file a motion to strike with the administrative law judge and shall serve a copy on each party before a response is required under this subpart or, if a response is not required, not later than 10 days after service of the pleading.

- (5) Motion for decision. A party may make a motion for decision, regarding all or any part of the proceedings, at any time before the administrative law judge has issued an initial decision in the proceedings. The administrative law judge shall grant a party's motion for decision if the pleadings, depositions, answers to interrogatories, admissions, matters that the administrative law judge has officially noticed, or evidence introduced during the hearing show that there is no genuine issue of material fact and that the party making the motion is entitled to a decision as a matter of law. The party making the motion for decision has the burden of showing that there is no genuine issue of material fact disputed by the parties.
- (6) Motion for disqualification. A party may file a motion for disqualification with the administrative law judge and shall serve a copy on each party. A party may file the motion at any time after the administrative law judge has been assigned to the proceedings but shall make the motion before the administrative law judge files an initial decision in the proceedings.
- (i) Motion and supporting affidavit. A party shall state the grounds for disqualification, including, but not limited to, personal bias, pecuniary interest, or other factors showing disqualification, in the motion for disqualification. A party shall submit an affidavit with the motion for disqualification that sets forth, in detail, the matters alleged to constitute grounds for disqualification.
- (ii) Answer. A party shall respond to the motion for disqualification not later than 5 days after service of the motion for disqualification.
- (iii) Decision on motion for disqualification. The administrative law judge shall render a decision on the motion for disqualification not later than 15 days after the motion has been filed. If the administrative law judge finds that

the motion for disqualification and supporting affidavit show a basis for disqualification, the administrative law judge shall withdraw from the proceedings immediately. If the administrative law judge finds that disqualification is not warranted, the administrative law judge shall deny the motion and state the grounds for the denial on the record. If the administrative law judge fails to rule on a party's motion for disqualification within 15 days after the motion has been filed, the motion is deemed granted.

(iv) *Appeal*. A party may appeal the administrative law judge's denial of the motion for disqualification in accordance with §13.219(b) of this subpart.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended by 71 FR 70464, Dec. 5, 2006]

§ 13.219 Interlocutory appeals.

- (a) General. Unless otherwise provided in this subpart, a party may not appeal a ruling or decision of the administrative law judge to the FAA decisionmaker until the initial decision has been entered on the record. A decision or order of the FAA decisionmaker on the interlocutory appeal does not constitute a final order of the Administrator for the purposes of judicial appellate review as provided in §13.235 of this subpart.
- (b) Interlocutory appeal for cause. If a party files a written request for an interlocutory appeal for cause with the administrative law judge, or orally requests an interlocutory appeal for cause, the proceedings are stayed until the administrative law judge issues a decision on the request. If the administrative law judge grants the request, the proceedings are stayed until the FAA decisionmaker issues a decision on the interlocutory appeal. The administrative law judge shall grant an interlocutory appeal for cause if a party shows that delay of the appeal would be detrimental to the public interest or would result in undue prejudice to any party.
- (c) Interlocutory appeals of right. If a party notifies the administrative law judge of an interlocutory appeal of right, the proceedings are stayed until the FAA decisionmaker issues a decision on the interlocutory appeal. A

party may file an interlocutory appeal with the FAA decisionmaker, without the consent of the administrative law judge, before an initial decision has been entered in the case of:

- (1) A ruling or order by the administrative law judge barring a person from the proceedings.
- (2) Failure of the administrative law judge to dismiss the proceedings in accordance with \$13.215 of this subpart.
- (3) A ruling or order by the administrative law judge in violation of §13.205(b) of this subpart.
- (d) Procedure. A party shall file a notice of interlocutory appeal, with supporting documents, with the FAA decisionmaker and the hearing docket clerk, and shall serve a copy of the notice and supporting documents on each party and the administrative law judge, not later than 10 days after the administrative law judge's decision forming the basis of an interlocutory appeal of right or not later than 10 days after the administrative law judge's decision granting an interlocutory appeal for cause, whichever is appropriate. A party shall file a reply brief, if any, with the FAA decisionmaker and serve a copy of the reply brief on each party, not later than 10 days after service of the appeal brief. The FAA decisionmaker shall render a decision on the interlocutory appeal, on the record and as a part of the decision in the proceedings, within a reasonable time after receipt of the interlocutory appeal.
- (e) The FAA decisionmaker may reject frivolous, repetitive, or dilatory appeals, and may issue an order precluding one or more parties from making further interlocutory appeals in a proceeding in which there have been frivolous, repetitive, or dilatory interlocutory appeals.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended by Amdt. 13–23, 55 FR 45983, Oct. 31, 1990; 71 FR 70464, Dec. 5, 2006]

§13.220 Discovery.

(a) Initiation of discovery. Any party may initiate discovery described in this section, without the consent or approval of the administrative law judge, at any time after a complaint has been filed in the proceedings.

- (b) Methods of discovery. The following methods of discovery are permitted under this section: depositions on oral examination or written questions of any person; written interrogatories directed to a party; requests for production of documents or tangible items to any person; and requests for admission by a party. A party is not required to file written interrogatories and responses, requests for production of documents or tangible items and responses, and requests for admission and response with the administrative law judge or the hearing docket clerk. In the event of a discovery dispute, a party shall attach a copy of these documents in support of a motion made under this section.
- (c) Service on the agency. A party shall serve each discovery request directed to the agency or any agency employee on the agency attorney of record.
- (d) Time for response to discovery requests. Unless otherwise directed by this subpart or agreed by the parties, a party shall respond to a request for discovery, including filing objections to a request for discovery, not later than 30 days of service of the request.
- (e) Scope of discovery. Subject to the limits on discovery set forth in paragraph (f) of this section, a party may discover any matter that is not privileged and that is relevant to the subject matter of the proceeding. A party may discover information that relates to the claim or defense of any party including the existence, description, nature, custody, condition, and location of any document or other tangible item and the identity and location of any person having knowledge of discoverable matter. A party may discover facts known, or opinions held, by an expert who any other party expects to call to testify at the hearing. A party has no ground to object to a discovery request on the basis that the information sought would not be admissible at the hearing if the information sought during discovery is reasonably calculated to lead to the discovery of admissible evidence.
- (f) Limiting discovery. The administrative law judge shall limit the frequency and extent of discovery permitted by this section if a party shows that—

- (1) The information requested is cumulative or repetitious:
- (2) The information requested can be obtained from another less burdensome and more convenient source:
- (3) The party requesting the information has had ample opportunity to obtain the information through other discovery methods permitted under this section: or
- (4) The method or scope of discovery requested by the party is unduly burdensome or expensive.
- (g) Confidential orders. A party or person who has received a discovery request for information that is related to a trade secret, confidential or sensitive material, competitive or commercial information, proprietary data, or information on research and development, may file a motion for a confidential order with the administrative law judge and shall serve a copy of the motion for a confidential order on each party.
- (1) The party or person making the motion must show that the confidential order is necessary to protect the information from disclosure to the public
- (2) If the administrative law judge determines that the requested material is not necessary to decide the case, the administrative law judge shall preclude any inquiry into the matter by any party.
- (3) If the administrative law judge determines that the requested material may be disclosed during discovery, the administrative law judge may order that the material may be discovered and disclosed under limited conditions or may be used only under certain terms and conditions.
- (4) If the administrative law judge determines that the requested material is necessary to decide the case and that a confidential order is warranted, the administrative law judge shall provide:
- (i) An opportunity for review of the document by the parties off the record;
- (ii) Procedures for excluding the information from the record; and
- (iii) Order that the parties shall not disclose the information in any manner and the parties shall not use the information in any other proceeding.

- (h) Protective orders. A party or a person who has received a request for discovery may file a motion for protective order with the administrative law judge and shall serve a copy of the motion for protective order on each party. The party or person making the motion must show that the protective order is necessary to protect the party or the person from annoyance, embarrassment, oppression, or undue burden or expense. As part of the protective order, the administrative law judge may:
 - (1) Deny the discovery request;
- (2) Order that discovery be conducted only on specified terms and conditions, including a designation of the time or place for discovery or a determination of the method of discovery; or
- (3) Limit the scope of discovery or preclude any inquiry into certain matters during discovery.
- (i) Duty to supplement or amend responses. A party who has responded to a discovery request has a duty to supplement or amend the response, as soon as the information is known, as follows:
- (1) A party shall supplement or amend any response to a question requesting the identity and location of any person having knowledge of discoverable matters.
- (2) A party shall supplement or amend any response to a question requesting the identity of each person who will be called to testify at the hearing as an expert witness and the subject matter and substance of that witness' testimony.
- (3) A party shall supplement or amend any response that was incorrect when made or any response that was correct when made but is no longer correct, accurate, or complete.
- (j) *Depositions*. The following rules apply to depositions taken pursuant to this section:
- (1) Form. A deposition shall be taken on the record and reduced to writing. The person being deposed shall sign the deposition unless the parties agree to waive the requirement of a signature.
- (2) Administration of oaths. Within the United States, or a territory or possession subject to the jurisdiction of the United States, a party shall take a deposition before a person authorized to administer oaths by the laws of the

- United States or authorized by the law of the place where the examination is held. In foreign countries, a party shall take a deposition in any manner allowed by the Federal Rules of Civil Procedure.
- (3) Notice of deposition. A party shall serve a notice of deposition, stating the time and place of the deposition and the name and address of each person to be examined, on the person to be deposed, on the administrative law judge, on the hearing docket clerk, and on each party not later than 7 days before the deposition. A party may serve a notice of deposition less than 7 days before the deposition only with consent of the administrative law judge. If a subpoena duces tecum is to be served on the person to be examined, the party shall attach a copy of the subpoena duces tecum that describes the materials to be produced at the deposition to the notice of deposition.
- (4) Use of depositions. A party may use any part or all of a deposition at a hearing authorized under this subpart only upon a showing of good cause. The deposition may be used against any party who was present or represented at the deposition or who had reasonable notice of the deposition.
- (k) Interrogatories. A party, the party's attorney, or the party's representative may sign the party's responses to interrogatories. A party shall answer each interrogatory separately and completely in writing. If a party objects to an interrogatory, the party shall state the objection and the reasons for the objection. An opposing party may use any part or all of a party's responses to interrogatories at a hearing authorized under this subpart to the extent that the response is relevant, material, and not repetitious.
- (1) A party shall not serve more than 30 interrogatories to each other party. Each subpart of an interrogatory shall be counted as a separate interrogatory.
- (2) A party shall file a motion for leave to serve additional interrogatories on a party with the administrative law judge before serving additional interrogatories on a party. The administrative law judge shall grant the motion only if the party shows good cause for the party's failure to inquire about the information previously and that

the information cannot reasonably be obtained using less burdensome discovery methods or be obtained from other sources.

- (1) Requests for admission. A party may serve a written request for admission of the truth of any matter within the scope of discovery under this section or the authenticity of any document described in the request. A party shall set forth each request for admission separately. A party shall serve copies of documents referenced in the request for admission unless the documents have been provided or are reasonably available for inspection and copying.
- (1) Time. A party's failure to respond to a request for admission, in writing and signed by the attorney or the party, not later than 30 days after service of the request, is deemed an admission of the truth of the statement or statements contained in the request for admission. The administrative law judge may determine that a failure to respond to a request for admission is not deemed an admission of the truth if a party shows that the failure was due to circumstances beyond the control of the party or the party's attorney.
- (2) Response. A party may object to a request for admission and shall state the reasons for objection. A party may specifically deny the truth of the matter or describe the reasons why the party is unable to truthfully deny or admit the matter. If a party is unable to deny or admit the truth of the matter, the party shall show that the party has made reasonable inquiry into the matter or that the information known to, or readily obtainable by, the party is insufficient to enable the party to admit or deny the matter. A party may admit or deny any part of the request for admission. If the administrative law judge determines that a response does not comply with the requirements of this rule or that the response is insufficient, the matter is deemed admitted.
- (3) Effect of admission. Any matter admitted or deemed admitted under this section is conclusively established for the purpose of the hearing and appeal.
- (m) Motion to compel discovery. A party may make a motion to compel discovery if a person refuses to answer

- a question during a deposition, a party fails or refuses to answer an interrogatory, if a person gives an evasive or incomplete answer during a deposition or when responding to an interrogatory, or a party fails or refuses to produce documents or tangible items. During a deposition, the proponent of a question may complete the deposition or may adjourn the examination before making a motion to compel if a person refuses to answer.
- (n) Failure to comply with a discovery order or order to compel. If a party fails to comply with a discovery order or an order to compel, the administrative law judge, limited to the extent of the party's failure to comply with the discovery order or motion to compel, may:
- (1) Strike that portion of a party's pleadings;
- (2) Preclude prehearing or discovery motions by that party;
- (3) Preclude admission of that portion of a party's evidence at the hearing; or
- (4) Preclude that portion of the testimony of that party's witnesses at the hearing.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended by Amdt. 13–23, 55 FR 45983, Oct. 31, 1990]

§13.221 Notice of hearing.

- (a) *Notice*. The administrative law judge shall give each party at least 60 days notice of the date, time, and location of the hearing.
- (b) Date, time, and location of the hearing. The administrative law judge to whom the proceedings have been assigned shall set a reasonable date, time, and location for the hearing. The administrative law judge shall consider the need for discovery and any joint procedural or discovery schedule submitted by the parties when determining the hearing date. The administrative law judge shall give due regard to the convenience of the parties, the location where the majority of the witnesses reside or work, and whether the location is served by a scheduled air carrier.
- (c) Earlier hearing. With the consent of the administrative law judge, the parties may agree to hold the hearing

on an earlier date than the date specified in the notice of hearing.

§13.222 Evidence.

- (a) General. A party is entitled to present the party's case or defense by oral, documentary, or demonstrative evidence, to submit rebuttal evidence, and to conduct any cross-examination that may be required for a full and true disclosure of the facts.
- (b) Admissibility. A party may introduce any oral, documentary, or demonstrative evidence in support of the party's case or defense. The administrative law judge shall admit any oral, documentary, or demonstrative evidence introduced by a party but shall exclude irrelevant, immaterial, or unduly repetitious evidence.
- (c) Hearsay evidence. Hearsay evidence is admissible in proceedings governed by this subpart. The fact that evidence submitted by a party is hearsay goes only to the weight of the evidence and does not affect its admissibility.

§13.223 Standard of proof.

The administrative law judge shall issue an initial decision or shall rule in a party's favor only if the decision or ruling is supported by, and in accordance with, the reliable, probative, and substantial evidence contained in the record. In order to prevail, the party with the burden of proof shall prove the party's case or defense by a preponderance of reliable, probative, and substantial evidence.

§13.224 Burden of proof.

- (a) Except in the case of an affirmative defense, the burden of proof is on the agency.
- (b) Except as otherwise provided by statute or rule, the proponent of a motion, request, or order has the burden of proof.
- (c) A party who has asserted an affirmative defense has the burden of proving the affirmative defense.

§ 13.225 Offer of proof.

A party whose evidence has been excluded by a ruling of the administrative law judge may offer the evidence for the record on appeal.

§ 13.226 Public disclosure of evidence.

- (a) The administrative law judge may order that any information contained in the record be withheld from public disclosure. Any person may object to disclosure of information in the record by filing a written motion to withhold specific information with the administrative law judge and serving a copy of the motion on each party. The party shall state the specific grounds for non-disclosure in the motion.
- (b) The administrative law judge shall grant the motion to withhold information in the record if, based on the motion and any response to the motion, the administrative law judge determines that disclosure would be detrimental to aviation safety, disclosure would not be in the public interest, or that the information is not otherwise required to be made available to the public.

§13.227 Expert or opinion witnesses.

An employee of the agency may not be called as an expert or opinion witness, for any party other than the FAA, in any proceeding governed by this subpart. An employee of a respondent may not be called by an agency attorney as an expert or opinion witness for the FAA in any proceeding governed by this subpart to which the respondent is a party.

§13.228 Subpoenas.

- (a) Request for subpoena. A party may obtain a subpoena to compel the attendance of a witness at a deposition or hearing or to require the production of documents or tangible items from the hearing docket clerk. The hearing docket clerk shall deliver the subpoena, signed by the hearing docket clerk or an administrative law judge but otherwise in blank, to the party. The party shall complete the subpoena, stating the title of the action and the date and time for the witness' attendance or production of documents or items. The party who obtained the subpoena shall serve the subpoena on the witness
- (b) Motion to quash or modify the subpoena. A party, or any person upon whom a subpoena has been served, may file a motion to quash or modify the subpoena with the administrative law

judge at or before the time specified in the subpoena for compliance. The applicant shall describe, in detail, the basis for the application to quash or modify the suppoena including, but not limited to, a statement that the testimony, document, or tangible evidence is not relevant to the proceeding, that the subpoena is not reasonably tailored to the scope of the proceeding, or that the subpoena is unreasonable and oppressive. A motion to quash or modify the subpoena will stay the effect of the subpoena pending a decision by the administrative law judge on the motion.

(c) Enforcement of subpoena. Upon a showing that a person has failed or refused to comply with a subpoena, a party may apply to the local federal district court to seek judicial enforcement of the subpoena in accordance with 49 U.S.C. 46104 in cases under the Federal aviation statute.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended at 71 FR 70465, Dec. 5, 2006]

§ 13.229 Witness fees.

(a) General. Unless otherwise authorized by the administrative law judge, the party who applies for a subpoena to compel the attendance of a witness at a deposition or hearing, or the party at whose request a witness appears at a deposition or hearing, shall pay the witness fees described in this section.

(b) Amount. Except for an employee of the agency who appears at the direction of the agency, a witness who appears at a deposition or hearing is entitled to the same fees and mileage expenses as are paid to a witness in a court of the United States in comparable circumstances.

§13.230 Record.

(a) Exclusive record. The transcript of all testimony in the hearing, all exhibits received into evidence, and all motions, applications, requests, and rulings shall constitute the exclusive record for decision of the proceedings and the basis for the issuance of any orders in the proceeding. Any proceedings regarding the disqualification of an administrative law judge shall be included in the record.

(b) Examination and copying of record. Any person may examine the record at the Hearing Docket, Federal Aviation Administration, 600 Independence Avenue, SW., Wilbur Wright Building—Room 2014, Washington, DC 20591. Documents may also be examined and copied at the U.S. Department of Transportation, Docket Operations, West Building Ground Floor, Room W12–140, 1200 New Jersey Avenue, SE., Washington, DC 20590. Any person may have a copy of the record after payment of reasonable costs to copy the record.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended at 70 FR 8238, Feb. 18, 2005; 72 FR 68474, Dec. 5, 2007]

§13.231 Argument before the administrative law judge.

(a) Arguments during the hearing. During the hearing, the administrative law judge shall give the parties a reasonable opportunity to present arguments on the record supporting or opposing motions, objections, and rulings if the parties request an opportunity for argument. The administrative law judge may request written arguments during the hearing if the administrative law judge finds that submission of written arguments would be reasonable.

(b) Final oral argument. At the conclusion of the hearing and before the administrative law judge issues an initial decision in the proceedings, the parties are entitled to submit oral proposed findings of fact and conclusions of law, exceptions to rulings of the administrative law judge, and supporting arguments for the findings, conclusions, or exceptions. At the conclusion of the hearing, a party may waive final oral argument.

(c) Posthearing briefs. The administrative law judge may request written posthearing briefs before the administrative law judge issues an initial decision in the proceedings if the administrative law judge finds that submission of written arguments would be reasonable. If a party files a written posthearing brief, the party shall include proposed findings of fact and conclusions of law, exceptions to rulings of the administrative law judge, and supporting arguments for the findings, conclusions, or exceptions. The administrative law judge shall give the parties a reasonable opportunity, not more than 30 days after receipt of the

transcript, to prepare and submit the briefs.

§13.232 Initial decision.

- (a) Contents. The administrative law judge shall issue an initial decision at the conclusion of the hearing. In each oral or written decision, the administrative law judge shall include findings of fact and conclusions of law, and the grounds supporting those findings and conclusions, upon all material issues of fact, the credibility of witnesses, the applicable law, any exercise of the administrative law judge's discretion, the amount of any civil penalty found appropriate by the administrative law judge, and a discussion of the basis for any order issued in the proceedings. The administrative law judge is not required to provide a written explanation for rulings on objections, procedural motions, and other matters not directly relevant to the substance of the initial decision. If the administrative law judge refers to any previous unreported or unpublished initial decision, the administrative law judge shall make copies of that initial decision available to all parties and the FAA decisionmaker.
- (b) *Oral decision*. Except as provided in paragraph (c) of this section, at the conclusion of the hearing, the administrative law judge shall issue the initial decision and order orally on the record.
- (c) Written decision. The administrative law judge may issue a written initial decision not later than 30 days after the conclusion of the hearing or submission of the last posthearing brief if the administrative law judge finds that issuing a written initial decision is reasonable. The administrative law judge shall serve a copy of any written initial decision on each party.
- (d) Order assessing civil penalty. Unless appealed pursuant to §13.233 of this subpart, the initial decision issued by the administrative law judge shall be considered an order assessing civil penalty if the administrative law judge finds that an alleged violation occurred and determines that a civil penalty, in an amount found appropriate by the administrative law judge, is warranted.

§13.233 Appeal from initial decision.

- (a) Notice of appeal. A party may appeal the initial decision, and any decision not previously appealed pursuant to §13.219, by filing a notice of appeal with the FAA decisionmaker. A party must file the notice of appeal in the FAA Hearing Docket using the appropriate address listed in §13.210(a). A party shall file the notice of appeal not later than 10 days after entry of the oral initial decision on the record or service of the written initial decision on the parties and shall serve a copy of the notice of appeal on each party.
- (b) Issues on appeal. In any appeal from a decision of an administrative law judge, the FAA decisionmaker considers only the following issues:
- (1) Whether each finding of fact is supported by a preponderance of reliable, probative, and substantial evidence:
- (2) Whether each conclusion of law is made in accordance with applicable law, precedent, and public policy; and
- (3) Whether the administrative law judge committed any prejudicial errors that support the appeal.
- (c) Perfecting an appeal. Unless otherwise agreed by the parties, a party shall perfect an appeal, not later than 50 days after entry of the oral initial decision on the record or service of the written initial decision on the party, by filing an appeal brief with the FAA decisionmaker.
- (1) Extension of time by agreement of the parties. The parties may agree to extend the time for perfecting the appeal with the consent of the FAA decisionmaker. If the FAA decisionmaker grants an extension of time to perfect the appeal, the appellate docket clerk shall serve a letter confirming the extension of time on each party.
- (2) Written motion for extension. If the parties do not agree to an extension of time for perfecting an appeal, a party desiring an extension of time may file a written motion for an extension with the FAA decisionmaker and shall serve a copy of the motion on each party. The FAA decisionmaker may grant an extension if good cause for the extension is shown in the motion.

- (d) Appeal briefs. A party shall file the appeal brief with the FAA decision-maker and shall serve a copy of the appeal brief on each party.
- (1) A party shall set forth, in detail, the party's specific objections to the initial decision or rulings in the appeal brief. A party also shall set forth, in detail, the basis for the appeal, the reasons supporting the appeal, and the relief requested in the appeal. If the party relies on evidence contained in the record for the appeal, the party shall specifically refer to the pertinent evidence contained in the transcript in the appeal brief.
- (2) The FAA decisionmaker may dismiss an appeal, on the FAA decisionmaker's own initiative or upon motion of any other party, where a party has filed a notice of appeal but fails to perfect the appeal by timely filing an appeal brief with the FAA decisionmaker.
- (e) Reply brief. Unless otherwise agreed by the parties, any party may file a reply brief with the FAA decisionmaker not later than 35 days after the appeal brief has been served on that party. The party filing the reply brief shall serve a copy of the reply brief on each party. If the party relies on evidence contained in the record for the reply, the party shall specifically refer to the pertinent evidence contained in the reply brief.
- (1) Extension of time by agreement of the parties. The parties may agree to extend the time for filing a reply brief with the consent of the FAA decision-maker. If the FAA decisionmaker grants an extension of time to file the reply brief, the appellate docket clerk shall serve a letter confirming the extension of time on each party.
- (2) Written motion for extension. If the parties do not agree to an extension of time for filing a reply brief, a party desiring an extension of time may file a written motion for an extension with the FAA decisionmaker and shall serve a copy of the motion on each party. The FAA decisionmaker may grant an extension if good cause for the extension is shown in the motion.
- (f) Other briefs. The FAA decision-maker may allow any person to submit an *amicus curiae* brief in an appeal of an initial decision. A party may not file

- more than one appeal brief or reply brief. A party may petition the FAA decisionmaker, in writing, for leave to file an additional brief and shall serve a copy of the petition on each party. The party may not file the additional brief with the petition. The FAA decisionmaker may grant leave to file an additional brief if the party demonstrates good cause for allowing additional argument on the appeal. The FAA decisionmaker will allow a reasonable time for the party to file the additional brief.
- (g) Number of copies. A party shall file the original appeal brief or the original reply brief, and two copies of the brief, with the FAA decisionmaker.
- (h) Oral argument. The FAA decision-maker has sole discretion to permit oral argument on the appeal. On the FAA decisionmaker's own initiative or upon written motion by any party, the FAA decisionmaker may find that oral argument will contribute substantially to the development of the issues on appeal and may grant the parties an opportunity for oral argument.
- (i) Waiver of objections on appeal. If a party fails to object to any alleged error regarding the proceedings in an appeal or a reply brief, the party waives any objection to the alleged error. The FAA decisionmaker is not required to consider any objection in an appeal brief or any argument in the reply brief if a party's objection is based on evidence contained on the record and the party does not specifically refer to the pertinent evidence from the record in the brief.
- (j) FAA decisionmaker's decision on appeal. The FAA decisionmaker will review the record, the briefs on appeal, and the oral argument, if any, when considering the issues on appeal. The FAA decisionmaker may affirm, modify, or reverse the initial decision, make any necessary findings, or may remand the case for any proceedings that the FAA decisionmaker determines may be necessary.
- (1) The FAA decisionmaker may raise any issue, on the FAA decisionmaker's own initiative, that is required for proper disposition of the proceedings. The FAA decisionmaker will give the parties a reasonable opportunity to submit arguments on the new issues

before making a decision on appeal. If an issue raised by the FAA decisionmaker requires the consideration of additional testimony or evidence, the FAA decisionmaker will remand the case to the administrative law judge for further proceedings and an initial decision related to that issue. If an issue raised by the FAA decisionmaker is solely an issue of law or the issue was addressed at the hearing but was not raised by a party in the briefs on appeal, a remand of the case to the administrative law judge for further proceedings is not required but may be provided in the discretion of the FAA decisionmaker.

- (2) The FAA decisionmaker will issue the final decision and order of the Administrator on appeal in writing and will serve a copy of the decision and order on each party. Unless a petition for review is filed pursuant to §13.235, a final decision and order of the Administrator shall be considered an order assessing civil penalty if the FAA decisionmaker finds that an alleged violation occurred and a civil penalty is warranted.
- (3) A final decision and order of the Administrator after appeal is precedent in any other civil penalty action. Any issue, finding or conclusion, order, ruling, or initial decision of an administrative law judge that has not been appealed to the FAA decisionmaker is not precedent in any other civil penalty action.

[Amdt. 13–21, 55 FR 27575, July 3, 1990, as amended by Amdt. 13–32; 69 FR 59498, Oct. 4, 2004; 70 FR 13345, Mar. 21, 2005]

§ 13.234 Petition to reconsider or modify a final decision and order of the FAA decisionmaker on appeal.

(a) General. Any party may petition the FAA decisionmaker to reconsider or modify a final decision and order issued by the FAA decisionmaker on appeal from an initial decision. A party shall file a petition to reconsider or modify with the FAA decisionmaker not later than 30 days after service of the FAA decisionmaker's final decision and order on appeal and shall serve a copy of the petition on each party. The FAA decisionmaker will not reconsider or modify an initial decision and order issued by an administrative law judge

that has not been appealed by any party to the FAA decisionmaker.

- (b) Form and number of copies. A party shall file a petition to reconsider or modify, in writing, with the FAA decisionmaker. The party shall file the original petition with the FAA decisionmaker and shall serve a copy of the petition on each party.
- (c) Contents. A party shall state briefly and specifically the alleged errors in the final decision and order on appeal, the relief sought by the party, and the grounds that support, the petition to reconsider or modify.
- (1) If the petition is based, in whole or in part, on allegations regarding the consequences of the FAA decision-maker's decision, the party shall describe these allegations and shall describe, and support, the basis for the allegations.
- (2) If the petition is based, in whole or in part, on new material not previously raised in the proceedings, the party shall set forth the new material and include affidavits of prospective witnesses and authenticated documents that would be introduced in support of the new material. The party shall explain, in detail, why the new material was not discovered through due diligence prior to the hearing.
- (d) Repetitious and frivolous petitions. The FAA decisionmaker will not consider repetitious or frivolous petitions. The FAA decisionmaker may summarily dismiss repetitious or frivolous petitions to reconsider or modify.
- (e) Reply petitions. Any other party may reply to a petition to reconsider or modify, not later than 10 days after service of the petition on that party, by filing a reply with the FAA decisionmaker. A party shall serve a copy of the reply on each party.
- (f) Effect of filing petition. Unless otherwise ordered by the FAA decision-maker, filing of a petition pursuant to this section will not stay or delay the effective date of the FAA decision-maker's final decision and order on appeal and shall not toll the time allowed for judicial review.
- (g) FAA decisionmaker's decision on petition. The FAA decisionmaker has sole discretion to grant or deny a petition to reconsider or modify. The FAA decisionmaker will grant or deny a petition

to reconsider or modify within a reasonable time after receipt of the petition or receipt of the reply petition, if any. The FAA decisionmaker may affirm, modify, or reverse the final decision and order on appeal, or may remand the case for any proceedings that the FAA decisionmaker determines may be necessary.

[Amdt. 13–21, 55 FR 27575, July 3, 1990; 55 FR 29293, July 18, 1990; Amdt. 13–23, 55 FR 45983, Oct. 31, 1990]

§ 13.235 Judicial review of a final decision and order.

- (a) In cases under the Federal aviation statute, a party may seek judicial review of a final decision and order of the Administrator, as provided in 49 U.S.C. 46110(a), and, as applicable, in 49 U.S.C. 46301(d)(7)(D)(iii), 46301(g), or 47532.
- (b) In cases under the Federal hazardous materials transportation statute, a party may seek judicial review of a final decision and order of the Administrator, as provided in 49 U.S.C. 5127
- (c) A party seeking judicial review of a final order issued by the Administrator may file a petition for review in the United States Court of Appeals for the District of Columbia Circuit or in the United States Court of Appeals for the circuit in which the party resides or has its principal place of business.
- (d) The party must file the petition for review no later than 60 days after service of the Administrator's final decision and order.

[Doc. No. FAA–2006–26477, 71 FR 70465, Dec. 5, 20061

Subpart H—Civil Monetary Penalty Inflation Adjustment

SOURCE: Docket No. 28762, 61 FR 67445, Dec. 20, 1996, unless otherwise noted.

§13.301 Scope and purpose.

- (a) This subpart sets out the current adjusted maximum civil monetary penalties or range of minimum and maximum civil monetary penalties for each statutory civil penalty subject to the FAA's jurisdiction under title 49 of the U.S. Code. These penalties have been adjusted for inflation in conformity with the Federal Civil Penalties Inflation Adjustment Act of 1990, 28 U.S.C. 2461 (note), as amended by the Debt Collection Improvement Act of 1996, Public Law 104-134, April 26, 1996, and the Federal Civil Penalties Inflation Adjustment Act Improvements Act of 2015, Public Law 114-74, November 2, 2015, in order to maintain the deterrent effect of civil monetary penalties and to promote compliance with the law.
- (b) Each adjustment to the maximum civil monetary penalty or the range of minimum and maximum civil monetary penalties, as applicable, made in accordance with this subpart applies prospectively from the date it becomes effective to actions initiated under this part, notwithstanding references to a specific maximum civil monetary penalty or range of minimum and maximum civil monetary penalties contained elsewhere in this part.
- (c) Minimum and maximum civil monetary penalties within the jurisdiction of the FAA are as follows:

TABLE OF MINIMUM AND MAXIMUM CIVIL MONETARY PENALTY AMOUNTS FOR CERTAIN VIOLATIONS OCCURRING ON OR AFTER AUGUST 5, 2016

			•		
United States Code citation	Civil monetary penalty description	Minimum penalty amount	New or ad- justed min- imum penalty amount	Maximum penalty amount when last established or adjusted by Congress	New or adjusted maximum penalty amount
49 U.S.C. 5123(a), subparagraph (1).	Violation of hazardous materials transportation law.	Deleted 7/6/ 2012.	N/A	\$75,000 per violation, adjusted 7/6/2012.	\$77,114.
49 U.S.C. 5123(a), subparagraph (2).	Violation of hazardous materials transportation law resulting in death, serious illness, severe injury, or substantial property destruction.	Deleted 7/6/ 2012.	N/A	\$175,000 per violation, adjusted 7/6/2012.	\$179,933.
49 U.S.C. 5123(a), subparagraph (3).	Violation of hazardous materials transportation law relating to training.	\$450 per violation, adjusted.	\$463	\$75,000 per violation, adjusted 7/6/2012.	\$77,114.

Table of Minimum and Maximum Civil Monetary Penalty Amounts for Certain Violations Occurring on or After August 5, 2016—Continued

United States Code citation	Civil monetary penalty description	Minimum penalty amount	New or ad- justed min- imum penalty amount	Maximum penalty amount when last established or adjusted by Congress	New or ad- justed max- imum pen- alty amount
49 U.S.C. 46301(a)(1).	Violation by a person other than an individual or small business concern under 49 U.S.C.	N/A	N/A	\$25,000 per violation, established 12/12/2003.	\$32,140.
49 U.S.C. 46301(a)(1).	46301(a)(1)(A) or (B). Violation by an airman serving as an airman under 49 U.S.C. 46301(a)(1)(A) or (B) (but not covered by 46301(a)(5)(A) or (B).	N/A	N/A	\$1,100 per violation, adjusted 12/12/2003.	\$1,414.
49 U.S.C. 46301(a)(1).	Violation by an individual or small business concern under 49 U.S.C. 46301(a)(1)(A) or (B) (but not covered in 49 U.S.C. 46301(a)(5)).	N/A	N/A	\$1,100 per violation, adjusted 12/12/2003.	\$1,414.
49 U.S.C. 46301(a)(3).	Violation of 49 U.S.C. 47107(b) (or any assurance made under such section) or 49 U.S.C. 47133.	N/A	N/A	Increase above otherwise applicable maximum amount not to exceed 3 times the amount of revenues that are used in violation of such section.	No change.
49 U.S.C. 46301(a)(5)(A).	Violation by an individual or small business concern (except an airman serving as an airman) under 49 U.S.C. 46301(a)(5)(A)(i) or (ii).	N/A	N/A	\$10,000 per violation, established 12/12/2003.	\$12,856.
49 U.S.C. 46301(a)(5)(B)(i).	Violation by an individual or small business concern related to the transportation of hazardous materials.	N/A	N/A	\$10,000 per violation, established 12/12/2003.	\$12,856.
49 U.S.C. 46301(a)(5)(B)(ii).	Violation by an individual or small business concern related to the registration or recordation under 49 U.S.C. chapter 441, of an aircraft not used to provide air transportation.	N/A	N/A	\$10,000 per violation, established 12/12/2003.	\$12,856.
49 U.S.C. 46301(a)(5)(B)(iii).	Violation by an individual or small business concern of 49 U.S.C. 44718(d), relating to limitation on construction or establishment of landfills.	N/A	N/A	\$10,000 per violation, established 12/12/2003.	\$12,856.
49 U.S.C. 46301(a)(5)(B)(iv).	Violation by an individual or small business concern of 49 U.S.C. 44725, relating to the safe dis- posal of life-limited aircraft parts.	N/A	N/A	\$10,000 per violation, established 12/12/2003.	\$12,856.
49 U.S.C. 46301(b)	Tampering with a smoke alarm device.	N/A	N/A	\$2,000 per violation, established 12/22/1987.	\$4,126.
49 U.S.C. 46302	Knowingly providing false infor- mation about alleged violation involving the special aircraft ju- risdiction of the United States.	N/A	N/A	\$10,000 per violation, established 10/12/1984.	\$22,587.
49 U.S.C. 46318	Interference with cabin or flight crew.	N/A	N/A	\$25,000, established 4/5/ 2000.	\$34,172.
49 U.S.C. 46319	Permanent closure of an airport without providing sufficient notice.	N/A	N/A	\$10,000 per day, established 12/12/2003.	\$12,856.
49 U.S.C. 47531	Violation of 49 U.S.C. 47528– 47530, relating to the prohibi- tion of operating certain aircraft not complying with stage 3 noise levels.	N/A	N/A	See 49 U.S.C. 46301(a)(1)(A) and (a)(5), above.	No change.

[Docket No. 28762, 61 FR 67445, Dec. 20, 1996, as amended by Docket No. FAA–2016–7004, Amdt. 13–38, 81 FR 43467, July 5, 2016; 81 FR 51080, Aug. 5, 2016]

Subpart I—Flight Operational Quality Assurance Programs

§ 13.401 Flight Operational Quality Assurance Program: Prohibition against use of data for enforcement purposes.

- (a) Applicability. This section applies to any operator of an aircraft who operates such aircraft under an approved Flight Operational Quality Assurance (FOQA) program.
- (b) Definitions. For the purpose of this section, the terms—
- (1) Flight Operational Quality Assurance (FOQA) program means an FAA-approved program for the routine collection and analysis of digital flight data gathered during aircraft operations, including data currently collected pursuant to existing regulatory provisions, when such data is included in an approved FOQA program.
- (2) FOQA data means any digital flight data that has been collected from an individual aircraft pursuant to an FAA-approved FOQA program, regardless of the electronic format of that data.
- (3) Aggregate FOQA data means the summary statistical indices that are associated with FOQA event categories, based on an analysis of FOQA data from multiple aircraft operations.
- (c) Requirements. In order for paragraph (e) of this section to apply, the operator must submit, maintain, and adhere to a FOQA Implementation and Operation Plan that is approved by the Administrator and which contains the following elements:
- (1) A description of the operator's plan for collecting and analyzing flight recorded data from line operations on a routine basis, including identification of the data to be collected;
- (2) Procedures for taking corrective action that analysis of the data indicates is necessary in the interest of safety;
- (3) Procedures for providing the FAA with aggregate FOQA data;
- (4) Procedures for informing the FAA as to any corrective action being undertaken pursuant to paragraph (c)(2) of this section.
- (d) Submission of aggregate data. The operator will provide the FAA with ag-

gregate FOQA data in a form and manner acceptable to the Administrator.

- (e) Enforcement. Except for criminal or deliberate acts, the Administrator will not use an operator's FOQA data or aggregate FOQA data in an enforcement action against that operator or its employees when such FOQA data or aggregate FOQA data is obtained from a FOQA program that is approved by the Administrator
- (f) Disclosure. FOQA data and aggregate FOQA data, if submitted in accordance with an order designating the information as protected under part 193 of this chapter, will be afforded the nondisclosure protections of part 193 of this chapter.
- (g) Withdrawal of program approval. The Administrator may withdraw approval of a previously approved FOQA program for failure to comply with the requirements of this chapter. Grounds for withdrawal of approval may include, but are not limited to—
- (1) Failure to implement corrective action that analysis of available FOQA data indicates is necessary in the interest of safety; or
- (2) Failure to correct a continuing pattern of violations following notice by the agency; or also
- (3) Willful misconduct or willful violation of the FAA regulations in this chapter.

[Doc. No. FAA-2000-7554, 66 FR 55048, Oct. 31, 2001; Amdt. 13-30, 67 FR 31401, May 9, 2002]

PART 14—RULES IMPLEMENTING THE EQUAL ACCESS TO JUSTICE ACT OF 1980

Subpart A—General Provisions

Sec.

14.01 Purpose of these rules.

14.02 Proceedings covered.

14.03 Eligibility of applicants.

14.04 $\,$ Standards for awards.

14.05 Allowance fees and expenses.

Subpart B—Information Required From Applicants

14.10 Contents of application.

14.11 Net worth exhibit.

14.12 Documentation of fees and expenses.

Subpart C—Procedures for Considering Applications

- 14.20 When an application may be filed.
- 14.21 Filing and service of documents.
- 14.22 Answer to application.
- 14.23 Reply.
- 14.24 Comments by other parties.
- 14.25 Settlement.
- 14.26 Further proceedings.
- 14.27 Decision.
- 14.28 Review by FAA decisionmaker.
- 14.29 Judicial review.
- 14.30 Payment of award.

AUTHORITY: 5 U.S.C. 504; 49 U.S.C. 106(f), 40113, 46104 and 47122.

SOURCE: Docket No. 25958, 54 FR 46199, Nov. 1, 1989, unless otherwise noted.

Subpart A—General Provisions

§14.01 Purpose of these rules.

The Equal Access to Justice Act, 5 U.S.C. 504 (the Act), provides for the award of attorney fees and other expenses to eligible individuals and entities who are parties to certain administrative proceedings (adversary adjudications) before the Federal Aviation Administration (FAA). An eligible party may receive an award when it prevails over the FAA, unless the agency's position in the proceeding was substantially justified or special circumstances make an award unjust. The rules in this part describe the parties eligible for awards and the proceedings that are covered. They also explain how to apply for awards, and the procedures and standards that the FAA Decisionmaker will use to make them. As used hereinafter, the term "agency" applies to the FAA.

§14.02 Proceedings covered.

(a) The Act applies to certain adversary adjudications conducted by the FAA under 49 CFR part 17 and the Acquisition Management System (AMS). These are adjudications under 5 U.S.C. 554, in which the position of the FAA is represented by an attorney or other representative who enters an appearance and participates in the proceeding. This subpart applies to proceedings under 49 U.S.C. 46301, 46302, and 46303 and to the Default Adjudicative Process under part 17 of this chapter and the AMS.

- (b) If a proceeding includes both matters covered by the Act and matters specifically excluded from coverage, any award made will include only fees and expenses related to covered issues.
- (c) Fees and other expenses may not be awarded to a party for any portion of the adversary adjudication in which such party has unreasonably protracted the proceedings.

[54 FR 46199, Nov. 1, 1989, as amended by Amdt. 14-03, 64 FR 32935, June 18, 1999]

§ 14.03 Eligibility of applicants.

- (a) To be eligible for an award of attorney fees and other expenses under the Act, the applicant must be a party to the adversary adjudication for which it seeks an award. The term "party" is defined in 5 U.S.C. 504(b)(1)(B) and 5 U.S.C. 551(3). The applicant must show that it meets all conditions or eligibility set out in this subpart.
- (b) The types of eligible applicants are as follows:
- (1) An individual with a net worth of not more than \$2 million at the time the adversary adjudication was initi-
- (2) The sole owner of an unincorporated business who has a net worth of not more than \$7 million, including both personal and business interests, and not more than 500 employees at the time the adversary adjudication was initiated;
- (3) A charitable or other tax-exempt organization described in section 501(c)(3) of the Internal Revenue Code (26 U.S.C. 501(c)(3)) with not more than 500 employees at the time the adversary adjudication was initiated; and
- (4) A cooperative association as defined in section 15(a) of the Agricultural Marketing Act (12 U.S.C. 1141j(a)) with not more than 500 employees at the time the adversary adjudication was initiated; and
- (5) Any other partnership, corporation, association, or public or private organization with a net worth of not more than \$7 million and not more than 500 employees at the time the adversary adjudication was initiated.
- (c) For the purpose of eligibility, the net worth and number of employees of an applicant shall be determined as of the date the proceeding was initiated.

§ 14.04

- (d) An applicant who owns an unincorporated business will be considered an "individual" rather than a "sole owner of an unincorporated business" if the issues on which the applicant prevails are related primarily to personal interests rather than to business interest.
- (e) The employees of an applicant include all persons who regularly perform services for remuneration for the applicant, under the applicant's direction and control. Part-time employees shall be included on a proportional basis.
- (f) The net worth and number of employees of the applicant and all of its affiliates shall be aggregated to determine eligibility. Any individual, corporation, or other entity that directly or indirectly controls or owns a majority of the voting shares or other interest of the applicant, or any corporation or other entity of which the applicant directly or indirectly owns or controls a majority of the voting shares or other interest, will be considered an affiliate for purposes of this part, unless the ALJ or adjudicative officer determines that such treatment would be unjust and contrary to the purposes of the Act in light of the actual relationship between the affiliated entities. In addition, the ALJ or adjudicative officer may determine that financial relationships of the applicant, other than those described in this paragraph, constitute special circumstances that would make an award unjust.
- (g) An applicant that participates in a proceeding primarily on behalf of one or more other persons or entities that would be ineligible if not itself eligible for an award.

[54 FR 46199, Nov. 1, 1989, as amended by Amdt. 14-03, 64 FR 32935, June 18, 1999]

§14.04 Standards for awards.

(a) A prevailing applicant may receive an award for attorney fees and other expenses incurred in connection with a proceeding, or in a significant and discrete substantive portion of the proceeding, unless the position of the agency over which the applicant has prevailed was substantially justified. Whether or not the position of the FAA was substantially justified shall be determined on the basis of the record (in-

cluding the record with respect to the action or failure to act by the agency upon which the civil action is based) which was made in the civil action for which fees and other expenses are sought. The burden of proof that an award should not be made to an eligible prevailing applicant is on the agency counsel, who may avoid an award by showing that the agency's position was reasonable in law and fact.

(b) An award will be reduced or denied if the applicant has unduly or unreasonably protracted the proceeding or if special circumstances make the award sought unjust.

§ 14.05 Allowance fees and expenses.

- (a) Awards will be based on rates customarily charged by persons engaged in the business of acting as attorneys, agents, and expert witnesses, even if the services were made available without charge or at a reduced rate to the applicant.
- (b) No award for the fee of an attorney or agent under this part may exceed \$125 per hour, or such rate as prescribed by 5 U.S.C. 504. No award to compensate an expert witness may exceed the highest rate at which the agency pays expert witnesses. However, an award may also include the reasonable expenses of the attorney, agent, or witness as a separate item, if the attorney, agent, or witness ordinarily charges clients separately for such expenses.
- (c) In determining the reasonableness of the fee sought for an attorney, agent, or expert witness, the ALJ or adjudicative officer shall consider the following:
- (1) If the attorney, agent, or witness is in private practice, his or her customary fee for similar services, or if an employee of the applicant, the fully allocated cost of the services;
- (2) The prevailing rate for similar services in the community in which the attorney, agent, or witness ordinarily performs services:
- (3) The time actually spent in the representation of the applicant;
- (4) The time reasonably spent in light of the difficulty or complexity of the issues in the proceeding; and
- (5) Such other factors as may bear on the value of the services provided.

- (d) The reasonable cost of any study, analysis, engineering report, test, project, or similar matter prepared on behalf of a party may be awarded, to the extent that the charge for the service does not exceed the prevailing rate for similar services, and the study or other matter was necessary for preparation of the applicant's case.
- (e) Fees may be awarded only for work performed after the issuance of a complaint, or in the Default Adjudicative Process for a protest or contract dispute under part 17 of this chapter and the AMS.

[Amdt. 13–18, 53 FR 34655, Sept. 7, 1988, as amended by Amdt. 14–1, 55 FR 15131, Apr. 20, 1990; Amdt. 14–03, 64 FR 32935, June 18, 1999]

Subpart B—Information Required From Applicants

§14.10 Contents of application.

- (a) An application for an award of fees and expenses under the Act shall identify the applicant and the proceeding for which an award is sought. The application shall show that the applicant has prevailed and identify the position of the agency in the proceeding that the applicant alleges was not substantially justified. Unless the applicant is an individual, the application shall also state the number of employees of the applicant and describe briefly the type and purpose of its organization or business.
- (b) The application shall also include a statement that the applicant's net worth does not exceed \$2 million (if an individual) or \$7 million (for all other applicants, including their affiliates) at the time the adversary adjudication was initiated. However, an applicant may omit this statement if:
- (1) It attaches a copy of a ruling by the Internal Revenue Service that it qualifies as an organization described in section 501(c)(3) of the Internal Revenue Code (26 U.S.C. 501(c)(3)), or in the case of a tax-exempt organization not required to obtain a ruling from the Internal Revenue Service on its exempt status, a statement that describes the basis for the applicant's belief that it qualifies under such section; or
- (2) It states that it is a cooperative association as defined in section 15(a)

- of the Agricultural Marketing Act (12 U.S.C. 1141j(a)).
- (c) The application shall state the amount of fees and expenses for which an award is sought.
- (d) The application may also include any other matters that the applicant wishes this agency to consider in determining whether and in what amount an award should be made.
- (e) The application shall be signed by the applicant or an authorized officer or attorney for the applicant. It shall also contain or be accompanied by a written verification under oath or under penalty of perjury that the information provided in the application is true and correct.
- (f) If the applicant is a partnership, corporation, association, organization, or sole owner of an unincorporated business, the application shall state that the applicant did not have more than 500 employees at the time the adversary adjudication was initiated, giving the number of its employees and describing briefly the type and purpose of its organization or business.

§14.11 Net worth exhibit.

- (a) Each applicant except a qualified tax-exempt organization or cooperative association must provide with its application a detailed exhibit showing the net worth of the applicant and any affiliates when the proceeding was initiated. If any individual, corporation, or other entity directly or indirectly controls or owns a majority of the voting shares or other interest of the applicant, or if the applicant directly or indirectly owns or controls a majority of the voting shares or other interest of any corporation or other entity, the exhibit must include a showing of the net worth of all such affiliates or of the applicant including the affiliates. The exhibit may be in any form convenient to the applicant that provides full disclosure of the applicant's and its affiliates' assets and liabilities and is sufficient to determine whether the applicant qualifies under the standards in this part. The administrative law judge may require an applicant to file additional information to determine the eligibility for an award.
- (b) The net worth exhibit shall describe any transfers of assets from, or

§ 14.12

obligations incurred by, the applicant or any affiliate, occurring in the oneyear period prior to the date on which the proceeding was initiated, that reduced the net worth of the applicant and its affiliates below the applicable net worth ceiling. If there were no such transactions, the applicant shall so state.

- (c) Ordinarily, the net worth exhibit will be included in the public record of the proceeding. However, an applicant that objects to public disclosure of the net worth exhibit, or any part of it, may submit that portion of the exhibit directly to the ALJ or adjudicative officer in a sealed envelope labeled "Confidential Financial Information," accompanied by a motion to withhold the information.
- (1) The motion shall describe the information sought to be withheld and explain, in detail, why it should be exempt under applicable law or regulation, why public disclosure would adversely affect the applicant, and why disclosure is not required in the public interest.
- (2) The net worth exhibit shall be served on the FAA counsel, but need not be served on any other party to the proceeding.
- (3) If the ALJ or adjudicative officer finds that the net worth exhibit, or any part of it, should not be withheld from disclosure, it shall be placed in the public record of the proceeding. Otherwise, any request to inspect or copy the exhibit shall be disposed of in accordance with the FAA's established procedures.

[54 FR 46199, Nov. 1, 1989, as amended by Amdt. 14–03, 64 FR 32935, June 18, 1999]

§ 14.12 Documentation of fees and expenses.

The application shall be accompanied by full documentation of the fees and expenses, including the cost of any study, analysis, engineering report, test, project or similar matter, for which an award is sought. A separate itemized statement shall be submitted for each professional firm or individual whose services are covered by the application, showing the hours spent in connection with the proceedings by each individual, a description of the specific services performed, the rate at

which each fee has been computed, any expenses for which reimbursement is sought, the total amount claimed, and the total amount paid or payable by the applicant or by any other person or entity for the services provided. The administrative law judge may require the applicant to provide vouchers, receipts, or other substantiation for any expenses claimed.

Subpart C—Procedures for Considering Applications

§ 14.20 When an application may be filed.

- (a) An application may be filed whenever the applicant has prevailed in the proceeding, but in no case later than 30 days after the FAA Decisionmaker's final disposition of the proceeding, or service of the order of the Administrator in a proceeding under the AMS.
- (b) If review or reconsideration is sought or taken of a decision to which an applicant believes it has prevailed, proceedings for the award of fees shall be stayed pending final disposition of the underlying controversy.
- (c) For purposes of this part, final disposition means the later of:
- (1) Under part 17 of this chapter and the AMS, the date on which the order of the Administrator is served;
- (2) The date on which an unappealed initial decision becomes administratively final;
- (3) Issuance of an order disposing of any petitions for reconsideration of the FAA Decisionmaker's final order in the proceeding;
- (4) If no petition for reconsideration is filed, the last date on which such a petition could have been filed; or
- (5) Issuance of a final order or any other final resolution of a proceeding, such as a settlement or voluntary dismissal, which is not subject to a petition for reconsideration.

[54 FR 46199, Nov. 1, 1989, as amended by Amdt. 14–03, 64 FR 32936, June 18, 1999]

§14.21 Filing and service of documents.

Any application for an award or other pleading or document related to an application shall be filed and served on all parties to the proceeding in the same manner as other pleadings in the proceeding, except as provided in §14.11(b) for confidential financial information. Where the proceeding was held under part 17 of this chapter and the AMS, the application shall be filed with the FAA's attorney and with the Office of Dispute Resolution for Acquisition.

[Doc. No. FAA-1998-4379, 64 FR 32936, June 18, 19991

§14.22 Answer to application.

(a) Within 30 days after service of an application, counsel representing the agency against which an award is sought may file an answer to the application. Unless agency counsel requests an extension of time for filing or files a statement of intent to negotiate under paragraph (b) of the section, failure to file an answer within the 30-day period may be treated as a consent to the award requested.

(b) If the FAA's counsel and the applicant believe that the issues in the fee application can be settled, they may jointly file a statement of their intent to negotiate a settlement. The filing of this statement shall extend the time for filing an answer for an additional 30 days, and further extensions may be granted by the ALJ or adjudicative officer upon request by the FAA's counsel and the applicant.

(c) The answer shall explain in detail any objections to the award requested and identify the facts relied on in support of agency counsel's position. If the answer is based on any alleged facts not already in the record of the proceeding, agency counsel shall include with the answer either supporting affidavits or a request for further proceedings under §14.26.

[54 FR 46199, Nov. 1, 1989, as amended by Amdt. 14-03, 64 FR 32936, June 18, 1999]

$\S 14.23$ Reply.

Within 15 days after service of an answer, the applicant may file a reply. If the reply is based on any alleged facts not already in the record of the proceeding, the applicant shall include with the reply either supporting affidavits or a request for further proceedings under §14.26.

§ 14.24 Comments by other parties.

Any party to a proceeding other than the applicant and the FAA's counsel may file comments on an application within 30 days after it is served, or on an answer within 15 days after it is served. A commenting party may not participate further in proceedings on the application unless the ALJ or adjudicative officer determines that the public interest requires such participation in order to permit full exploration of matters raised in the comments.

[Doc. No. FAA–1998–4379, 64 FR 32936, June 18, 1999]

§ 14.25 Settlement.

The applicant and agency counsel may agree on a proposed settlement of the award before final action on the application, either in connection with a settlement of the underlying proceeding, or after the underlying proceeding has been concluded. If a prevailing party and agency counsel agree on a proposed settlement of an award before an application has been filed, the application shall be filed with the proposed settlement.

§14.26 Further proceedings.

(a) Ordinarily the determination of an award will be made on the basis of the written record; however, on request of either the applicant or agency counsel, or on his or her own initiative, the ALJ or adjudicative officer assigned to the matter may order further proceedings, such as an informal conference, oral argument, additional written submissions, or an evidentiary hearing. Such further proceedings shall be held only when necessary for full and fair resolution of the issues arising from the application and shall be conducted as promptly as possible.

(b) A request that the administrative law judge order further proceedings under this section shall specifically identify the information sought or the disputed issues and shall explain why the additional proceedings are necessary to resolve the issues.

[54 FR 46199, Nov. 1, 1989, as amended by Amdt. 14–03, 64 FR 32936, June 18, 1999]

§ 14.27

§ 14.27 Decision.

(a) The ALJ shall issue an initial decision on the application within 60 days after completion of proceedings on the application.

(b) An adjudicative officer in a proceeding under part 17 of this chapter and the AMS shall prepare a findings and recommendations for the Office of Dispute Resolution for Acquisition.

(c) A decision under paragraph (a) or (b) of this section shall include written findings and conclusions on the applicant's eligibility and status as prevailing party and an explanation of the reasons for any difference between the amount requested and the amount awarded. The decision shall also include, if at issue, findings on whether the FAA's position was substantially justified, or whether special circumstances make an award unjust.

[Doc. No. FAA-1998-4379, 64 FR 32936, June 18, 1999]

§14.28 Review by FAA decisionmaker.

(a) In proceedings other than those under part 17 of this chapter and the AMS, either the applicant or the FAA counsel may seek review of the initial decision on the fee application in accordance with subpart G of part 13 of this chapter, specifically §13.233. Additionally, the FAA Decisionmaker may decide to review the decision on his/her own initiative. If neither the applicant nor the FAA's counsel seeks review within 30 days after the decision is issued, it shall become final. Whether to review a decision is a matter within the discretion of the FAA Decisionmaker. If review is taken, the FAA Decisionmaker will issue a final decision on the application or remand the application to the ALJ who issue the initial fee award determination for further proceedings.

(b) In proceedings under part 17 of this chapter and the AMS, the adjudicative officer shall prepare findings and recommendations for the Office of Dispute Resolution for Acquisition with recommendations as to whether or not an award should be made, the amount of the award, and the reasons therefor. The Office of Dispute Resolution for Acquisition shall submit a recommended order to the Administrator

after the completion of all submissions related to the EAJA application. Upon the Administrator's action, the order shall become final, and may be reviewed under 49 U.S.C. 46110.

[Doc. No. FAA-1998-4379, 64 FR 32936, June 18, 1999, as amended at 70 FR 8238, Feb. 18, 2005]

§14.29 Judicial review.

If an applicant is dissatisfied with the determination of fees and other expenses made under this subsection, pursuant 5 U.S.C. 504(c)(2), that applicant may, within thirty (30) days after the determination is made, appeal the determination to the court of the United States having jurisdiction to review the merits of the underlying decision of the FAA adversary adjudication. The court's determination on any appeal heard under this paragraph shall be based solely on the factual record made before the FAA. The court may modify the determination of fees and other expenses only if the court finds that the failure to make an award of fees and other expenses, or the calculation of the amount of the award, was unsupported by substantial evidence.

§ 14.30 Payment of award.

An applicant seeking payment of an award shall submit to the disbursing official of the FAA a copy of the FAA Decisionmaker's final decision granting the award, accompanied by a statement that the applicant will not seek review of the decision in the United States courts. Applications for award grants in cases involving the FAA shall be sent to: The Office of Accounting and Audit, AAA-1, Federal Aviation Administration, 800 Independence Avenue, SW., Washington, DC 20591. The agency will pay the amount awarded to the applicant within 60 days, unless judicial review of the award or of the underlying decision of the adversary adjudication has been sought by the applicant or any other party to the proceeding.

PART 15—ADMINISTRATIVE CLAIMS UNDER FEDERAL TORT CLAIMS ACT

Subpart A—General Procedures

Sec.

- 15.1 Scope of regulations.
- 15.3 Administrative claim, when presented; appropriate office.
- 15.5 Administrative claim, who may file.
- 15.7 Administrative claims; evidence and information to be submitted.
- 15.9 Investigation and examination.

Subpart B—Indemnification Under Section 1118 of the Federal Aviation Act of 1958

- 15.101 Applicability.
- 15.103 Exclusions.
- 15.105 Filing of requests for indemnification.
- 15.107 Notification requirements.
- 15.109 Settlements.
- 15.111 Conduct of litigation.
- 15.113 Indemnification agreements.
- 15.115 Payment.

AUTHORITY: 5 U.S.C. 301; 28 U.S.C. 2672, 2675; 49 U.S.C. 106(g), 40113, 44721.

Subpart A—General Procedures

SOURCE: Docket No. 25264, 52 FR 18171, May 13, 1987, unless otherwise noted.

§15.1 Scope of regulations.

(a) These regulations apply to claims asserted under the Federal Tort Claims Act, as amended, for money damages against the United States for injury to, or loss of property, or for personal injury or death, caused by the negligent or wrongful act or omission of an employee of the FAA acting within the scope of office or employment. The regulations in this part supplement the Attorney General's regulations in 28 CFR part 14, as amended. The regulations in 28 CFR part 14, as amended, and the regulations in this part apply to consideration by the FAA of administrative claims under the Federal Tort Claims Act.

§ 15.3 Administrative claim, when presented; appropriate office.

(a) A claim is deemed to have been presented when the FAA receives, at a place designated in paragraph (b) of this section, an executed Standard Form 95 or other written notification of an incident, accompanied by a claim

for money damages in a sum certain for injury to, or loss of, property or for personal injury or death, alleged to have occurred by reason of the incident. A claim which should have been presented to the FAA but which was mistakenly filed with another Federal agency, is deemed presented to the FAA on the date the claim is received by the FAA at a place designated in paragraph (b) of this section. A claim addressed to, or filed with, the FAA by mistake will be transferred to the appropriate Federal agency, if that agency can be determined, or returned to the claimant.

- (b) Claims shall be delivered or mailed to the Assistant Chief Counsel, Litigation Division, AGC-400, Federal Aviation Administration, 800 Independence Avenue, SW., Washington, DC 20591, or alternatively, may be mailed or delivered to the Regional Counsel in any of the FAA Regional Offices or the Assistant Chief Counsel, Europe, Africa, and Middle East Area Office.
- (c) Claim forms are available at each location listed in paragraph (b) of this section.
- (d) A claim presented in accordance with this section may be amended by the claimant at any time prior to final FAA action or prior to the exercise of the claimant's option, under 28 U.S.C. 2675(a), to deem the agency's failure to make a final disposition of his or her claim within 6 months after it was filed as a final denial. Each amendment to a claim shall be submitted in writing and signed by the claimant or the claimant's duly authorized agent or legal representative. Upon the timely filing of an amendment to a pending claim, the FAA has 6 months thereafter in which to make a final disposition of the claim as amended, and the claimant's option under 28 U.S.C. 2675(a) does not accrue until 6 months after the filing of the amendment.

[Doc. No. 18884, 44 FR 63723, Nov. 5, 1979, as amended by Amdt. 15–1, 54 FR 39290, Sept. 25, 1989; Amdt. 15–4, 62 FR 46866, Sept. 4, 1997]

§15.5 Administrative claim, who may file.

(a) A claim for injury to, or loss of, property may be presented by the owner of the property interest which is

§ 15.7

the subject of the claim or by the owner's duly authorized agent or legal representative.

- (b) A claim for personal injury may be presented by the injured person or that person's duly authorized agent or legal representative.
- (c) A claim based on death may be presented by the executor or administrator of the decedent's estate or by any other person legally entitled to assert such a claim under applicable State law.
- (d) A claim for loss wholly compensated by an insurer with the rights of a subrogee may be presented by the insurer. A claim for loss partially compensated by an insurer with the rights of a subrogee may be presented by the insurer or the insured individually, as their respective interest appear, or jointly. Whenever an insurer presents a claim asserting the rights of a subrogee, it shall present with its claim appropriate evidence that it has the rights of a subrogee.
- (e) A claim presented by an agent or legal representative shall be presented in the name of the claimant, be signed by the agent or legal representative, show the title or legal capacity of the person signing, and be accompanied by evidence of authority to present a claim on behalf of the claimant as agent, executor, administrator, parent, guardian, or other representative.

§ 15.7 Administrative claims; evidence and information to be submitted.

- (a) *Death*. In support of a claim based on death, the claimant may be required to submit the following evidence or information:
- (1) An authenticated death certificate or other competent evidence showing cause of death, date of death, and age of the decedent.
- (2) The decedent's employment or occupation at time of death, including monthly or yearly salary or earnings (if any), and the duration of last employment or occupation.
- (3) Full names, addresses, birth dates, kinship, and marital status of the decedent's survivors, including identification of those survivors who were dependent for support upon the decedent at the time of death.

- (4) Degree of support afforded by the decedent to each survivor dependent upon decedent for support at the time of death.
- (5) Decedent's general, physical, and mental conditions before death.
- (6) Itemized bills for medical and burial expenses incurred by reason of the incident causing death or itemized receipts of payment for such expenses.
- (7) If damages for pain and suffering prior to death are claimed, a physician's detailed statement specifying the injuries suffered, duration of pain and suffering, any drugs administered for pain, and the decedent's physical condition in the interval between injury and death.
- (8) Any other evidence or information which may have a bearing on either the responsibility of the United States for the death or the amount of damages claimed.
- (b) *Personal injury*. In support of a claim for personal injury, including pain and suffering, the claimant may be required to submit the following evidence or information:
- (1) A written report by the attending physician or dentist setting forth the nature and extent of the injuries, nature and extent of treatment, any degree of temporary or permanent disability, the prognosis, period of hospitalization, and any diminished earning capacity.
- (2) In addition to the report required by paragraph (b)(1) of this section, the claimant may be required to submit to a physical or mental examination by a physician employed by the FAA or another Federal agency. A copy of the report of the examining physician is made available to the claimant upon the claimant's written request if the claimant has, upon request, furnished the report required by paragraph (b)(1), and has made or agrees to make available to the FAA any other physician's reports previously or thereafter made on the physical or mental condition which is the subject matter of the
- (3) Itemized bills for medical, dental, and hospital expenses incurred or itemized receipts of payment for such expenses.
- (4) If the prognosis reveals the necessity for future treatment, a statement

of expected expenses for such treatment.

- (5) If a claim is made for loss of time from employment, a written statement from the claimant's employer showing actual time lost from employment, whether the claimant is a full or partime employee, and wages or salary actually lost.
- (6) If a claim is made for loss of income and the claimant is self-employed, documentary evidence showing the amount of earnings actually lost.
- (7) Any other evidence or information which may have a bearing on the responsibility of the United States for the personal injury or the damages claimed.
- (c) *Property damage*. In support of a claim for injury to or loss of property, real or personal, the claimant may be required to submit the following evidence or information:
- (1) Proof of ownership of the property interest which is the subject of the claim.
- (2) A detailed statement of the amount claimed with respect to each item of property.
- (3) An itemized receipt of payment for necessary repairs or itemized written estimates of the cost of such repairs.
- (4) A statement listing date of purchase, purchase price, and salvage value, where repair is not economical.
- (5) Any other evidence or information which may have a bearing on either the responsibility of the United States for the injury to or loss of property or the damages claimed.

§15.9 Investigation and examination.

The FAA may investigate a claim or conduct a physical examination of a claimant. The FAA may request any other Federal agency to investigate a claim or conduct a physical examination of a claimant and provide a report of the investigation or examination to the FAA.

Subpart B—Indemnification Under Section 1118 of the Federal Aviation Act of 1958

Source: Amdt. 15–2, 55 FR 18710, May 3, 1990, unless otherwise noted.

§15.101 Applicability.

This subpart prescribes procedural requirements for the indemnification of a publisher of aeronautical charts or maps under section 1118 of the Federal Aviation Act of 1958, as amended, when the publisher incurs liability as a result of publishing—

- (a) A chart or map accurately depicting a defective or deficient flight procedure or airway that was promulgated by the FAA; or
 - (b) Aeronautical data that—
- (1) Is visually displayed in the cockpit of an aircraft: and
- (2) When visually displayed, accurately depicts a defective or deficient flight procedure or airway promulgated by the FAA.

§15.103 Exclusions.

- A publisher that requests indemnification under this part will not be indemnified if—
- (a) The complaint filed against the publisher, or demand for payment against the publisher, first occurred before December 19, 1985;
- (b) The publisher does not negotiate a good faith settlement;
- (c) The publisher does not conduct a good faith defense;
- (d) The defective or deficient flight procedure or airway—
- (1) Was not promulgated by the FAA;
- (2) Was not accurately depicted on the publisher's chart or map;
- (3) Was not accurately displayed on a visual display in the cockpit, or
- (4) Was obviously defective or deficient;
- (e) The publisher does not give notice as required by §15.107 of this part and that failure is prejudicial to the Government; or
- (f) The publisher does not appeal a lower court's decision pursuant to a request by the Administrator under §15.111(d)(2) of this part.

§ 15.105 Filing of requests for indemnification.

- A request for indemnification under this part—
 - (a) May be filed by—
- (1) A publisher described in §15.101 of this part; or
- (2) The publisher's duly authorized agent or legal representative;

§ 15.107

- (b) Shall be filed with the Chief Counsel, Federal Aviation Administration, 800 Independence Avenue SW., Washington, DC 20591; and
- (c) Shall state the basis for the publisher's assertion that indemnification under this part is required.

§15.107 Notification requirements.

- A request for indemnification will not be considered by the FAA unless the following conditions are met:
- (a) The publisher must notify the Chief Counsel of the FAA, within the time limits prescribed in paragraph (b) or (c) of this section, of the publisher's first receipt of a demand for payment, or service of a complaint in any proceeding, federal or state, in which it appears that indemnification under this part may be required.
- (b) For each complaint filed, or demand for payment made, on or after December 19, 1985, and before June 4, 1990, the notice required by paragraph (a) of this section must be received by the FAA on or before July 2, 1990.
- (c) For each complaint filed, or demand for payment made, on or after June 4, 1990, the notice required by paragraph (a) of this section must be received by the FAA within 60 days after the day the publisher first receives the demand for payment or service of the complaint.
- (d) Within 5 days after the day a judgment is rendered against the publisher in any proceeding, or within 30 days of the denial of an appeal, whichever is later, the publisher must notify the FAA Chief Counsel that—
- (1) There is an adverse judgment against the publisher: and
- (2) The publisher has a claim for indemnification against the FAA arising out of that judgment.

§15.109 Settlements.

- (a) A publisher may not settle a claim with another party, for which the publisher has sought, or intends to seek, indemnification under this part, unless—
- (1) The publisher submits a copy of the proposed settlement, and a statement justifying the settlement, to the Chief Counsel of the FAA; and
- (2) The Administrator and where necessary, the appropriate official of the

- Department of Justice, approves the proposed settlement.
- (3) The publisher submits a signed release that clearly releases the United States from any further liability to the publisher and the claimant.
- (b) If the Administrator does not approve the proposed settlement, the Administrator will—
- (1) So notify the publisher by registered mail within 60 days of receipt of the proposed settlement; and
- (2) Explain why the request for indemnification was not approved.
- (c) If the Administrator approves the proposed settlement, the Administrator will so notify the publisher by registered mail within 60 days after the FAA's receipt of the proposed settlement.
- (d) If the Administrator does not have sufficient information to approve or disapprove the proposed settlement, the Administrator will request, within 60 days after receipt of the proposed settlement, the additional information needed to make a determination.

§15.111 Conduct of litigation.

- (a) If a lawsuit is filed against the publisher and the publisher has sought, or intends to seek, indemnification under this part, the publisher shall—
- (1) Give notice as required by \$15.107 of this part;
- (2) If requested by the United States—
- (i) Implead the United States as a third-party defendant in the action; and
- (ii) Arrange for the removal of the action to Federal Court;
- (3) Promptly provide any additional information requested by the United States; and
- (4) Cooperate with the United States in the defense of the lawsuit.
- (b) If the lawsuit filed against the publisher results in a proposed settlement, the publisher shall submit that proposed settlement to the FAA for approval in accordance with §15.109 of this part.
- (c) If the lawsuit filed against the publisher results in a judgment against the publisher and the publisher has sought, or intends to seek, indemnification under this part as a result of

the adverse judgment, the publisher shall—

- (1) Give notice to the FAA as required by §15.107(d) of this part;
- (2) Submit a copy of the trial court's decision to the FAA Chief Counsel not more than 5 business days after the adverse judgment is rendered; and
- (3) If an appeal is taken from the adverse judgment, submit a copy of the appellate decision to the FAA Chief Counsel not more than 30 days after that decision is rendered.
- (d) Within 60 days after receipt of the trial court's decision, the Administrator by registered mail will—
- (1) Notify the publisher that indemnification is required under this part;
- (2) Request that the publisher appeal the trial court's adverse decision; or
- (3) Notify the publisher that it is not entitled to indemnification under this part and briefly state the basis for the denial.

§15.113 Indemnification agreements.

- (a) Upon a finding of the Administrator that indemnification is required under this part, and after obtaining the concurrence of the United States Department of Justice, the FAA will promptly enter into an indemnification agreement providing for the payment of the costs specified in paragraph (c) of this section.
- (b) The indemnification agreement will be signed by the Chief Counsel and the publisher.
- (c) The FAA will indemnify the publisher for—
- (1) Compensatory damages awarded by the court against the publisher;
- (2) Reasonable costs and fees, including reasonable attorney fees at a rate not to exceed that permitted under the Equal Access to Justice Act (5 U.S.C. 504), and any postjudgment interest, if the publisher conducts a good faith defense, or pursues a good faith appeal, at the request, or with the concurrence, of the FAA.
- (d) Except as otherwise provided in this section, the FAA will not indemnify the publisher for—
 - (1) Punitive or exemplary damages;
- (2) Civil or criminal fines or any other litigation sanctions;
 - (3) Postjudgment interest;
 - (4) Costs;

- (5) Attorney fees; or
- (6) Other incidental expenses.
- (e) The indemnification agreement must provide that the Government will be subrogated to all claims or rights of the publisher, including third-party claims, cross-claims, and counterclaims.

§15.115 Payment.

After execution of the indemnification agreement, the FAA will submit the agreement to the United States Department of Justice and request payment, in accordance with the agreement, from the Judgment Fund.

PART 16—RULES OF PRACTICE FOR FEDERALLY-ASSISTED AIRPORT ENFORCEMENT PROCEEDINGS

Subpart A—General Provisions

Sec.

- 16.1 Applicability and description of part.
- 16.3 Definitions.
- 16.5 Separation of functions.

Subpart B—General Rules Applicable to Complaints, Proceedings Initiated by the FAA, and Appeals

- 16.11 General processes.
- 16.13 Filing of documents.
- 16.15 Service of documents on the parties and the agency.
- 16.17 Computation of time.
- 16.19 Motions.

Subpart C—Special Rules Applicable to Complaints

- 16.21 Pre-complaint resolution.
- 16.23 Pleadings.
- 16.25 Dismissals.
- 16.26 Motions to dismiss and motions for summary judgment.
- 16.27 Incomplete complaints.
- 16.29 Investigations.
- 16.31 Director's determinations after investigations.
- 16.33 Final decisions without hearing.
- 16.34 Consent orders.

Subpart D—Special Rules Applicable to Proceedings Initiated by the FAA

- 16.101 Basis for the initiation of agency action.
- 16.103 Notice of investigation.
- 16.105 Failure to resolve informally.

§ 16.1

Subpart E—Proposed Orders of Compliance

16.109 Orders terminating eligibility for grants, cease and desist orders, and other compliance orders.

Subpart F—Hearings

- 16.201 Notice and order of hearing.
- 16.202 Powers of a hearing officer.
- 16.203 Appearances, parties, and rights of parties.
- 16.207 Intervention and other participation.
- 16.209 Extension of time.
- 16.211 Prehearing conference.
- 16.213 Discovery.
- 16.215 Depositions.
- 16.217 Witnesses.
- 16.219 Subpoenas.
- 16.221 Witness fees.
- 16.223 Evidence.
- 16.225 Public disclosure of evidence.
- 16.227 Standard of proof.
- 16.229 Burden of proof.
- 16.231 Offer of proof.
- 16.233 Record.
- 16.235 Argument before the hearing officer.
- 16.237 Waiver of procedures.
- 16.241 Initial decisions, orders, and appeals.
- 16.243 Consent orders.
- 16.245 Associate Administrator review after a hearing.

Subpart G-Judicial Review

16.247 Judicial review of a final decision and order.

Subpart H-Ex Parte Communications

- 16.301 Prohibited ex parte communications.16.303 Procedures for handling ex parte communications.
- 16.305 Requirement to show cause and imposition of sanction.

AUTHORITY: 49 U.S.C. 106(g), 322, 1110, 1111, 1115, 1116, 1718(a) and (b), 1719, 1723, 1726, 1727, 40103(e), 40113, 40116, 44502(b), 46101, 46104, 47104, 47106(e), 47107, 47108, 47111(d), 47122, 47123–47125, 47133, 47151–47153, 48103.

SOURCE: Docket No. 27783, 61 FR 54004, Oct. 16, 1996, unless otherwise noted.

Subpart A—General Provisions

§ 16.1 Applicability and description of part.

(a) General. The provisions of this part govern all Federal Aviation Administration (FAA) proceedings involving Federally-assisted airports, except for complaints or requests for determination filed with the Secretary under 14 CFR part 302, whether the pro-

ceedings are instituted by order of the FAA or by filing a complaint with the FAA under the following authorities:

- (1) 49 U.S.C. 40103(e), prohibiting the grant of exclusive rights for the use of any landing area or air navigation facility on which Federal funds have been expended (formerly section 308 of the Federal Aviation Act of 1958, as amended).
- (2) Requirements of the Anti-Head Tax Act, 49 U.S.C. 40116.
- (3) The assurances and other Federal obligations contained in grant-in-aid agreements issued under the Federal Airport Act of 1946, 49 U.S.C. 1101 *et seq*. (repealed 1970).
- (4) The assurances and other Federal obligations contained in grant-in-aid agreements issued under the Airport and Airway Development Act of 1970, as amended, 49 U.S.C. 1701 et seq.
- (5) The assurances and other Federal obligations contained in grant-in-aid agreements issued under the Airport and Airway Improvement Act of 1982 (AAIA), as amended and recodified, 49 U.S.C. 47101 *et seq.*, specifically section 511(a), 49 U.S.C. 47107, and 49 U.S.C. 47133.
- (6) Section 505(d) of the Airport and Airway Improvement Act of 1982, and the requirements concerning civil rights and/or Disadvantaged Business Enterprise (DBE) issues contained in 49 U.S.C. 47107(e) and 49 U.S.C. 47113; 49 U.S.C. 47123; 49 U.S.C. 322, as amended; 49 CFR parts 23 and/or 26; and/or grant assurance 30 and/or grant assurance 37.
- (7) Obligations contained in property deeds for property transferred pursuant to section 16 of the Federal Airport Act (49 U.S.C. 1115), section 23 of the Airport and Airway Development Act (49 U.S.C. 1723), or section 516 of the Airport and Airway Improvement Act (49 U.S.C. 47125).
- (8) Obligations contained in property deeds for property transferred under the Surplus Property Act (49 U.S.C. 47151–47153).
- (b) Other agencies. Where a grant assurance concerns a statute, executive order, regulation, or other authority that provides an administrative process for the investigation or adjudication of complaints by a Federal agency other than the FAA, persons shall use the administrative process established

by those authorities. Where a grant assurance concerns a statute, executive order, regulation, or other authority that enables a Federal agency other than the FAA to investigate, adjudicate, and enforce compliance under those authorities on its own initiative, the FAA may defer to that Federal agency.

(c) Other enforcement. If a complaint or action initiated by the FAA involves a violation of the 49 U.S.C. subtitle VII or FAA regulations, except as specified in paragraphs (a)(1) and (a)(2) of this section, the FAA may take investigative and enforcement action under 14 CFR part 13, "Investigative and Enforcement Procedures."

(d) Effective date. This part applies to a complaint filed with the FAA and to an investigation initiated by the FAA on or after December 16, 1996.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56141, Sept. 12, 2013]

§ 16.3 Definitions.

Terms defined in the Acts are used as so defined. As used in this part:

Act means a statute listed in §16.1 and any regulation, agreement, or document of conveyance issued or made under that statute.

Administrator means the Administrator of the FAA.

Agency means the FAA.

Agency attorney means the Deputy Chief Counsel; the Assistant Chief Counsel and attorneys in the Airports/ Environmental Law Division of the Office of the Chief Counsel; the Assistant Chief Counsel and attorneys in an FAA region or center who represent the FAA during the investigation of a complaint or at a hearing on a complaint, and who prosecute on behalf of the FAA, as appropriate. An agency attorney shall not include the Chief Counsel; the Assistant Chief Counsel for Litigation, or any attorney on the staff of the Assistant Chief Counsel for Litigation, who advises the Associate Administrator regarding an initial decision of the hearing officer or any appeal to the Associate Administrator or who is supervised in that action by a person who provides such advice in an action covered by this part.

Agency employee means any employee of the FAA.

Associate Administrator means the FAA Associate Administrator for Airports or a designee. For the purposes of this part only, Associate Administrator also means the Assistant Administrator for Civil Rights or a designee for complaints that the FAA Associate Administrator for Airports transfers to the Assistant Administrator for Civil Rights.

Complainant means the person submitting a complaint.

Complaint means a written document meeting the requirements of this part and filed under this part:

(1) By a person directly and substantially affected by anything allegedly done or omitted to be done by any person in contravention of any provision of any Act, as defined in this section, as to matters within the jurisdiction of the Administrator, or

(2) By a person under 49 CFR 26.105(c) against a recipient of FAA funds alleged to have violated a provision of 49 CFR parts 23 and/or 26.

Decisional employee means the Administrator, Deputy Administrator, Associate Administrator, Director, hearing officer, or other FAA employee who is or who may reasonably be expected to be involved in the decisional process of the proceeding.

Director means the Director of the FAA Office of Airport Compliance and Management Analysis, or a designee. For the purposes of this part only, Director also means the Deputy Assistant Administrator for Civil Rights for complaints that the Director of the FAA Office of Airport Compliance and Management Analysis transfers to the Deputy Assistant Administrator for Civil Rights or designee.

Electronic filing means the process of sending electronic mail (email) to the FAA Part 16 Docket Clerk, with scanned documents attached, as a Portable Document Format (PDF) file.

Ex parte communication means an oral or written communication not on the public record with respect to which reasonable prior notice to all parties is not given, but it shall not include requests for status reports on any matter or proceeding covered by this part, or

§ 16.5

communications between FAA employees who participate as parties to a hearing pursuant to 16.203(b) of this part and other parties to a hearing.

Hearing officer means an attorney designated by the Deputy Chief Counsel in a hearing order to serve as a hearing officer in a hearing under this part. The following are not designated as hearing officers: the Chief Counsel and Deputy Chief Counsel; the Regional or Center Counsel and attorneys in the FAA region or center in which the noncompliance has allegedly occurred or is occurring; the Assistant Chief Counsel and attorneys in the Airports and Environmental Law Division of the FAA Office of the Chief Counsel; and the Assistant Chief Counsel and attorneys in the Litigation Division of the FAA Office of Chief Counsel.

Initial decision means a decision made by the hearing officer in a hearing under subpart F of this part.

Mail means U.S. first class mail; U.S. certified mail; and U.S. express mail. Unless otherwise noted, mail also means electronic mail containing PDF copies of pleadings or documents required herein.

Noncompliance means anything done or omitted to be done by any person in contravention of any provision of any Act, as defined in this section, as to matters within the jurisdiction of the Administrator.

Party means the complainant(s) and the respondent(s) named in the complaint and, after an initial determination providing an opportunity for hearing is issued under §16.31 and subpart E of this part, the agency.

Person in addition to its meaning under 49 U.S.C. 40102(a)(33), includes a public agency as defined in 49 U.S.C. 47102(a)(15).

Personal delivery means same-day hand delivery or overnight express delivery service.

Respondent means any person named in a complaint as a person responsible for noncompliance.

Sponsor means:

(1) Any public agency which, either individually or jointly with one or more other public agencies, has received Federal financial assistance for airport development or planning under the Federal Airport Act, Airport and

Airway Development Act or Airport and Airway Improvement Act;

- (2) Any private owner of a public-use airport that has received financial assistance from the FAA for such airport; and
- (3) Any person to whom the Federal Government has conveyed property for airport purposes under section 13(g) of the Surplus Property Act of 1944, as amended.

Writing or written includes paper documents that are filed and/or served by mail, personal delivery, facsimile, or email (as attached PDF files).

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56141, Sept. 12, 2013]

§ 16.5 Separation of functions.

- (a) Proceedings under this part, including hearings under subpart F of this part, will be prosecuted by an agency attorney.
- (b) After issuance of an initial determination in which the FAA provides the opportunity for a hearing, an agency employee engaged in the performance of investigative or prosecutorial functions in a proceeding under this part will not, in that case or a factually related case, participate or give advice in an initial decision by the hearing officer, or a final decision by the Associate Administrator or designee on written appeal, and will not. except as counsel or as witness in the public proceedings, engage in any substantive communication regarding that case or a related case with the hearing officer, the Associate Administrator on written appeal, or agency employees advising those officials in that capacity.
- (c) The Chief Counsel, the Assistant Chief Counsel for Litigation, or an attorney on the staff of the Assistant Chief Counsel for Litigation advises the Associate Administrator regarding an initial decision, an appeal, or a final decision regarding any case brought under this part.

Subpart B—General Rules Applicable to Complaints, Proceedings Initiated by the FAA, and Appeals

§16.11 General processes.

- (a) Under the authority of 49 U.S.C. 40113 and 47121, the Director may conduct investigations, issue orders, and take such other actions as are necessary to fulfill the purposes of this part. This includes the extension of any time period prescribed, where necessary or appropriate for a fair and complete consideration of matters before the agency, prior to issuance of the Director's Determination.
- (b) Notwithstanding any other provision of this part, upon finding that circumstances require expedited handling of a particular case or controversy, the Director may issue an order directing any of the following prior to the issuance of the Director's Determination:
- (1) Shortening the time period for any action under this part consistent with due process:
- (2) If other adequate opportunity to respond to pleadings is available, eliminating the reply, rebuttal, or other actions prescribed by this part:
- (3) Designating alternative methods of service; or
- (4) Directing such other measures as may be required.
- (c) Other than those matters concerning a Corrective Action Plan, the jurisdiction of the Director terminates upon the issuance of the Director's Determination. All matters arising during the appeal period, such as requests for extension of time to make an appeal, will be addressed by the Associate Administrator.
- (d) The Director may transfer to the FAA Deputy Assistant Administrator for Civil Rights or Office of Civil Rights designee the authority to prepare and issue Director's Determinations pursuant to §16.31 for complaints alleging violations of section 505(d) of the Airport and Airway Improvement Act of 1982, and the requirements concerning civil rights and/or Disadvantaged Business Enterprise (DBE) issues contained in 49 U.S.C. 47107(e) and 49 U.S.C. 47113; 49 U.S.C. 47123; 49 U.S.C. 322, as amended; 49 CFR parts 23 and/or

26; and/or grant assurance 30 and/or grant assurance 37.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56142, Sept. 12, 2013]

§16.13 Filing of documents.

Except as otherwise provided in this part, documents shall be filed with the FAA during a proceeding under this part as follows:

- (a) Filing address. Documents filed under this Part shall be filed with the Office of the Chief Counsel, Attention: FAA Part 16 Docket Clerk, AGC-600, Federal Aviation Administration, 800 Independence Avenue SW., Washington, DC 20591. Documents to be filed with a hearing officer shall be filed at the address and in the manner stated in the hearing order.
- (b) Date and method of filing. Filing of any document shall be by personal delivery or mail as defined in this part, by facsimile (when confirmed by filing on the same date by one of the foregoing methods), or electronically as set forth in paragraph (h) of this section. Unless the date is shown to be inaccurate, documents filed with the FAA shall be deemed to be filed on the date of personal delivery, on the mailing date shown on the certificate of service, on the date shown on the postmark if there is no certificate of service, on the send date shown on the facsimile (provided filing has been confirmed through one of the foregoing methods), or on the mailing date shown by other evidence if there is no certificate of service and no postmark. Unless the date is shown to be inaccurate, documents filed electronically shall be deemed to be filed on the date shown on the certificate of service or, if none, the date of electronic transmission to the last party required to be served.
- (c) Number of copies. With the exception of electronic filing or unless otherwise specified, an executed original and three copies of each document shall be filed with the FAA Part 16 Docket Clerk. One of the three copies shall not be stapled, bound or holepunched. Copies need not be signed, but the name of the person signing the original shall be shown. If a hearing order has been issued in the case, one of the three copies shall be filed with

§ 16.13

the hearing officer unless otherwise prescribed by the hearing officer.

- (d) Form. Documents filed under this part shall:
 - (1) Be typewritten or legibly printed;(2) Include, in the case of docketed
- proceedings, the docket number of the proceeding on the front page; and
- (3) Be marked to identify personal, privileged or proprietary information. Decisions for the publication and release of these documents will be made in accordance with 5 U.S.C. 552 and 49 CFR part 7.
- (e) Signing of documents and other papers. The original of every document filed shall be signed by the person filing it or the person's duly authorized representative. The signature shall serve as a certification that the signer has read the document and, based on reasonable inquiry and to the best of the signer's knowledge, information, and belief, the document is—
 - (1) Consistent with this part;
- (2) Warranted by existing law or that a good faith argument exists for extension, modification, or reversal of existing law; and
- (3) Not interposed for any improper purpose, such as to harass or to cause unnecessary delay or needless increase in the cost of the administrative process.
- (f) Designation of person to receive service. The initial document filed by any person shall state on the first page the name, physical address, telephone number, facsimile number, if any, and email address, if filing electronically, of the person(s) to be served with documents in the proceeding. If any of these items change during the proceeding, the person shall promptly file notice of the change with the FAA Part 16 Docket Clerk and the hearing officer and shall serve the notice on all parties.
- (g) *Docket numbers*. Each submission identified as a complaint under this part by the submitting person will be assigned a docket number.
- (h) Electronic filing. (1) The initial complaint may be served electronically upon the respondent only if the respondent has previously agreed with the complainant in writing to participate in electronic filing. Documents may be filed under this Part electronically by sending an email containing

- (an) attachment(s) of (a) PDF file(s) of the required pleading to the FAA Docket Clerk, and the person designated in paragraph (h)(3) of this section.
- (2) The subject line of the email must contain the names of the complainant and respondent, and must contain the FAA docket number (if assigned). The size of each email must be less than 10 MB. Email attachments containing executable files (e.g., .exe and .vbs files) will not be accepted.
- (3) The email address at which the parties may file the documents described in this section is 9-AWA-AGC-Part-16@faa.gov. No acknowledgement or receipt will be provided by the FAA to parties using this method. A party filing electronically as described in this section must provide to the FAA Part 16 Docket Clerk and the opposing party an email address of the person designated by the party to receive pleadings.
- (4) By filing a pleading or document electronically as described in this section, a party waives the rights under this part for service by the opposing party and the FAA by methods other than email. If a party subsequently decides to "opt-out" of electronic filing, that party must so notify the FAA Part 16 Docket Clerk and the other party in writing, from which time the FAA and the parties will begin serving the opting-out party in accordance with §§16.13 and 16.15. This subsection only exempts the parties from the filand service requirements in §16.13(a) (with the exception that "Documents to be filed with a hearing officer shall be filed at the address and in the manner stated in the hearing order."), the method of filing requirements in §16.13(b), and the number of documents requirements in §16.13(c).
- (i) Internet accessibility of documents filed in the Hearing Docket. (1) Unless protected from public disclosure, all documents filed in the Hearing Docket are accessible through the Federal Docket Management System (FDMS): http://www.regulations.gov. To access a particular case file, use the FDMS number assigned to the case.
- (2) Determinations issued by the Director and Associate Administrator in

Federal Aviation Administration, DOT

Part 16 cases, indexes of decisions, contact information for the FAA Hearing Docket, the rules of practice, and other information are available on the FAA Office of Airports' Web site at: http://part16.airports.faa.gov/index.cfm.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56142, Sept. 12, 2013]

§16.15 Service of documents on the parties and the agency.

Except as otherwise provided in this part, documents shall be served as follows:

(a) Whom must be served. Copies of all documents filed with the FAA Part 16 Docket Clerk shall be served by the persons filing them on all parties to the proceeding. A certificate of service shall accompany all documents when they are tendered for filing and shall certify concurrent service on the FAA and all parties. Certificates of service shall be in substantially the following form:

I hereby certify that I have this day served the foregoing [name of document] on the following persons at the following addresses, facsimile numbers (if also served by facsimile), or email address (if served electronically in accordance with §16.13(h)), by [specify method of service]:

[list persons, addresses, facsimile numbers, email addresses (as applicable)]

Dated this __day of __, 20__. [signature], for [party]

- (b) Method of service. Except as otherwise agreed by the parties and, if applicable, the hearing officer, the method of service is the same as set forth in §16.13(b) for filing documents.
- (c) Where service shall be made. Service shall be made to the persons identified in accordance with §16.13(f). If no such person has been designated, service shall be made on the party.
- (d) Presumption of service. There shall be a presumption of lawful service—
- (1) When acknowledgment of receipt is by a person who customarily or in the ordinary course of business receives mail at the address of the party or of the person designated under §16.13(f);
- (2) When a properly addressed envelope, sent to the most current address submitted under §16.13(f), has been re-

turned as undeliverable, unclaimed, or refused: or

- (3) When the party serving the document electronically has a confirmation statement demonstrating that the email was properly sent to a party correctly addressed.
- (e) *Date of service*. The date of service shall be determined in the same manner as the filing date under §16.13(b).

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56143, Sept. 12, 2013]

§16.17 Computation of time.

This section applies to any period of time prescribed or allowed by this part, by notice or order of the hearing officer, or by an applicable statute.

- (a) The date of an act, event, or default, after which a designated time period begins to run, is not included in a computation of time under this part.
- (b) The last day of a time period is included in a computation of time unless it is a Saturday, Sunday, or legal holiday for the FAA, in which case, the time period runs until the end of the next day that is not a Saturday, Sunday, or legal holiday.
- (c) Whenever a party has the right or is required to do some act within a prescribed period after service of a document upon the party, and the document is served on the party by first class mail or certified mail, 5 days shall be added to the prescribed period.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56143, Sept. 12, 2013]

§ 16.19 Motions.

- (a) General. An application for an order or ruling not otherwise specifically provided for in this part shall be by motion. Unless otherwise ordered by the agency, the filing of a motion will not stay the date that any action is permitted or required by this part.
- (b) Form and contents. Unless made during a hearing, motions shall be made in writing, shall state with particularity the relief sought and the grounds for the relief sought, and shall be accompanied by affidavits or other evidence relied upon. Motions introduced during hearings may be made orally on the record, unless the hearing officer directs otherwise.

- (c) Answers to motions. Except as otherwise provided in this part, or except when a motion is made during a hearing, any party may file an answer in support of or in opposition to a motion. accompanied by affidavits or other evidence relied upon, provided that the answer to the motion is filed within 10 days after the motion has been served upon the person answering, or any other period set by the hearing officer. Where a motion is made during a hearing, the answer and the ruling thereon may be made at the hearing, or orally or in writing within the time set by the hearing officer.
- (d) Deferred actions on motions. A ruling on a motion made before the time set for the issuance of the Director's Determination may be deferred to and included with the Director's Determination.
- (e) Extension by motion. A party shall file a written motion for an extension of time not later than 3 business days before the document is due unless good cause for the late filing is shown. A party filing a motion for extension should attempt to obtain the concurrence of the opposing party. A party filing a written motion for an extension of time shall file the motion as required under §16.13, and serve a copy of the motion on all parties and the docket clerk as required under §16.15.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56143, Sept. 12, 2013]

Subpart C—Special Rules Applicable to Complaints

§ 16.21 Pre-complaint resolution.

(a) Except for those persons filing under 49 CFR 26.105(c), prior to filing a complaint under this part, a person directly and substantially affected by the alleged noncompliance shall initiate and engage in good faith efforts to resolve the disputed matter informally with those individuals or entities believed responsible for the noncompliance. These efforts at informal resolution may include, without limitation, at the parties' expense, mediation, arbitration, or the use of a dispute resolution board, or other form of third party assistance. The FAA Airports District Office, FAA Airports Field Of-

- fice, FAA Regional Airports Division responsible for administering financial assistance to the sponsor, or the FAA Office of Civil Rights will be available upon request to assist the parties with informal resolution.
- (b) Except for complaints filed under 49 CFR 26.105(c), a complaint will be dismissed under §16.27 unless the person or authorized representative filing the complaint certifies that:
- (1) The complainant has made substantial and reasonable good faith efforts to resolve the disputed matter informally prior to filing the complaint; and
- (2) There is no reasonable prospect for practical and timely resolution of the dispute.
- (c) The certification required under paragraph (b) of this section, shall include a brief description of the party's efforts to obtain informal resolution but shall not include information on monetary or other settlement offers made but not agreed upon in writing by all parties. Such efforts to resolve informally should be relatively recent and be demonstrated by pertinent documentation. There is no required form or process for informal resolution, but in each case the requirements to resolve the matter informally must meet the requirements of this paragraph.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56143, Sept. 12, 2013]

§16.23 Pleadings.

- (a) A person directly and substantially affected by any alleged noncompliance or a person qualified under 49 CFR 26.105(c) may file a complaint under this part. A person doing business with an airport and paying fees or rentals to the airport shall be considered directly and substantially affected by alleged revenue diversion as defined in 49 U.S.C. 47107(b).
- (b) Complaints filed under this part shall—
- (1) State the name and address of each person who is the subject of the complaint and, with respect to each person, the specific provisions of each Act that the complainant believes were violated;

- (2) Include all documents then available in the exercise of reasonable diligence, to be offered in support of the complaint, and to be served upon all persons named in the complaint as persons responsible for the alleged action(s) or omission(s) upon which the complaint is based;
- (3) Provide a concise but complete statement of the facts relied upon to substantiate each allegation; and
- (4) Except for complaints filed under 49 CFR 26.105(c), describe how the complainant was directly and substantially affected by the things done or omitted to be done by the respondents.
- (c) Unless the complaint is dismissed pursuant to §16.25 or §16.27, the FAA notifies the complainant and respondent in writing within 20 days after the date the FAA receives the complaint that the complaint has been docketed.
- (d) The respondent shall file an answer within 20 days of the date of service of the FAA notification or, if a motion is filed under §16.26, within 20 days of the date of service of an FAA order denying all or part of that motion.
- (e) The complainant may file a reply within 10 days of the date of service of the answer.
- (f) The respondent may file a rebuttal within 10 days of the date of service of the complainant's reply.
- (g) The answer, reply, and rebuttal shall, like the complaint, be accompanied by supporting documentation upon which the parties rely.
- (h) The answer shall deny or admit the allegations made in the complaint or state that the person filing the document is without sufficient knowledge or information to admit or deny an allegation, and shall assert any affirmative defense.
- (i) The answer, reply, and rebuttal shall each contain a concise but complete statement of the facts relied upon to substantiate the answers, admissions, denials, or averments made.
- (j) Amendments or supplements to the pleadings described in this section will not be allowed without showing good cause through a motion and supporting documents.
- (k) Burden of proof. Except as used in subpart F of this part,
- (1) The burden of proof is on the complainant to show noncompliance with

- an Act or any regulation, order, agreement or document of conveyance issued under the authority of an Act.
- (2) Except as otherwise provided by statute or rule, the proponent of a motion, request, or order has the burden of proof.
- (3) A party who has asserted an affirmative defense has the burden of proving the affirmative defense.
- (1) Except for good cause shown through motion and supporting documents, discovery is not permitted except as provided in §§ 16.213 and 16.215.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56143, Sept. 12, 2013]

§ 16.25 Dismissals.

- (a) Within 20 days after the receipt of the complaint, unless a motion has been filed under §16.26, the Director will dismiss a complaint, or any claim made in a complaint, with prejudice if:
- (1) It appears on its face to be outside the jurisdiction of the Administrator under the Acts listed in §16.1:
- (2) On its face it does not state a claim that warrants an investigation or further action by the FAA; or
- (3) The complainant lacks standing to file a complaint under §§16.3 and 16.23.
- (b) A dismissal under this section will include the reasons for the dismissal.

[Amdt. 16-1, 78 FR 56144, Sept. 12, 2013]

§ 16.26 Motions to dismiss and motions for summary judgment.

- (a) In lieu of an answer, the respondent may file a motion to dismiss the complaint or a motion for summary judgment on the complaint. The respondent may move for dismissal of the entire complaint or move for dismissal of particular issues from adjudication. The motion must be filed within 20 days after the date of service of the FAA notification of docketing.
- (b) Motions to dismiss. (1) A motion to dismiss shall be accompanied by a concise statement of the reasons for seeking dismissal. The respondent must show that the complaint should be dismissed, with prejudice, if:
- (i) It appears on its face to be outside the jurisdiction of the Administrator under the Acts listed in §16.1;

- (ii) On its face it does not state a claim that warrants an investigation or further action by the FAA; or
- (iii) The complainant lacks standing to file a complaint under §§16.3 and 16.23.
- (2) A motion to dismiss may seek dismissal of the entire complaint or the dismissal of specified claims in the complaint. A motion to dismiss shall be accompanied by a supporting memorandum of points and authorities.
- (3) A complainant may file an answer to the motion to dismiss within 10 days of the date the motion is served on the complainant, or within any other period set by the Director. The answer shall be accompanied by a concise statement of reasons for opposing dismissal, and may be accompanied by affidavits and other documentary evidence in support of that contention.
- (4) Within 30 days of the date an answer to a motion to dismiss is due under this section, the Director may issue an order disposing of the motion. If the Director denies the motion to dismiss in whole or in part, or grants the motion in part, then within 20 days of when the order is served on the respondent, the respondent shall file an answer to the complaint.
- (5) If the Director does not act on the motion to dismiss within 30 days of the date an answer to a motion is due under this section, the respondent shall file an answer to the complaint within the next 20 days.
- (c) Motions for summary judgment. (1) A motion for summary judgment may be based upon the ground that there is no genuine issue of material fact for adjudication and that the complaint, when viewed in the light most favorable to the complainant, should be summarily adjudicated in favor of the respondent as a matter of law. A motion for summary judgment may seek dismissal of the entire complaint or dismissal of specified claims or issues in the complaint.
- (2) The motion for summary judgment shall be accompanied by a concise statement of the material facts as to which the respondent contends there is no genuine issue of material fact. The motion may include affidavits and documentary evidence in support of the

- contention that there is no genuine issue of material fact in dispute.
- (3) A complainant may file an answer to the motion for summary judgment within 10 days of the date the motion is served on the complainant, or within any other period set by the Director. The answer shall be accompanied by a concise statement of the material facts as to which the complainant contends there is a genuine issue, and may be accompanied by affidavits and other documentary evidence in support of that contention.
- (4) Within 30 days of the date an answer to a motion for summary judgment is due under this section, the Director may issue an order disposing of the motion. If the Director denies the motion in whole or in part, or grants the motion in part, then within 20 days of when the order is served on the respondent, the respondent shall file an answer to the complaint.
- (5) If the Director does not act on the motion for summary judgment within 30 days of the date an answer to a motion is due under this section, the respondent shall file an answer to the complaint within the next 20 days.

[Amdt. 16-1, 78 FR 56144, Sept. 12, 2013]

§ 16.27 Incomplete complaints.

- (a) If a complaint is not dismissed pursuant to §16.25, but is deficient as to one or more of the requirements set forth in §16.21 or §16.23(b), the Director will dismiss the complaint within 20 days after receiving it. Dismissal will be without prejudice to the refiling of the complaint after amendment to correct the deficiencies. The Director's dismissal will include the reasons for the dismissal.
- (b) Dismissals under this section are not initial determinations, and appeals from decisions under this section will not be permitted.

[Amdt. 16-1, 78 FR 56144, Sept. 12, 2013]

§16.29 Investigations.

(a) If, based on the pleadings, there appears to be a reasonable basis for further investigation, the FAA investigates the subject matter of the complaint.

- (b) The investigation may include one or more of the following, at the sole discretion of the FAA:
- (1) A review of the written submissions or pleadings of the parties, as supplemented by any informal investigation the FAA considers necessary and by additional information furnished by the parties at FAA request. In rendering its initial determination, the FAA may rely entirely on the complaint and the responsive pleadings provided under this subpart. Each party shall file documents that it considers sufficient to present all relevant facts and argument necessary for the FAA to determine whether the sponsor is in compliance.
- (2) Obtaining additional oral and documentary evidence by use of the agency's authority to compel production of such evidence under 49 U.S.C. 40113 and 46104, and 49 U.S.C. 47122. The Administrator's statutory authority to issue compulsory process has been delegated to the Chief Counsel, the Deputy Chief Counsel, the Assistant Chief Counsel for Airports and Environmental Law, and each Assistant Chief Counsel for a region or center.
- (3) Conducting or requiring that a sponsor conduct an audit of airport financial records and transactions as provided in 49 U.S.C. 47107 and 47121.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1 78 FR 56145, Sept. 12, 2013]

§16.31 Director's Determinations after investigations.

- (a) After consideration of the pleadings and other information obtained by the FAA after investigation, the Director will render an initial determination and serve it upon each party within 120 days of the date the last pleading specified in §16.23 was due.
- (b)(1) The Director's Determination shall include findings of fact and conclusions of law, accompanied by explanations and based upon all material issues of fact, credibility of the evidence, law and discretion presented on the record, together with a statement of the reasons therefor.
- (2) The Director shall issue a determination or rule in a party's favor only if the determination or ruling is in accordance with law and supported by a

- preponderance of the reliable, probative, and substantial evidence contained in the record.
- (c) A party adversely affected by the Director's Determination may appeal the initial determination as provided in §16.33. However, if the Director's Determination that is appealed contains a Corrective Action Plan, the Director has the discretion to suspend the Corrective Action Plan until the appeal is resolved.
- (d) If the Director's Determination finds the respondent in noncompliance and proposes the issuance of a compliance order, the initial determination will include notice of opportunity for a hearing under subpart F of this part if a hearing is required by statute or otherwise provided by the FAA. A hearing may be required by statute if the FAA determination would terminate eligibility for grants under 49 U.S.C. 47114(c) or (e), or terminate payments on a grant agreement under 49 U.S.C. subchapter 471. The respondent may elect or waive a hearing, as provided in subpart E of this part.
- (e) The Director will not consider requests for rehearing, reargument, reconsideration, or modification of a Director's Determination without a finding of good cause.

[Amdt. 16–1, 78 FR 56145, Sept. 12, 2013]

§16.33 Final decisions without hearing.

- (a) The Associate Administrator may transfer to the FAA Assistant Administrator for Civil Rights the responsibility to prepare and issue Final Agency Decisions pursuant to this section for appeals with issues concerning civil rights.
- (b) The Associate Administrator will issue a final decision on appeal from the Director's Determination, without a hearing, where—
- (1) The complaint is dismissed after investigation;
- (2) A hearing is not required by statute and is not otherwise made available by the FAA; or
- (3) The FAA provides opportunity for a hearing to the respondent and the respondent waives the opportunity for a hearing as provided in subpart E of this part.

- (c) In the cases described in paragraph (b) of this section, within 30 days after the date of service of the initial determination, a party adversely affected by the Director's Determination may file in accordance with §16.13 and serve in accordance with §16.15 a simultaneous Notice of Appeal and Brief.
- (d) A reply to an appeal brief may be filed within 20 days after the date of service of the appeal.
- (e) On appeal, the Associate Administrator will consider the issues addressed in any order on a motion to dismiss or motion for summary judgment and any issues accepted in the Director's Determination using the following analysis:
- (1) Are the findings of fact each supported by a preponderance of reliable, probative, and substantial evidence contained in the record?
- (2) Are conclusions made in accordance with law, precedent and policy?
- (3) Are the questions on appeal substantial?
- (4) Have any prejudicial errors occurred?
- (f) Any new issues or evidence presented in an appeal or reply will not be considered unless accompanied by a petition and good cause found as to why the new issue or evidence was not presented to the Director. Such a petition must:
 - (1) Set forth the new matter;
- (2) Contain affidavits of prospective witnesses, authenticated documents, or both, or an explanation of why such substantiation is unavailable; and
- (3) Contain a statement explaining why such new issue or evidence could not have been discovered in the exercise of due diligence prior to the date on which the evidentiary record closed.
- (g) The Associate Administrator will issue a final decision and order within 60 days after the due date of the reply.
- (h) If no appeal is filed within the time period specified in paragraph (c) of this section, the Director's Determination becomes the final decision and order of the FAA without further action. A Director's Determination that becomes final, because there is no administrative appeal, is not judicially reviewable.
- (i) No requests for rehearing, reargument, reconsideration, or modification

of a final order will be considered without a finding of good cause.

[Amdt. 16-1, 78 FR 56145, Sept. 12, 2013]

§16.34 Consent orders.

- (a) The parties may agree at any time before the issuance of a final agency decision to dispose of the case by proposing a consent order. Good faith efforts to resolve a complaint through issuance of a consent order may continue throughout the administrative process. However, except as provided in §16.11(a), such efforts may not serve as the basis for extensions of the times set forth in this part.
- (b) A proposal for a consent order, specified in paragraph (a) of this section, shall include:
 - (1) A proposed consent order:
- (2) An admission of all jurisdictional facts; and
- (3) An express waiver of the right to further procedural steps and of all rights of judicial review.
- (c) If the parties agree to dispose of a case by issuance of a consent order before the FAA issues a Director's Determination, the proposal for a consent order is submitted jointly by the parties to the Director, together with a request to adopt the consent order and dismiss the case. The Director may issue the consent order as an order of the FAA and terminate the proceeding.

[Amdt. 16-1, 78 FR 56145, Sept. 12, 2013]

Subpart D—Special Rules Applicable to Proceedings Initiated by the FAA

§ 16.101 Basis for the initiation of agency action.

The FAA may initiate its own investigation of any matter within the applicability of this part without having received a complaint. The investigation may include, without limitation, any of the actions described in §16.29(b).

§ 16.103 Notice of investigation.

Following the initiation of an investigation under §16.101, the FAA sends a notice to the person(s) subject to investigation. The notice will set forth the areas of the agency's concern and the reasons therefor; request a response to

the notice within 30 days of the date of service; and inform the respondent that the FAA will, in its discretion, invite good faith efforts to resolve the matter.

§ 16.105 Failure to resolve informally.

If the matters addressed in the FAA notices are not resolved informally, the FAA may issue a Director's Determination under \$16.31.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56146, Sept. 12, 2013]

Subpart E—Proposed Orders of Compliance

§ 16.109 Orders terminating eligibility for grants, cease and desist orders, and other compliance orders.

- (a) The agency will provide the opportunity for a hearing if, in the Director's determination, the agency issues or proposes to issue an order terminating eligibility for grants pursuant to 49 U.S.C. 47106(d), an order suspending the payment of grant funds pursuant to 49 U.S.C. 47111(d); an order withholding approval of any new application to impose a passenger facility charge pursuant to 49 U.S.C. 47111(e); a cease and desist order; an order directing the refund of fees unlawfully collected; or any other compliance order issued by the Administrator to carry out the provisions of the Acts, and required to be issued after notice and opportunity for a hearing. In cases in which a hearing is not required by statute, the FAA may provide opportunity for a hearing at its discretion.
- (b) In a case in which the agency provides the opportunity for a hearing, the Director's Determination issued under §16.31 will include a statement of the availability of a hearing under subpart F of this part.
- (1) Within 20 days after service of a Director's Determination under §16.31 that provides an opportunity for a hearing a person subject to the proposed compliance order may—
- (i) Request a hearing under subpart F of this part;
- (ii) Waive hearing and appeal the Director's Determination in writing, as provided in §16.33;

- (iii) File, jointly with a complainant, a motion to withdraw the complaint and to dismiss the proposed compliance action; or
- (iv) Submit, jointly with the agency, a proposed consent order under §16.34(c).
- (2) If the respondent fails to file an appeal in writing within the time periods provided in paragraph (c) of this section, the Director's Determination becomes final.
- (c) The Director may either direct the respondent to submit a Corrective Action Plan or initiate proceedings to revoke and/or deny the respondent's application for Airport Improvement Program discretionary grants under 49 U.S.C. 47115 and general aviation airport grants under 49 U.S.C. 47114(d) when a Director's Determination finds a respondent in noncompliance and does not provide for a hearing.
- (d) In the event that the respondent fails to submit, in accordance with a Director's Determination, a Corrective Action Plan acceptable to the FAA within the time provided, unless extended by the FAA for good cause, and/or if the respondent fails to complete the Corrective Action Plan as specified therein, the Director may initiate action to revoke and/or deny applications for Airport Improvement Program discretionary grants under 49 U.S.C. 47115 and general aviation airport grants under 49 U.S.C. 47114(d).
- (e) For those violations that cannot be remedied through corrective action, the Director may initiate action to revoke and/or deny the respondent's applications for Airport Improvement Program discretionary grants under 49 U.S.C. 47115 and general aviation airport grants under 49 U.S.C. 47114(d).
- (f) When the Director concludes that the respondent has fully complied with the Corrective Action Plan and/or when the Director determines that the respondent has corrected the areas of noncompliance, the Director will terminate the proceeding.
- (g) A complainant's standing terminates upon the issuance of a Director's Determination that finds a respondent in noncompliance on all identified issues. The complainant may not appeal the Director's Determination if

the Director finds noncompliance on all identified issues.

[Amdt. 16-1, 78 FR 56146, Sept. 12, 2013]

Subpart F—Hearings

§ 16.201 Notice and order of hearing.

- (a) If a respondent is provided the opportunity for hearing in an initial determination and does not waive hearing, the Deputy Chief Counsel within 10 days after the respondent elects a hearing will issue and serve on the respondent and complainant a hearing order. The hearing order will set forth:
- (1) The allegations in the complaint, or notice of investigation, and the chronology and results of the investigation preliminary to the hearing;
- (2) The relevant statutory, judicial, regulatory, and other authorities;
 - (3) The issues to be decided;
- (4) Such rules of procedure as may be necessary to supplement the provisions of this part;
- (5) The name and address of the person designated as hearing officer, and the assignment of authority to the hearing officer to conduct the hearing in accordance with the procedures set forth in this part; and
- (6) The date by which the hearing officer is directed to issue an initial decision.
- (b) Where there are no genuine issues of material fact requiring oral examination of witnesses, the hearing order may contain a direction to the hearing officer to conduct a hearing by submission of briefs and oral argument without the presentation of testimony or other evidence.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56146, Sept. 12, 2013]

§ 16.202 Powers of a hearing officer.

In accordance with the rules of this subpart, a hearing officer may:

- (a) Give notice of, and hold, prehearing conferences and hearings;
- (b) Administer oaths and affirmations:
- (c) Issue subpoenas authorized by law and issue notices of deposition requested by the parties;
- (d) Limit the frequency and extent of discovery;

- (e) Rule on offers of proof;
- (f) Receive relevant and material evidence:
- (g) Regulate the course of the hearing in accordance with the rules of this part to avoid unnecessary and duplicative proceedings in the interest of prompt and fair resolution of the matters at issue;
- (h) Hold conferences to settle or to simplify the issues by consent of the parties:
- (i) Dispose of procedural motions and requests;
 - (j) Examine witnesses; and
- (k) Make findings of fact and conclusions of law, and issue an initial decision.

§ 16.203 Appearances, parties, and rights of parties.

- (a) Appearances. Any party may appear and be heard in person.
- (1) Any party may be accompanied, represented, or advised by an attorney licensed by a State, the District of Columbia, or a territory of the United States to practice law or appear before the courts of that State or territory, or by another person authorized by the hearing officer to be the party's representative.
- (2) An attorney, or other duly authorized representative, who represents a party shall file a notice of appearance in accordance with §16.13.
- (b) Parties and agency participation. (1) The parties to the hearing are the complainant(s) and respondent(s) named in the hearing order, and the agency. The style of any pleadings filed under this Subpart shall name the respondent as the Appellant, and the Federal Aviation Administration as the Agency.
- (2) Unless otherwise specified in the hearing order, the agency attorney will serve as prosecutor for the agency from the date of issuance of the Director's Determination providing an opportunity for hearing.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56146, Sept. 12, 2013]

§ 16.207 Intervention and other participation.

(a) Intervention and participation by other persons are permitted only at the hearing stage of the complaint process and with the written approval of the hearing officer.

- (b) A person may submit a written motion for leave to intervene as a party. Except for good cause shown, a motion for leave to intervene shall be submitted not later than 10 days after the notice of hearing and hearing order.
- (c) If the hearing officer finds that intervention will not unduly broaden the issues or delay the proceedings and, if the person has an interest that will benefit the proceedings, the hearing officer may grant a motion for leave to intervene. The hearing officer may determine the extent to which an intervenor may participate in the proceedings.
- (d) Other persons may petition the hearing officer for leave to participate in the hearing. Participation is limited to the filing of a posthearing brief and reply to the hearing officer and the Associate Administrator. Such a brief shall be filed and served on all parties in the same manner as the parties' posthearing briefs are filed.
- (e) Participation under this section is at the discretion of the hearing officer, and no decision permitting participation shall be deemed to constitute an expression that the participant has such a substantial interest in the proceeding as would entitle it to judicial review of such decision.

[Amdt. 16-1, 78 FR 56146, Sept. 12, 2013]

§16.209 Extension of time.

- (a) Extension by oral agreement. The parties may agree to extend for a reasonable period of time for filing a document under this part. If the parties agree, the hearing officer shall grant one extension of time to each party. The party seeking the extension of time shall submit a draft order to the hearing officer to be signed by the hearing officer and filed with the hearing docket. The hearing officer may grant additional oral requests for an extension of time where the parties agree to the extension.
- (b) Extension by motion. A party shall file a written motion for an extension of time with the hearing officer not later than 7 days before the document is due unless good cause for the late filing is shown. A party filing a written

motion for an extension of time shall serve a copy of the motion on each party.

- (c) Failure to rule. If the hearing officer fails to rule on a written motion for an extension of time by the date the document was due, the motion for an extension of time is deemed denied.
- (d) Effect on time limits. In a hearing required by section 519(b) of the Airport and Airways Improvement Act, as amended in 1987, 49 U.S.C. 47106(e) and 47111(d), the due date for the hearing officer's initial decision and for the final agency decision are extended by the length of the extension granted by the hearing officer only if the hearing officer grants an extension of time as a result of an agreement by the parties as specified in paragraph (a) of this section or, if the hearing officer grants an extension of time as a result of the sponsor's failure to adhere to the hearing schedule. In any other hearing, an extension of time granted by the hearing officer for any reason extends the due date for the hearing officer's initial decision and for the final agency decision by the length of time of the hearing officer's decision.

§16.211 Prehearing conference.

- (a) Prehearing conference notice. The hearing officer schedules a prehearing conference and serves a prehearing conference notice on the parties promptly after being designated as a hearing officer.
- (1) The prehearing conference notice specifies the date, time, place, and manner (in person or by telephone) of the prehearing conference.
- (2) The prehearing conference notice may direct the parties to exchange proposed witness lists, requests for evidence and the production of documents in the possession of another party, responses to interrogatories, admissions, proposed procedural schedules, and proposed stipulations before the date of the prehearing conference.
- (b) The prehearing conference. The prehearing conference is conducted by telephone or in person, at the hearing officer's discretion. The prehearing conference addresses matters raised in the prehearing conference notice and

such other matters as the hearing officer determines will assist in a prompt, full and fair hearing of the issues.

(c) Prehearing conference report. At the close of the prehearing conference, the hearing officer rules on any requests for evidence and the production of documents in the possession of other parties, responses to interrogatories, and admissions; on any requests for depositions; on any proposed stipulations; and on any pending applications for subpoenas as permitted by §16.219. In addition, the hearing officer establishes the schedule, which shall provide for the issuance of an initial decision not later than 110 days after issuance of the Director's Determination order unless otherwise provided in the hearing order.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56147, Sept. 12, 2013]

§16.213 Discovery.

- (a) Discovery is limited to requests for admissions, requests for production of documents, interrogatories, and depositions as authorized by §16.215.
- (b) The hearing officer shall limit the frequency and extent of discovery permitted by this section if a party shows that—
- (1) The information requested is cumulative or repetitious;
- (2) The information requested may be obtained from another less burdensome and more convenient source;
- (3) The party requesting the information has had ample opportunity to obtain the information through other discovery methods permitted under this section; or
- (4) The method or scope of discovery requested by the party is unduly burdensome or expensive.

§ 16.215 Depositions.

- (a) General. For good cause shown, the hearing officer may order that the testimony of a witness may be taken by deposition and that the witness produce documentary evidence in connection with such testimony. Generally, an order to take the deposition of a witness is entered only if:
- (1) The person whose deposition is to be taken would be unavailable at the hearing:

- (2) The deposition is deemed necessary to perpetuate the testimony of the witness; or
- (3) The taking of the deposition is necessary to prevent undue and excessive expense to a party and will not result in undue burden to other parties or in undue delay.
- (b) Application for deposition. Any party desiring to take the deposition of a witness shall make application therefor to the hearing officer in writing, with a copy of the application served on each party. The application shall include:
- (1) The name and residence of the witness;
- (2) The time and place for the taking of the proposed deposition;
- (3) The reasons why such deposition should be taken; and
- (4) A general description of the matters concerning which the witness will be asked to testify.
- (c) Order authorizing deposition. If good cause is shown, the hearing officer, in his or her discretion, issues an order authorizing the deposition and specifying the name of the witness to be deposed, the location and time of the deposition and the general scope and subject matter of the testimony to be taken.
- (d) Procedures for deposition. (1) Witnesses whose testimony is taken by deposition shall be sworn or shall affirm before any questions are put to them. Each question propounded shall be recorded and the answers of the witness transcribed verbatim.
- (2) Objections to questions or evidence shall be recorded in the transcript of the deposition. The interposing of an objection shall not relieve the witness of the obligation to answer questions, except where the answer would violate a privilege.
- (3) The written transcript shall be subscribed by the witness, unless the parties by stipulation waive the signing, or the witness is ill, cannot be found, or refuses to sign. The reporter shall note the reason for failure to sign.
- (e) Depositions of agency employees. (1) Depositions of Agency Employees will not be allowed except under the provisions of 49 CFR part 9.

(2) Such depositions will be allowed only with the specific written permission of the Chief Counsel or his or her designee.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56147, Sept. 12, 2013]

§16.217 Witnesses.

- (a) Each party may designate as a witness any person who is able and willing to give testimony that is relevant and material to the issues in the hearing case, subject to the limitation set forth in paragraph (b) of this section.
- (b) The hearing officer may exclude testimony of witnesses that would be irrelevant, immaterial, or unduly repetitious.
- (c) Any witness may be accompanied by counsel. Counsel representing a nonparty witness has no right to examine the witness or otherwise participate in the development of testimony.

§ 16.219 Subpoenas.

- (a) Request for subpoena. A party may apply to the hearing officer, within the time specified for such applications in the prehearing conference report, for a subpoena to compel testimony at a hearing or to require the production of documents only from the following persons:
 - (1) Another party;
- (2) An officer, employee, or agent of another party;
- (3) Any other person named in the complaint as participating in or benefiting from the actions of the respondent alleged to have violated any Act;
- (4) An officer, employee, or agent of any other person named in the complaint as participating in or benefiting from the actions of the respondent alleged to have violated any Act.
- (b) Issuance and service of subpoena. (1) The hearing officer issues the subpoena if the hearing officer determines that the evidence to be obtained by the subpoena is relevant and material to the resolution of the issues in the case.
- (2) Subpoenas shall be served by personal service, or upon an agent designated in writing for the purpose, or by certified mail, return receipt addressed to such person or agent. Whenever service is made by registered or

- certified mail, the date of mailing shall be considered as the time when service is made.
- (3) A subpoena issued under this part is effective throughout the United States or any territory or possession thereof.
- (c) Motions to quash or modify subpoena. (1) A party or any person upon whom a subpoena has been served may file a motion to quash or modify the subpoena with the hearing officer at or before the time specified in the subpoena for the filing of such motions. The applicant shall describe in detail the basis for the application to quash or modify the subpoena including, but not limited to, a statement that the testimony, document, or tangible evidence is not relevant to the proceeding, that the subpoena is not reasonably tailored to the scope of the proceeding, or that the subpoena is unreasonable and oppressive.
- (2) A motion to quash or modify the subpoena stays the effect of the subpoena pending a decision by the hearing officer on the motion.

§16.221 Witness fees.

- (a) The party on whose behalf a witness appears is responsible for paying any witness fees and mileage expenses.
- (b) Except for employees of the United States summoned to testify as to matters related to their public employment, witnesses summoned by subpoena shall be paid the same fees and mileage expenses as are paid to a witness in a court of the United States in comparable circumstances.

§ 16.223 Evidence.

- (a) General. A party may submit direct and rebuttal evidence in accordance with this section.
- (b) Requirement for written testimony and evidence. Except in the case of evidence obtained by subpoena, or in the case of a special ruling by the hearing officer to admit oral testimony, a party's direct and rebuttal evidence shall be submitted in written form in advance of the oral hearing pursuant to the schedule established in the hearing officer's prehearing conference report. Written direct and rebuttal fact testimony shall be certified by the witness as true and correct. Subject to the

same exception (for evidence obtained by subpoena or subject to a special ruling by the hearing officer), oral examination of a party's own witness is limited to certification of the accuracy of written evidence, including correction and updating, if necessary, and reexamination following cross-examination by other parties.

- (c) Subpoenaed testimony. Testimony of witnesses appearing under subpoena may be obtained orally.
- (d) Cross-examination. A party may conduct cross-examination that may be required for disclosure of the facts, subject to control by the hearing officer for fairness, expedition and exclusion of extraneous matters.
- (e) Hearsay evidence. Hearsay evidence is admissible in proceedings governed by this part. The fact that evidence is hearsay goes to the weight of evidence and does not affect its admissibility.
- (f) Admission of evidence. The hearing officer admits evidence introduced by a party in support of its case in accordance with this section, but may exclude irrelevant, immaterial, or unduly repetitious evidence.
- (g) Expert or opinion witnesses. An employee of the FAA or DOT may not be called as an expert or opinion witness for any party other than the agency except as provided in Department of Transportation regulations at 49 CFR part 9.

§ 16.225 Public disclosure of evidence.

- (a) Except as provided in this section, the hearing shall be open to the public.
- (b) The hearing officer may order that any information contained in the record be withheld from public disclosure. Any person may object to disclosure of information in the record by filing a written motion to withhold specific information with the hearing officer. The person shall state specific grounds for nondisclosure in the motion.
- (c) The hearing officer shall grant the motion to withhold information from public disclosure if the hearing officer determines that disclosure would be in violation of the Privacy Act, would reveal trade secrets or privileged or confidential commercial or financial infor-

mation, or is otherwise prohibited by law.

§ 16.227 Standard of proof.

The hearing officer shall issue an initial decision or rule in a party's favor only if the decision or ruling is in accordance with law and supported by a preponderance of the reliable, probative, and substantial evidence contained in the record.

[Amdt. 16–1, as amended at 78 FR 56147, Sept. 12. 2013]

§16.229 Burden of proof.

As used in this subpart, the burden of proof is as follows:

- (a) The burden of proof of noncompliance with an Act or any regulation, order, agreement or document of conveyance issued under the authority of an Act is on the agency.
- (b) Except as otherwise provided by statute or rule, the proponent of a motion, request, or order has the burden of proof.
- (c) A party who has asserted an affirmative defense has the burden of proving the affirmative defense.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56147, Sept. 12, 2013]

§ 16.231 Offer of proof.

A party whose evidence has been excluded by a ruling of the hearing officer may offer the evidence on the record when filing an appeal.

§16.233 Record.

- (a) Exclusive record. The transcript of all testimony in the hearing, all exhibits received into evidence, all motions, applications requests and rulings, all documents included in the hearing record and the Director's Determination shall constitute the exclusive record for decision in the proceedings and the basis for the issuance of any orders.
- (b) Examination and copy of record. A copy of the record will be filed by the FAA Part 16 Docket Clerk in the Federal Docket Management System (FDMS). Any person desiring to review the record may then do so at http://www.regulations.gov.

 $[{\rm Amdt.\ 16\text{--}1,\ 78\ FR\ 56147,\ Sept.\ 12,\ 2013}]$

§16.235 Argument before the hearing officer.

(a) Argument during the hearing. During the hearing, the hearing officer shall give the parties reasonable opportunity to present oral argument on the record supporting or opposing motions, objections, and rulings if the parties request an opportunity for argument. The hearing officer may direct written argument during the hearing if the hearing officer finds that submission of written arguments would not delay the hearing.

(b) Posthearing briefs. The hearing officer may request or permit the parties to submit posthearing briefs. The hearing officer may provide for the filing of simultaneous reply briefs as well, if such filing will not unduly delay the issuance of the hearing officer's initial decision. Posthearing briefs shall include proposed findings of fact and conclusions of law; exceptions to rulings of the hearing officer; references to the record in support of the findings of fact; and supporting arguments for the proposed findings, proposed conclusions, and exceptions.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56147, Sept. 12, 2013]

§16.237 Waiver of procedures.

- (a) The hearing officer shall waive such procedural steps as all parties to the hearing agree to waive before issuance of an initial decision.
- (b) Consent to a waiver of any procedural step bars the raising of this issue on appeal.
- (c) The parties may not by consent waive the obligation of the hearing officer to enter an initial decision on the record

§ 16.241 Initial decisions, order, and appeals.

- (a) The hearing officer shall issue an initial decision based on the record developed during the proceeding and shall send the initial decision to the parties not later than 110 days after the Director's Determination unless otherwise provided in the hearing order.
- (b) Each party adversely affected by the hearing officer's initial decision may file an appeal with the Associate Administrator within 15 days of the

date the initial decision is issued. Each party may file a reply to an appeal within 10 days after it is served on the party. Filing and service of appeals and replies shall be by personal delivery.

- (c) If an appeal is filed, the Associate Administrator reviews the record and issues a final agency decision and order within 60 days of the due date of the reply. If no appeal is filed, the Associate Administrator may take review of the case on his or her own motion. If the Associate Administrator finds that the respondent is not in compliance with any Act or any regulation, agreement, or document of conveyance issued or made under such Act, the final agency order includes, in accordance with §16.245(d), a statement of corrective action, if appropriate, and identifies sanctions for continued noncompliance.
- (d) If no appeal is filed, and the Associate Administrator does not take review of the initial decision on the Associate Administrator's own motion, the initial decision shall take effect as the final agency decision and order on the sixteenth day after the actual date the initial decision is issued.
- (e) The failure to file an appeal is deemed a waiver of any rights to seek judicial review of an initial decision that becomes a final agency decision by operation of paragraph (d) of this section.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56147, Sept. 12, 2013]

§ 16.243 Consent orders.

- (a) The agency attorney and the respondents may agree at any time before the issuance of a final decision and order to dispose of the case by issuance of a consent order. Good faith efforts to resolve a complaint through issuance of a consent order may continue throughout the administrative process. Except as provided in §16.209, such efforts may not serve as the basis for extensions of the times set forth in this part.
- (b) A proposal for a consent order, specified in paragraph (a) of this section, shall include:
 - (1) A proposed consent order;
- (2) An admission of all jurisdictional facts:

- (3) An express waiver of the right to further procedural steps and of all rights of judicial review; and
- (4) The hearing order, if issued, and an acknowledgment that the hearing order may be used to construe the terms of the consent order.
- (c) If the issuance of a consent order has been agreed upon by all parties to the hearing, the proposed consent order shall be filed with the hearing officer, along with a draft order adopting the consent decree and dismissing the case, for the hearing officer's adoption.
- (d) The deadline for the hearing officer's initial decision and the final agency decision is extended by the amount of days elapsed between the filing of the proposed consent order with the hearing officer and the issuance of the hearing officer's order continuing the hearing.
- (e) If the agency attorney and sponsor agree to dispose of a case by issuance of a consent order before the FAA issues a hearing order, the proposal for a consent order is submitted jointly to the official authorized to issue a hearing order, together with a request to adopt the consent order and dismiss the case. The official authorized to issue the hearing order issues the consent order as an order of the FAA and terminates the proceeding.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996, as amended at Amdt. 16–1, 78 FR 56147, Sept. 12, 2013]

§16.245 Associate Administrator review after a hearing.

- (a) The Associate Administrator may transfer to the FAA Assistant Administrator for Civil Rights the authority to prepare and issue Final Agency Decisions pursuant to §16.241 for appeals from a hearing concerning civil rights issues.
- (b) After a hearing is held, and, after considering the issues as set forth in §16.245(e), if the Associate Administrator determines that the hearing officer's initial decision or order should be changed, the Associate Administrator may:
- (1) Make any necessary findings and issue an order in lieu of the hearing officer's initial decision or order, or

- (2) Remand the proceeding for any such purpose as the Associate Administrator may deem necessary.
- (c) If the Associate Administrator takes review of the hearing officer's initial decision on the Associate Administrator's own motion, the Associate Administrator will issue a notice of review within 20 days of the actual date the initial decision is issued.
- (1) The notice sets forth the specific findings of fact and conclusions of law in the initial decision that are subject to review by the Associate Administrator.
- (2) Parties may file one brief on review to the Associate Administrator or rely on their posthearing brief to the hearing officer. A brief on review shall be filed not later than 10 days after service of the notice of review. Filing and service of a brief on review shall be by personal delivery.
- (3) The Associate Administrator issues a final agency decision and order within 30 days of the due date of the brief. If the Associate Administrator finds that the respondent is not in compliance with any Act or any regulation, agreement or document of conveyance issued under such Act, the final agency order includes a statement of corrective action, if appropriate.
- (d) When the final agency decision finds a respondent in noncompliance, and where a respondent fails to properly seek judicial review of the final agency decision as set forth in subpart G of this part, the Associate Administrator will issue an order remanding the case to the Director for the following action:
- (1) In the event that the respondent fails to submit, in accordance with the final agency decision, a Corrective Action Plan acceptable to the FAA within the time provided, unless extended by the FAA for good cause, and/or if the respondent fails to complete the Corrective Action Plan as specified therein, the Director may initiate action to revoke and/or deny applications for Airport Improvement Program grants issued under 49 U.S.C. 47114(c)-(e) and 47115. When the Director concludes that the respondent has fully complied with the Corrective Action Plan, the Director will issue an Order terminating the proceeding.

Federal Aviation Administration, DOT

- (2) For those violations that cannot be remedied through corrective action, the Director may initiate action to revoke and/or deny the respondent's applications for Airport Improvement Program grants issued under 49 U.S.C. 47114(c)–(e) and 47115.
- (e) On appeal from a hearing officer's initial decision, the Associate Administrator will consider the following questions:
- (1) Are the findings of fact each supported by a preponderance of reliable, probative and substantial evidence?
- (2) Are conclusions made in accordance with law, precedent and policy?
- (3) Are the questions on appeal substantial?
- (4) Have any prejudicial errors occurred?
- (f) Any new issues or evidence presented in an appeal or reply will not be allowed unless accompanied by a certified petition and good cause found as to why the new matter was not presented to the Director. Such a petition must:
- (1) Set forth the new issues or evidence:
- (2) Contain affidavits of prospective witnesses, authenticated documents, or both, or an explanation of why such substantiation is unavailable; and
- (3) Contain a statement explaining why such new matter could not have been discovered in the exercise of due diligence prior to the date on which the evidentiary record closed.
- (g) A Final Agency Decision may be appealed in accordance with subpart G of this part.

[Amdt. 16-1, 78 FR 56147, Sept. 12, 2013]

Subpart G—Judicial Review

Source: Docket No. 27783, 61 FR 54004, Oct. 16, 1996, unless otherwise noted. Redesignated by Amdt. 16-1, 78 FR 56148, Sept. 12, 2013

§16.247 Judicial review of a final decision and order.

(a) A person may seek judicial review, in a United States Court of Appeals, of a final decision and order of the Associate Administrator, and of an order of dismissal with prejudice issued by the Director, as provided in 49 U.S.C. 46110 or 49 U.S.C. 47106(d) and

- 47111(d). A party seeking judicial review shall file a petition for review with the Court not later than 60 days after the order has been served on the party or within 60 days after the entry of an order under 49 U.S.C. 46110.
- (b) The following do not constitute final decisions and orders subject to judicial review:
- (1) An FAA decision to dismiss a complaint without prejudice, as set forth in §16.27;
 - (2) A Director's Determination;
- (3) An initial decision issued by a hearing officer at the conclusion of a hearing:
- (4) A Director's Determination or an initial decision of a hearing officer becomes the final decision of the Associate Administrator because it was not appealed within the applicable time periods provided under §\$16.33(c) and 16.241(b).

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996. Redesignated and amended by Amdt. 16–1, 78 FR 56148, Sept. 12, 2013]

Subpart H—Ex Parte Communications

SOURCE: Docket No. 27783, 61 FR 54004, Oct. 16, 1996, unless otherwise noted. Redesignated at 78 FR 56148. Sept. 12, 2013.

§16.301 Prohibited ex parte communications.

- (a) The prohibitions of this section shall apply from the time a proceeding is noticed for hearing unless the person responsible for the communication has knowledge that it will be noticed, in which case the prohibitions shall apply at the time of the acquisition of such knowledge.
- (b) Except to the extent required for the disposition of ex parte matters as authorized by law:
- (1) No interested person outside the FAA and no FAA employee participating as a party shall make or knowingly cause to be made to any decisional employee an exparte communication relevant to the merits of the proceeding;
- (2) No FAA employee shall make or knowingly cause to be made to any interested person outside the FAA an ex parte communication relevant to the merits of the proceeding; or

(3) Ex parte communications regarding solely matters of agency procedure or practice are not prohibited by this section.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996. Redesignated at Amdt. 16–1, 78 FR 56148, Sept. 12, 2013]

§ 16.303 Procedures for handling exparte communications.

A decisional employee who receives or who makes or knowingly causes to be made a communication prohibited by §16.303 shall place in the public record of the proceeding:

- (a) All such written communications;
- (b) Memoranda stating the substance of all such oral communications; and
- (c) All written responses, and memoranda stating the substance of all oral responses, to the materials described in paragraphs (a) and (b) of this section.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996. Redesignated at Amdt. 16–1, 78 FR 56148, Sept. 12, 2013]

§ 16.305 Requirement to show cause and imposition of sanction.

- (a) Upon receipt of a communication knowingly made or knowingly caused to be made by a party in violation of §16.303, the Associate Administrator or his designee or the hearing officer may, to the extent consistent with the interests of justice and the policy of the underlying statutes, require the party to show cause why his or her claim or interest in the proceeding should not be dismissed, denied, disregarded, or otherwise adversely affected on account of such violation.
- (b) The Associate Administrator may, to the extent consistent with the interests of justice and the policy of the underlying statutes administered by the FAA, consider a violation of this subpart sufficient grounds for a decision adverse to a party who has knowingly committed such violation or knowingly caused such violation to occur.

[Doc. No. 27783, 61 FR 54004, Oct. 16, 1996. Redesignated at Amdt. 16–1, 78 FR 56148, Sept. 12, 2013]

PART 17—PROCEDURES FOR PRO-TESTS AND CONTRACT DISPUTES

Subpart A—General

Sec.

- 17.1 Applicability.
- 17.3 Definitions.
- 17.5 Delegation of authority.
- 17.7 Filing and computation of time.
- 17.9 Protective orders.

Subpart B—Protests

- 17.11 Matters not subject to protest.
- 17.13 Dispute resolution process for protests.
- 17.15 Filing a protest.
- 17.17 Initial protest procedures.
 - 17.19 Motions practice and dismissal or summary decision of protests.
 - 17.21 Adjudicative Process for protests.
 - 17.23 Protest remedies.

Subpart C—Contract Disputes

- 17.25 Dispute resolution process for contract disputes.
- 17.27 Filing a contract dispute.
- 17.29 Informal resolution period.
- 17.31 Dismissal or summary decision of contract disputes.
- 17.33 Adjudicative Process for contract disputes.

Subpart D—Alternative Dispute Resolution

- 17.35 Use of alternative dispute resolution.
- 17.37 Election of alternative dispute resolution process.
- 17.39 Confidentiality of ADR.

Subpart E—Finality and Review

- 17.41 Final orders.
- 17.43 Judicial review.
- 17.45 Conforming amendments.
- 17.47 Reconsideration.

Subpart F—Other Matters

- 17.49 Sanctions.
- 17.51 Decorum and professional conduct.
- $17.53\,$ Orders and subpoenas for testimony and document production.
- 17.55 Standing orders of the ODRA director.

Subpart G-Pre-Disputes

- 17.57 Dispute resolution process for Pre-disputes.
- 17.59 Filing a Pre-dispute.
- 17.61 Use of alternative dispute resolution.
- APPENDIX A TO PART 17—ALTERNATIVE DISPUTE RESOLUTION (ADR)

AUTHORITY: 5 U.S.C. 570-581, 49 U.S.C. 106(f)(2), 40110, 40111, 40112, 46102, 46014, 46105, 46109, and 46110.

SOURCE: 76 FR 55221, Sept. 7, 2011, unless otherwise noted.

Subpart A—General

§ 17.1 Applicability.

This part applies to all Acquisition Management System (AMS) bid protests and contract disputes involving the FAA that are filed at the Office of Dispute Resolution for Acquisition (ODRA) on or after October 7, 2011, with the exception of those contract disputes arising under or related to FAA contracts entered into prior to April 1, 1996, where such contracts have not been modified to be made subject to the FAA AMS. This part also applies to pre-disputes as described in subpart G of this part.

§17.3 Definitions.

- (a) Accrual means to come into existence as a legally enforceable claim.
- (b) Accrual of a contract claim means that all events relating to a claim have occurred, which fix liability of either the government or the contractor and permit assertion of the claim, regardless of when the claimant actually discovered those events. For liability to be fixed, some injury must have occurred. Monetary damages need not have been incurred, but if the claim is for money, such damages must be capable of reasonable estimation. The accrual of a claim or the running of the limitations period may be tolled on equitable grounds, including but not limited to active concealment, fraud, or if the facts were inherently unknowable.
- (c) Acquisition Management System (AMS) establishes the policies, guiding principles, and internal procedures for the FAA's acquisition system.
- (d) Adjudicative Process is an administrative adjudicatory process used to decide protests and contract disputes where the parties have not achieved resolution through informal communication or the use of ADR. The Adjudicative Process is conducted by a Dispute Resolution Officer (DRO) or Special Master selected by the ODRA Director to preside over the case in accordance with Public Law 108–176, Sec-

- tion 224, Codified at 49 U.S.C. 40110(d)(4).
- (e) Administrator means the Administrator of the Federal Aviation Administration.
- (f) Alternative Dispute Resolution (ADR) is the primary means of voluntary dispute resolution that is employed by the ODRA. See Appendix A of this part.
- (g) Compensated Neutral refers to an impartial third party chosen by the parties to act as a facilitator, mediator, or arbitrator functioning to resolve the protest or contract dispute under the auspices of the ODRA. The parties pay equally for the services of a compensated neutral, unless otherwise agreed to by the parties. An ODRA DRO or neutral cannot be a compensated neutral.
- (h) Contract Dispute, as used in this part, means a written request to the ODRA seeking, as a matter of right under an FAA contract subject to the AMS, the payment of money in a sum certain, the adjustment or interpretation of contract terms, or for other relief arising under, relating to, or involving an alleged breach of that contract. A contract dispute does not require, as a prerequisite, the issuance of a Contracting Officer final decision. Contract disputes, for purposes of ADR only, may also involve contracts not subject to the AMS.
- (i) Counsel refers to a Legal Representative who is an attorney licensed by a State, the District of Columbia, or a territory of the United States to practice law or appear before the courts of that State or territory.
- (j) Contractor is a party in contractual privity with the FAA and responsible for performance of a contract's requirements.
- (k) Discovery is the procedure whereby opposing parties in a protest or contract dispute may, either voluntarily or to the extent ordered by the ODRA, obtain testimony from, or documents and information held by, other parties or non-parties.
- (1) Dispute Resolution Officer (DRO) is an attorney and member of the ODRA staff. The term DRO can include the Director of the ODRA.

- (m) Interested party, in the context of a bid protest, is one whose direct economic interest has been or would be affected by the award or failure to award an FAA contract. Proposed subcontractors are not "interested parties" within this definition and are not eligible to submit protests to the ODRA. Subcontractors not in privity with the FAA are not interested parties in the context of a contract dispute.
- (n) Intervenor is an interested party other than the protester whose participation in a protest is allowed by the ODRA. For a post-award protest, the awardee of the contract that is the subject of the protest will be allowed, upon timely request, to participate as an intervenor in the protest. In such a protest, no other interested parties will be allowed to participate as intervenors.
- (o) Legal Representative is an individual(s) designated to act on behalf of a party in matters before the ODRA. Unless otherwise provided under §§17.15(c)(2), 17.27(a)(1), or 17.59(a)(6), a Notice of Appearance must be filed with the ODRA containing the name, address, telephone and facsimile (Fax) numbers of a party's legal representative.
- (p) Neutral refers to an impartial third party in the ADR process chosen by the parties to act as a facilitator, mediator, arbitrator, or otherwise to aid the parties in resolving a protest or contract dispute. A neutral can be a DRO or a person not an employee of the ODRA.
- (q) *ODRA* is the FAA's exclusive forum acting on behalf of the Administrator, pursuant to the statutory authority granted by Public Law 108–176, Section 224, to provide dispute resolution services and to adjudicate matters within its jurisdiction. The ODRA may also provide non-binding dispute resolution services in matters outside of its jurisdiction where mutually requested to do so by the parties involved.
- (r) Parties include the protester(s) or the contractor, the FAA, and any intervenor(s).
- (s) Pre-Disputes mean an issue(s) in controversy concerning an FAA contract or solicitation that, by mutual agreement of the parties, is filed with the ODRA. See subpart G of this part.

- (t) Product Team, as used in these rules, refers to the FAA organization(s) responsible for the procurement or contracting activity, without regard to funding source, and includes the Contracting Officer (CO). The Product Team, acting through assigned FAA counsel, is responsible for all communications with and submissions to the ODRA in pending matters.
- (u) Screening Information Request (SIR or Solicitation) means a request by the FAA for documentation, information, presentations, proposals, or binding offers concerning an approach to meeting potential acquisition requirements established by the FAA.
- (v) A Special Master is a non-FAA attorney or judge who has been assigned by the ODRA to act as its finder of fact, and to make findings and recommendations based upon AMS policy and applicable law and authorities in the Adjudicative Process.

§ 17.5 Delegation of authority.

- (a) The authority of the Administrator to conduct dispute resolution and adjudicative proceedings concerning acquisition matters is delegated to the Director of the ODRA.
- (b) The Director of the ODRA may redelegate to Special Masters and DROs such delegated authority in paragraph (a) of this section as deemed necessary by the Director for efficient resolution of an assigned protest or contract dispute, including the imposition of sanctions for the filing of frivolous pleadings, making false statements, or other disciplinary actions. See subpart F of this part.

§17.7 Filing and computation of time.

(a) Filing of a protest or contract dispute may be accomplished by overnight delivery, by hand delivery, by Fax, or, if permitted by Order of the ODRA, by electronic filing. A protest or contract dispute is considered to be filed on the date it is received by the ODRA during normal business hours. The ODRA's normal business hours are from 8:30 a.m. to 5 p.m. Eastern Time. A protest or contract dispute received after the time period prescribed for filing shall not be considered timely filed. Service shall also be made on the Contracting

Officer (CO) pursuant to §§17.15(e) and 17.27(d).

- (b) Submissions to the ODRA after the initial filing of a protest or contract dispute may be accomplished by any means available in paragraph (a) of this section. Copies of all such submissions shall be served on the opposing party or parties.
- (c) The time limits stated in this part are calculated in business days, which exclude weekends, Federal holidays and other days on which Federal Government offices in Washington, DC are not open. In computing time, the day of the event beginning a period of time shall not be included. If the last day of a period falls on a weekend or a Federal holiday, the first business day following the weekend or holiday shall be considered the last day of the period.
- (d) Electronic Filing—Procedures for electronic filing may be utilized where permitted by Order of the ODRA on a case-by-case basis or pursuant to a Standing Order of the ODRA permitting electronic filing.

§17.9 Protective orders.

- (a) The ODRA may issue protective orders addressing the treatment of protected information, including protected information in electronic form, either at the request of a party or upon its own initiative. Such information may include proprietary, confidential, or source-selection-sensitive material, or other information the release of which could result in a competitive advantage to one or more firms.
- (b) The terms of the ODRA's standard protective order may be altered to suit particular circumstances, by negotiation of the parties, subject to the approval of the ODRA. The protective order establishes procedures for application for access to protected information, identification and safeguarding of that information, and submission of redacted copies of documents omitting protected information.
- (c) After a protective order has been issued, counsel or consultants retained by counsel appearing on behalf of a party may apply for access to the material under the order by submitting an application to the ODRA, with copies furnished simultaneously to all parties. The application shall establish that the

applicant is not involved in competitive decision-making for any firm that could gain a competitive advantage from access to the protected information and that the applicant will diligently protect any protected information received from inadvertent disclosure. Objections to an applicant's admission shall be raised within two (2) days of the application, although the ODRA may consider objections raised after that time for good cause.

(d) Any violation of the terms of a protective order may result in the imposition of sanctions, including but not limited to removal of the violator from the protective order and reporting of the violator to his or her bar association(s), and the taking of other actions as the ODRA deems appropriate. Additional civil or criminal penalties may apply.

Subpart B—Protests

§17.11 Matters not subject to protest.

The following matters may not be protested before the ODRA, except for review of compliance with the AMS:

- (a) FAA purchases from or through, State, local, and tribal governments and public authorities;
- (b) FAA purchases from or through other Federal agencies;
 - (c) Grants:
 - (d) Cooperative agreements;
 - (e) Other transactions.

§ 17.13 Dispute resolution process for protests.

- (a) Protests concerning FAA SIRs, solicitations, or contract awards shall be resolved pursuant to this part.
- (b) Potential protestors should, where possible, attempt to resolve any issues concerning potential protests with the CO. Such attempts are not a prerequisite to filing a protest with the ODRA.
- (c) Offerors or prospective offerors shall file a protest with the ODRA in accordance with §17.15. The protest time limitations set forth in §17.15 will not be extended by attempts to resolve a potential protest with the CO. Other than the time limitations specified in §17.15 for the filing of protests, the ODRA retains the discretion to modify

any timeframes established herein in connection with protests.

- (d) In accordance with §17.17(b), the ODRA shall convene an initial status conference for the purpose of scheduling proceedings in the protest and to encourage the parties to consider using the ODRA's ADR process to attempt to resolve the protest, pursuant to subpart D of this part. It is the Agency's policy to use voluntary ADR to the maximum extent practicable. If the parties elect not to attempt ADR, or if ADR efforts do not completely resolve the protest, the protest will proceed under the ODRA Adjudicative Process set forth in subpart E of this part. Informal ADR techniques may be utilized simultaneously with ongoing adjudication.
- (e) The ODRA Director shall designate DROs, outside neutrals or Special Masters as potential neutrals for the resolution of protests through ADR. The ultimate choice of an ADR neutral is made by the parties participating in the ADR. The ODRA Director also shall, at his or her sole discretion, designate an adjudicating DRO or Special Master for each matter. A person serving as a neutral in an ADR effort in a matter, shall not serve as an adjudicating DRO or Special Master for that matter.
- (f) Multiple protests concerning the same SIR, solicitation, or contract award may be consolidated at the discretion of the ODRA Director, and assigned to a single DRO or Special Master for adjudication.
- (g) Procurement activities, and, where applicable, contractor performance pending resolution of a protest, shall continue during the pendency of a protest, unless there is a compelling reason to suspend all or part of the procurement activities or contractor performance. Pursuant to §§17.15(d) and 17.17(a), the ODRA may impose a temporary suspension and recommend suspension of award or contract performance, in whole or in part, for a compelling reason. A decision to suspend procurement activities or contractor performance is made in writing by the Administrator or the Administrator's delegee upon recommendation of the ODRA.

§17.15 Filing a protest.

- (a) An interested party may initiate a protest by filing with the ODRA in accordance with §17.7(a) within the timeframes set forth in this Section. Protests that are not timely filed shall be dismissed. The timeframes applicable to the filing of protests are as follows:
- (1) Protests based upon alleged SIR or solicitation improprieties that are apparent prior to bid opening or the time set for receipt of initial proposals shall be filed prior to bid opening or the time set for the receipt of initial proposals.
- (2) In procurements where proposals are requested, alleged improprieties that do not exist in the initial solicitation, but which are subsequently incorporated into the solicitation, must be protested not later than the next closing time for receipt of proposals following the incorporation.
- (3) For protests other than those related to alleged solicitation improprieties, the protest must be filed on the later of the following two dates:
- (i) Not later than seven (7) business days after the date the protester knew or should have known of the grounds for the protest; or
- (ii) If the protester has requested a post-award debriefing from the FAA Product Team, not later than five (5) business days after the date on which the Product Team holds that debriefing.
 - (b) Protests shall be filed at:
- (1) ODRA, AGC-70, Federal Aviation Administration, 800 Independence Avenue, SW., Room 323, Washington, DC 20591; Telephone: (202) 267-3290, Fax: (202) 267-3720; or
- (2) Other address as shall be published from time to time in the FEDERAL REGISTER.
- (c) A protest shall be in writing, and set forth:
- (1) The protester's name, address, telephone number, and FAX number;
- (2) The name, address, telephone number, and FAX number of the protester's legal representative, and who shall be duly authorized to represent the protester, to be the point of contact:

- (3) The SIR number or, if available, the contract number and the name of the CO:
- (4) The basis for the protester's status as an interested party;
- (5) The facts supporting the timeliness of the protest;
- (6) Whether the protester requests a protective order, the material to be protected, and attach a redacted copy of that material;
- (7) A detailed statement of both the legal and factual grounds of the protest, and one (1) copy of each relevant document;
- (8) The remedy or remedies sought by the protester, as set forth in §17.23;
- (9) The signature of the legal representative, or another person duly authorized to represent the protester.
- (d) If the protester wishes to request a suspension of the procurement or contract performance, in whole or in part, and believes that a compelling reason(s) exists to suspend the procurement or contract performance because of the protested action, the protester shall, in its initial filing:
- (1) Set forth such compelling reason(s), supply all facts and documents supporting the protester's position; and
 - (2) Demonstrate—
- (i) The protester has alleged a substantial case:
- (ii) The lack of a suspension would be likely to cause irreparable injury;
- (iii) The relative hardships on the parties favor a suspension; and
- (iv) That a suspension is in the public interest.
- (3) Failure of a protester to provide information or documents in support of a requested suspension or failure to address the elements of paragraph (d)(2) of this section may result in the summary rejection of the request for suspension, or a requirement that the protester supplement its request prior to the scheduling of a Product Team response to the request under §17.17(a).
- (e) Concurrent with the filing of a protest with the ODRA, the protester shall serve a copy of the protest on the CO and any other official designated in the SIR for receipt of protests, by means reasonably calculated to be received by the CO on the same day as it is to be received by the ODRA. The protest shall include a signed statement

- from the protester, certifying to the ODRA the manner of service, date, and time when a copy of the protest was served on the CO and other designated official(s).
- (f) Upon receipt of the protest, the CO shall notify the awardee of a challenged contract award in writing of the existence of the protest. The awardee and/or interested parties shall notify the ODRA in writing, of their interest in participating in the protest as intervenors within two (2) business days of receipt of the CO's notification, and shall, in such notice, designate a person as the point of contact for the ODRA.
- (g) The ODRA has discretion to designate the parties who shall participate in the protest as intervenors. In protests of awarded contracts, only the awardee may participate as an intervenor as a matter of right.

§17.17 Initial protest procedures.

- (a) If, as part of its initial protest filing, the protester requests a suspension of procurement activities or contractor performance in whole or in part, in accordance with §17.15(d), the Product Team shall submit a response to the request to the ODRA by no later than the close of business on the date of the initial scheduling conference or on such other date as is established by the ODRA. Copies of the response shall be furnished to the protester and any intervenor(s) so as to be received within the same timeframe. The protester and any intervenor(s) shall have the opportunity of providing additional comments on the response within two (2) business days of receiving it. Based on its review of such submissions, the ODRA, in its discretion, may-
 - (1) Decline the suspension request; or
- (2) Recommend such suspension to the Administrator or the Administrator's designee. The ODRA also may impose a temporary suspension of no more than ten (10) business days, where it is recommending that the Administrator impose a suspension.
- (b) Within five (5) business days of the filing of a protest, or as soon thereafter as practicable, the ODRA shall convene an initial status conference for purposes of:

- (1) Reviewing the ODRA's ADR and adjudication procedures and establishing a preliminary schedule;
- (2) Identifying legal or other preliminary or potentially dispositive issues and answering the parties' questions regarding the ODRA process;
- (3) Dealing with issues related to protected information and the issuance of any needed protective order:
- (4) Encouraging the parties to consider using ADR:
- (5) Appointing a DRO as a potential ADR neutral to assist the parties in considering ADR options and developing an ADR agreement; and
- (6) For any other reason deemed appropriate by the DRO or by the ODRA.
- (c) The Product Team and protester will have five (5) business days from the date of the initial status conference to decide whether they will attempt to use an ADR process in the case. With the agreement of the ODRA, ADR may be used concurrently with the adjudication of a protest. See §17.37(e).
- (d) If the Product Team and protester elect to use ADR proceedings to resolve the protest, they will agree upon the neutral to conduct the ADR proceedings (either an ODRA DRO or a compensated neutral of their own choosing) pursuant to §17.37, and shall execute and file with the ODRA a written ADR agreement. Agreement of any intervenor(s) to the use of ADR or the resolution of a dispute through ADR shall not be required.
- (e) If the Product Team or protester indicate that ADR proceedings will not be used, or if ADR is not successful in resolving the entire protest, the ODRA Director upon being informed of the situation, will schedule an adjudication of the protest.

§ 17.19 Motions practice and dismissal or summary decision of protests.

(a) Separate motions generally are discouraged in ODRA bid protests. Counsel and parties are encouraged to incorporate any such motions in their respective agency responses or comments. Parties and counsel are encouraged to attempt to resolve typical motions issues through the ODRA ADR process. The ODRA may rule on any non-dispositive motion, where appro-

- priate and necessary, after providing an opportunity for briefing on the motion by all affected parties. Unjustifiable, inappropriate use of motions may result in the imposition of sanctions. Where appropriate, a party may request by dispositive motion to the ODRA, or the ODRA may recommend or order, that:
- (1) The protest, or any count or portion of a protest, be dismissed for lack of jurisdiction, timeliness, or standing to pursue the protest;
- (2) The protest, or any count or portion of a protest, be dismissed, if frivolous or without basis in fact or law, or for failure to state a claim upon which relief may be had;
- (3) A summary decision be issued with respect to the protest, or any count or portion of a protest, if there are no material facts in dispute and a party is entitled to summary decision as a matter of law.
- (b) In connection with consideration of possible dismissal or summary decision, the ODRA shall consider any material facts in dispute, in a light most favorable to the party against whom the dismissal or summary decision would operate and draw all factual inferences in favor of the non-moving party.
- (c) Either upon motion by a party or on its own initiative, the ODRA may, at any time, exercise its discretion to:
- (1) Recommend to the Administrator dismissal or the issuance of a summary decision with respect to the entire protest:
- (2) Dismiss the entire protest or issue a summary decision with respect to the entire protest, if delegated that authority by the Administrator; or
- (3) Dismiss or issue a summary decision with respect to any count or portion of a protest.
- (d) A dismissal or summary decision regarding the entire protest by either the Administrator, or the ODRA by delegation, shall be construed as a final agency order. A dismissal or summary decision that does not resolve all counts or portions of a protest shall not constitute a final agency order, unless and until such dismissal or decision is incorporated or otherwise

adopted in a decision by the Administrator (or the ODRA, by delegation) regarding the entire protest.

(e) Prior to recommending or entering either a dismissal or a summary decision, either in whole or in part, the ODRA shall afford all parties against whom the dismissal or summary decision is to be entered the opportunity to respond to the proposed dismissal or summary decision.

§ 17.21 Adjudicative Process for protests.

- (a) Other than for the resolution of preliminary or dispositive matters, the Adjudicative Process for protests will be commenced by the ODRA Director pursuant to §17.17(e).
- (b) The Director of the ODRA shall appoint a DRO or a Special Master to conduct the adjudication proceedings, develop the administrative record, and prepare findings and recommendations for review of the ODRA Director.
- (c) The DRO or Special Master may conduct such proceedings and prepare procedural orders for the proceedings as deemed appropriate; and may require additional submissions from the parties.
- (d) The Product Team response to the protest will be due to be filed and served ten (10) business days from the commencement of the ODRA Adjudication process. The Product Team response shall consist of a written chronological, supported statement of proposed facts, and a written presentation of applicable legal or other defenses. The Product Team response shall cite to and be accompanied by all relevant documents, which shall be chronologically indexed, individually tabbed, and certified as authentic and complete. A copy of the response shall be furnished so as to be received by the protester and any intervenor(s) on the same date it is filed with the ODRA. In all cases, the Product Team shall indicate the method of service used.
- (e) Comments of the protester and the intervenor on the Product Team response will be due to be filed and served five (5) business days after their receipt of the response. Copies of such comments shall be provided to the other participating parties by the same means and on the same date as they

- are furnished to the ODRA. Comments may include any supplemental relevant documents.
- (f) The ODRA may alter the schedule for filing of the Product Team response and the comments for good cause or to accommodate the circumstances of a particular protest.
- (g) The DRO or Special Master may convene the parties and/or their representatives, as needed for the Adjudicative Process.
- (h) If, in the sole judgment of the DRO or Special Master, the parties have presented written material sufficient to allow the protest to be decided on the record presented, the DRO or Special Master shall have the discretion to decide the protest on that basis.
- (i) The parties may engage in limited, focused discovery with one another and, if justified, with non-parties, so as to obtain information relevant to the allegations of the protest.
- (1) The DRO or Special Master shall manage the discovery process, including limiting its length and availability, and shall establish schedules and deadlines for discovery, which are consistent with timeframes established in this part and with the FAA policy of providing fair and expeditious dispute resolution.
- (2) The DRO or Special Master may also direct the parties to exchange, in an expedited manner, relevant, non-privileged documents.
- (3) Where justified, the DRO or Special Master may direct the taking of deposition testimony, however, the FAA dispute resolution process does not contemplate extensive discovery.
- (4) The use of interrogatories and requests for admission is not permitted in ODRA bid protests.
- (5) Where parties cannot voluntarily reach agreement on a discovery-related issue, they may timely seek assistance from an ODRA ADR neutral or may file an appropriate motion with the ODRA. Parties may request a subpoena.
- (6) Discovery requests and responses are not part of the record and will not be filed with the ODRA, except in connection with a motion or other permissible filing.
- (7) Unless timely objection is made, documents properly filed with the

ODRA will be deemed admitted into the administrative record.

- (j) Hearings are not typically held in bid protests. The DRO or Special Master may conduct hearings, and may limit the hearings to the testimony of specific witnesses and/or presentations regarding specific issues. The DRO or Special Master shall control the nature and conduct of all hearings, including the sequence and extent of any testimony. Hearings will be conducted:
- (1) Where the DRO or Special Master determines that there are complex factual issues in dispute that cannot adequately or efficiently be developed solely by means of written presentations and/or that resolution of the controversy will be dependent on his/her assessment of the credibility of statements provided by individuals with first-hand knowledge of the facts; or
- (2) Upon request of any party to the protest, unless the DRO or Special Master finds specifically that a hearing is unnecessary and that no party will be prejudiced by limiting the record in the adjudication to the parties' written submissions. All witnesses at any such hearing shall be subject to cross-examination by the opposing party and to questioning by the DRO or Special Master.
- (k) The Director of the ODRA may review the status of any protest in the Adjudicative Process with the DRO or Special Master.
- (1) After the closing of the administrative record, the DRO or Special Master will prepare and submit findings and recommendations to the ODRA that shall contain the following:
 - (1) Findings of fact;
- (2) Application of the principles of the AMS, and any applicable law or authority to the findings of fact;
- (3) A recommendation for a final FAA order; and
- (4) If appropriate, suggestions for future FAA action.
- (m) In preparing findings and recommendations in protests, the DRO or Special Master, using the preponderance of the evidence standard, shall consider whether the Product Team actions in question were consistent with the requirements of the AMS, had a rational basis, and whether the Product

Team decision was arbitrary, capricious or an abuse of discretion. Notwithstanding the above, allegations that government officials acted with bias or in bad faith must be established by clear and convincing evidence.

- (n) The DRO or Special Master has broad discretion to recommend a remedy that is consistent with §17.23.
- (o) A DRO or Special Master shall submit findings and recommendations only to the Director of the ODRA or the Director's designee. The findings and recommendations will be released to the parties and to the public upon issuance of the final FAA order in the case. If an ODRA protective order was issued in connection with the protest, or if a protest involves proprietary or competition-sensitive information, a redacted version of the findings and recommendations, omitting any protected information, shall be prepared wherever possible and released to the public, as soon as is practicable, along with a copy of the final FAA order. Only persons admitted by the ODRA under the protective order and Government personnel shall be provided copies of the unredacted findings and recommendations that contain proprietary or competition-sensitive information.
- (p) Other than communications regarding purely procedural matters or ADR, there shall be no substantive *ex parte* communication between ODRA personnel and any principal or representative of a party concerning a pending or potentially pending matter. A potential or serving ADR neutral may communicate on an *ex parte* basis to establish or conduct the ADR.

§ 17.23 Protest remedies.

- (a) The ODRA has broad discretion to recommend and impose protest remedies that are consistent with the AMS and applicable law. Such remedies may include, but are not limited to one or a combination of, the following:
 - (1) Amend the SIR;
- (2) Refrain from exercising options under the contract:
 - (3) Issue a new SIR;
- (4) Require a recompetition or revaluation;
- (5) Terminate an existing contract for the FAA's convenience;

- (6) Direct an award to the protester;
- (7) Award bid and proposal costs; or
- (8) Any other remedy consistent with the AMS that is appropriate under the circumstances.
- (b) In determining the appropriate recommendation, the ODRA may consider the circumstances surrounding the procurement or proposed procurement including, but not limited to: the nature of the procurement deficiency; the degree of prejudice to other parties or to the integrity of the acquisition system; the good faith of the parties; the extent of performance completed; the feasibility of any proposed remedy; the urgency of the procurement; the cost and impact of the recommended remedy; and the impact on the Agency's mission.
- (c) Attorney's fees of a prevailing protester are allowable to the extent permitted by the Equal Access to Justice Act, 5 U.S.C. 504(a)(1) (EAJA) and 14 CFR part 14.

Subpart C—Contract Disputes

§ 17.25 Dispute resolution process for contract disputes.

- (a) All contract disputes arising under contracts subject to the AMS shall be resolved under this subpart.
- (b) Contract disputes shall be filed with the ODRA pursuant to §17.27.
- (c) The ODRA has broad discretion to recommend remedies for a contract dispute that are consistent with the AMS and applicable law, including such equitable remedies or other remedies as it deems appropriate.

§17.27 Filing a contract dispute.

- (a) Contract disputes must be in writing and should contain:
- (1) The contractor's name, address, telephone and Fax numbers and the name, address, telephone and Fax numbers of the contractor's legal representative(s) (if any) for the contract dispute:
- (2) The contract number and the name of the Contracting Officer;
- (3) A detailed chronological statement of the facts and of the legal grounds underlying the contract dispute, broken down by individual claim item, citing to relevant contract provi-

- sions and attaching copies of the contract and other relevant documents;
- (4) Information establishing the ODRA's jurisdiction and the timeliness of the contract dispute;
- (5) A request for a specific remedy, and the amount, if known, of any monetary remedy requested, together with pertinent cost information and documentation (e.g., invoices and cancelled checks). Supporting documentation should be broken down by individual claim item and summarized; and
- (6) The signature of a duly authorized representative of the initiating party.
- (b) Contract disputes shall be filed at the following address: ODRA, AGC-70, Federal Aviation Administration, 800 Independence Avenue, SW., Room 323, Washington, DC 20591; Telephone: (202) 267-3290, Fax: (202) 267-3720.
- (c) A contract dispute against the FAA shall be filed with the ODRA within two (2) years of the accrual of the contract claim involved. A contract dispute by the FAA against a contractor (excluding contract disputes alleging warranty issues, fraud or latent defects) likewise shall be filed within two (2) years of the accrual of the contract claim. If an underlying contract entered into prior to the effective date of this part provides for time limitations for filing of contract disputes with the ODRA, which differ from the aforesaid two (2) year period, the limitation periods in the contract shall control over the limitation period of this section. In no event will either party be permitted to file with the ODRA a contract dispute seeking an equitable adjustment or other damages after the contractor has accepted final contract payment, with the exception of FAA contract disputes related to warranty issues, gross mistakes amounting to fraud or latent defects. FAA contract disputes against the contractor based on warranty issues must be filed within the time specified under applicable contract warranty provisions. Any FAA contract disputes against the contractor based on gross mistakes amounting to fraud or latent defects shall be filed with the ODRA within two (2) years of the date on which the FAA knew or should have known of the presence of the fraud or latent defect.

- (d) A party shall serve a copy of the contract dispute upon the other party, by means reasonably calculated to be received on the same day as the filing is received by the ODRA.
- (e) With the exception of the time limitations established herein for the filing of contract disputes, the ODRA retains the discretion to modify any timeframe established herein in connection with contract disputes.

§17.29 Informal resolution period.

- (a) The ODRA process for contract disputes includes an informal resolution period of twenty (20) business days from the date of filing in order for the parties to attempt to informally resolve the contract dispute either through direct negotiation or with the assistance of the ODRA. The CO, with the advice of FAA legal counsel, has full discretion to settle contract disputes, except where the matter involves fraud.
- (b) During the informal resolution period, if the parties request it, the ODRA will appoint a DRO for ADR who will discuss ADR options with the parties, offer his or her services as a potential neutral, and assist the parties to enter into an agreement for a formal ADR process. A person serving as a neutral in an ADR effort in a matter shall not serve as an adjudicating DRO or Special Master for that matter.
- (c) The informal resolution period may be extended at the request of the parties for good cause.
- (d) If the matter has not been resolved informally, the parties shall file joint or separate statements with the ODRA no later than twenty (20) business days after the filing of the contract dispute. The ODRA may extend this time, pursuant to §17.27(e). The statement(s) shall include either:
- (1) A joint request for ADR, or an executed ADR agreement, pursuant to §17.37(d), specifying which ADR techniques will be employed; or
- (2) Written explanation(s) as to why ADR proceedings will not be used and why the Adjudicative Process will be needed.
- (e) If the contract dispute is not completely resolved during the informal resolution period, the ODRA's Adjudicative Process will commence unless

- the parties have reached an agreement to attempt a formal ADR effort. As part of such an ADR agreement the parties, with the concurrence of the ODRA, may agree to defer commencement of the adjudication process pending completion of the ADR or that the ADR and adjudication process will run concurrently. If a formal ADR is attempted but does not completely resolve the contract dispute, the Adjudicative Process will commence.
- (f) The ODRA shall hold a status conference with the parties within ten (10) business days, or as soon thereafter as is practicable, of the ODRA's receipt of a written notification that ADR proceedings will not be used, or have not fully resolved the Contract Dispute. The purpose of the status conference will be to commence the Adjudicative Process and establish the schedule for adjudication.
- (g) The submission of a statement which indicates that ADR will not be utilized will not in any way preclude the parties from engaging in non-binding ADR techniques during the Adjudicative Process, pursuant to subpart D of this part.

§ 17.31 Dismissal or summary decision of contract disputes.

- (a) Any party may request by motion, or the ODRA on its own initiative may recommend or direct, that a contract dispute be dismissed, or that a count or portion thereof be stricken, if:
 - (1) It was not timely filed;
- (2) It was filed by a subcontractor or other person or entity lacking standing;
- (3) It fails to state a matter upon which relief may be had; or
- (4) It involves a matter not subject to the jurisdiction of the ODRA.
- (b) Any party may request by motion, or the ODRA on its own initiative may recommend or direct, that a summary decision be issued with respect to a contract dispute, or any count or portion thereof if there are no material facts in dispute and a party is entitled to a summary decision as a matter of law.
- (c) In connection with any potential dismissal of a contract dispute, or summary decision, the ODRA will consider any material facts in dispute in a light

most favorable to the party against whom the dismissal or summary decision would be entered, and draw all factual inferences in favor of that party.

- (d) At any time, whether pursuant to a motion or on its own initiative and at its discretion, the ODRA may:
- (1) Dismiss or strike a count or portion of a contract dispute or enter a partial summary decision;
- (2) Recommend to the Administrator that the entire contract dispute be dismissed or that a summary decision be entered; or
- (3) With a delegation from the Administrator, dismiss the entire contract dispute or enter a summary decision with respect to the entire contract dispute.
- (e) An order of dismissal of the entire contract dispute or summary decision with respect to the entire contract dispute, issued either by the Administrator or by the ODRA, on the grounds set forth in this section, shall constitute a final agency order. An ODRA order dismissing or striking a count or portion of a contract dispute or entering a partial summary judgment shall not constitute a final agency order, unless and until such ODRA order is incorporated or otherwise adopted in a final agency decision of the Administrator or the Administrator's delegee regarding the remainder of the dispute.
- (f) Prior to recommending or entering either a dismissal or a summary decision, either in whole or in part, the ODRA shall afford all parties against whom the dismissal or summary decision would be entered the opportunity to respond to a proposed dismissal or summary decision.

§17.33 Adjudicative Process for contract disputes.

- (a) The Adjudicative Process for contract disputes will be commenced by the ODRA Director upon being notified by the ADR neutral or by any party that either—
- (1) The parties will not be attempting ADR; or
- (2) The parties have not settled all of the dispute issues via ADR, and it is unlikely that they can do so within the time period allotted and/or any reasonable extension.

- (b) In cases initiated by a contractor against the FAA, within twenty (20) business days of the commencement of the Adjudicative Process or as scheduled by the ODRA, the Product Team shall prepare and submit to the ODRA, with a copy to the contractor, a chronologically arranged and indexed substantive response, containing a legal and factual position regarding the dispute and all documents relevant to the facts and issues in dispute. The contractor will be entitled, at a specified time, to supplement the record with additional documents.
- (c) In cases initiated by the FAA against a contractor, within twenty (20) business days of the commencement of the Adjudicative Process or as scheduled by the ODRA, the contractor shall prepare and submit to the ODRA, with a copy to the Product Team counsel, a chronologically arranged and indexed substantive response, containing a legal and factual position regarding the dispute and all documents relevant to the facts and issues in dispute. The Product Team will be entitled, at a specified time, to supplement the record with additional documents.
- (d) Unless timely objection is made, documents properly filed with the ODRA will be deemed admitted into the administrative record. Discovery requests and responses are not part of the record and will not be filed with the ODRA, except in connection with a motion or other permissible filing. Designated, relevant portions of such documents may be filed, with the permission of the ODRA.
- (e) The Director of the ODRA shall assign a DRO or a Special Master to conduct adjudicatory proceedings, develop the administrative adjudication record and prepare findings and recommendations for the review of the ODRA Director or the Director's designee.
- (f) The DRO or Special Master may conduct a status conference(s) as necessary and issue such orders or decisions as are necessary to promote the efficient resolution of the contract disnute
- (g) At any such status conference, or as necessary during the Adjudicative Process, the DRO or Special Master will:

- (1) Determine the appropriate amount of discovery required;
- (2) Review the need for a protective order, and if one is needed, prepare a protective order pursuant to §17.9;
- (3) Determine whether any issue can be stricken; and
- (4) Prepare necessary procedural orders for the proceedings.
- (h) Unless otherwise provided by the DRO or Special Master, or by agreement of the parties with the concurrence of the DRO or Special Master, responses to written discovery shall be due within thirty (30) business days from the date received.
- (i) At a time or at times determined by the DRO or Special Master, and in advance of the decision of the case, the parties shall make individual final submissions to the ODRA and to the DRO or Special Master, which submissions shall include the following:
 - (1) A statement of the issues;
- (2) A proposed statement of undisputed facts related to each issue together with citations to the administrative record or other supporting materials:
- (3) Separate statements of disputed facts related to each issue, with appropriate citations to documents in the Dispute File, to pages of transcripts of any hearing or deposition, or to any affidavit or exhibit which a party may wish to submit with its statement;
- (4) Separate legal analyses in support of the parties' respective positions on disputed issues.
- (j) Each party shall serve a copy of its final submission on the other party by means reasonably calculated so that the other party receives such submissions on the same day it is received by the ODRA.
- (k) The DRO or Special Master may decide the contract dispute on the basis of the administrative record and the submissions referenced in this section, or may, in the DRO or Special Master's discretion, direct the parties to make additional presentations in writing. The DRO or Special Master may conduct hearings, and may limit the hearings to the testimony of specific witnesses and/or presentations regarding specific issues. The DRO or Special Master shall control the nature and conduct of all hearings, including

- the sequence and extent of any testimony. Evidentiary hearings on the record shall be conducted by the ODRA:
- (1) Where the DRO or Special Master determines that there are complex factual issues in dispute that cannot adequately or efficiently be developed solely by means of written presentations and/or that resolution of the controversy will be dependent on his/her assessment of the credibility of statements provided by individuals with first-hand knowledge of the facts; or
- (2) Upon request of any party to the contract dispute, unless the DRO or Special Master finds specifically that a hearing is unnecessary and that no party will be prejudiced by limiting the record in the adjudication to the parties' written submissions. All witnesses at any such hearing shall be subject to cross-examination by the opposing party and to questioning by the DRO or Special Master.
- (1) The DRO or Special Master shall prepare findings and recommendations, which will contain findings of fact, application of the principles of the AMS and other law or authority applicable to the findings of fact, and a recommendation for a final FAA order.
- (m) The DRO or Special Master shall conduct a de novo review using the preponderance of the evidence standard, unless a different standard is prescribed for a particular issue. Notwithstanding the above, allegations that government officials acted with bias or in bad faith must be established by clear and convincing evidence.
- (n) The Director of the ODRA may review the status of any contract dispute in the Adjudicative Process with the DRO or Special Master.
- (0) A DRO or Special Master shall submit findings and recommendations to the Director of the ODRA or the Director's designee. The findings and recommendations will be released to the parties and to the public, upon issuance of the final FAA order in the case. Should an ODRA protective order be issued in connection with the contract dispute, or should the matter involve proprietary or competition-sensitive information, a redacted version of the findings and recommendations

omitting any protected information, shall be prepared wherever possible and released to the public, as soon as is practicable, along with a copy of the final FAA order. Only persons admitted by the ODRA under the protective order and Government personnel shall be provided copies of the unredacted findings and recommendations.

- (p) Attorneys' fees of a qualified prevailing contractor are allowable to the extent permitted by the EAJA, 5 U.S.C. 504(a)(1). See 14 CFR part 14.
- (q) Other than communications regarding purely procedural matters or ADR, there shall be no substantive *ex parte* communication between ODRA personnel and any principal or representative of a party concerning a pending or potentially pending matter. A potential or serving ADR neutral may communicate on an ex parte basis to establish or conduct the ADR.

Subpart D—Alternative Dispute Resolution

§ 17.35 Use of alternative dispute resolution.

- (a) By statutory mandate, it is the policy of the FAA to use voluntary ADR to the maximum extent practicable to resolve matters pending at the ODRA. The ODRA therefore uses voluntary ADR as its primary means of resolving all factual, legal, and procedural controversies.
- (b) The parties are encouraged to make a good faith effort to explore ADR possibilities in all cases and to employ ADR in every appropriate case. The ODRA uses ADR techniques such as mediation, neutral evaluation, binding arbitration or variations of these techniques as agreed by the parties and approved by the ODRA. At the beginning of each case, the ODRA assigns a DRO as a potential neutral to explore ADR options with the parties and to convene an ADR process. See §17.35(b).
- (c) The ODRA Adjudicative Process will be used where the parties cannot achieve agreement on the use of ADR; where ADR has been employed but has not resolved all pending issues in dispute; or where the ODRA concludes that ADR will not provide an expeditious means of resolving a particular dispute. Even where the Adjudicative

Process is to be used, the ODRA, with the parties' consent, may employ informal ADR techniques concurrently with the adjudication.

§17.37 Election of alternative dispute resolution process.

- (a) The ODRA will make its personnel available to serve as Neutrals in ADR proceedings and, upon request by the parties, will attempt to make qualified non-FAA personnel available to serve as Neutrals through neutralsharing programs and other similar arrangements. The parties may elect to employ a mutually acceptable compensated neutral at their expense.
- (b) The parties using an ADR process to resolve a protest shall submit an executed ADR agreement containing the information outlined in paragraph (d) of this section to the ODRA pursuant to §17.17(c). The ODRA may extend this time for good cause.
- (c) The parties using an ADR process to resolve a contract dispute shall submit an executed ADR agreement containing the information outlined in paragraph (d) of this section to the ODRA pursuant to §17.29.
- (d) The parties to a protest or contract dispute who elect to use ADR must submit to the ODRA an ADR agreement setting forth:
- (1) The agreed ADR procedures to be used: and
- (2) The name of the neutral. If a compensated neutral is to be used, the agreement must address how the cost of the neutral's services will be reimbursed.
- (e) Non-binding ADR techniques are not mutually exclusive, and may be used in combination if the parties agree that a combination is most appropriate to the dispute. The techniques to be employed must be determined in advance by the parties and shall be expressly described in their ADR agreement. The agreement may provide for the use of any fair and reasonable ADR technique that is designed to achieve a prompt resolution of the matter. An ADR agreement for non-binding ADR shall provide for a termination of ADR proceedings and the commencement of adjudication under the Adjudicative Process, upon

the election of any party. Notwithstanding such termination, the parties may still engage with the ODRA in ADR techniques (neutral evaluation and/or informal mediation) concurrently with adjudication.

- (f) Binding arbitration is available through the ODRA, subject to the provisions of applicable law and the ODRA Binding Arbitration Guidance dated October 2001 as developed in consultation with the Department of Justice.
- (g) The parties may, where appropriate in a given case, submit to the ODRA a negotiated protective order for use in ADR in accordance with the requirements of §17.9.

$\S 17.39$ Confidentiality of ADR.

- (a) The provisions of the Administrative Dispute Resolution Act of 1996, 5 U.S.C. 571, et seq., shall apply to ODRA ADR proceedings.
- (b) The ODRA looks to the principles of Rule 408 of the Federal Rules of Evidence in deciding admissibility issues related to ADR communications.
- (c) ADR communications are not part of the administrative record unless otherwise agreed by the parties.

Subpart E—Finality and Review

§17.41 Final orders.

All final FAA orders regarding protests or contract disputes under this part are to be issued by the FAA Administrator or by a delegee of the Administrator.

§17.43 Judicial review.

- (a) A protester or contractor may seek review of a final FAA order, pursuant to 49 U.S.C. 46110, only after the administrative remedies of this part have been exhausted.
- (b) A copy of the petition for review shall be filed with the ODRA and the FAA Chief Counsel on the date that the petition for review is filed with the appropriate circuit court of appeals.

§17.45 Conforming amendments.

The FAA shall amend pertinent provisions of the AMS, standard contract forms and clauses, and any guidance to contracting officials, so as to conform to the provisions of this part.

§17.47 Reconsideration.

The ODRA will not entertain requests for reconsideration as a routine matter, or where such requests evidence mere disagreement with a decision or restatements of previous arguments. A party seeking reconsideration must demonstrate either clear errors of fact or law in the underlying decision or previously unavailable evidence that warrants reversal or modification of the decision. In order to be considered, requests for reconsideration must be filed within ten (10) business days of the date of issuance of the public version of the subject decision or order.

Subpart F—Other Matters

§ 17.49 Sanctions.

If any party or its representative fails to comply with an Order or Directive of the ODRA, the ODRA may enter such orders and take such other actions as it deems necessary and in the interest of justice.

§ 17.51 Decorum and professional conduct.

Legal representatives are expected to conduct themselves at all times in a civil and respectful manner appropriate to an administrative forum. Additionally, counsel are expected to conduct themselves at all times in a professional manner and in accordance with all applicable rules of professional conduct.

§ 17.53 Orders and subpoenas for testimony and document production.

- (a) Parties are encouraged to seek cooperative and voluntary production of documents and witnesses prior to requesting a subpoena or an order under this section.
- (b) Upon request by a party, or on his or her own initiative, a DRO or Special Master may, for good cause shown, order a person to give testimony by deposition and to produce records. Section 46104(c) of Title 49 of the United States Code governs the conduct of depositions or document production.
- (c) Upon request by a party, or on his or her own initiative, a DRO or Special Master may, for good cause shown, subpoena witnesses or records related to a

hearing from any place in the United States to the designated place of a hearing.

- (d) A subpoena or order under this section may be served by a United States marshal or deputy marshal, or by any other person who is not a party and not less than 18 years of age. Service upon a person named therein shall be made by personally delivering a copy to that person and tendering the fees for one day's attendance and the mileage provided by 28 U.S.C. 1821 or other applicable law; however, where the subpoena is issued on behalf of the Product Team, money payments need not be tendered in advance of attendance. The person serving the subpoena or order shall file a declaration of service with the ODRA, executed in the form required by 28 U.S.C. 1746. The declaration of service shall be filed promptly with the ODRA, and before the date on which the person served must respond to the subpoena or order.
- (e) Upon written motion by the person subpoenaed or ordered under this section, or by a party, made within ten (10) business days after service, but in any event not later than the time specified in the subpoena or order for compliance, the DRO may—
- (1) Rescind or modify the subpoena or order if it is unreasonable and oppressive or for other good cause shown, or
- (2) Require the party on whose behalf the subpoena or order was issued to advance the reasonable cost of producing documentary evidence. Where circumstances require, the DRO may act upon such a motion at any time after a copy has been served upon all parties.
- (f) The party that requests the DRO to issue a subpoena or order under this section shall be responsible for the payment of fees and mileage, as required by 49 U.S.C. 46104(d), for witnesses, officers who serve the order, and the officer before whom a deposition is taken.
- (g) Subpoenas and orders issued under this section may be enforced in a judicial proceeding under 49 U.S.C. 46104(b).

§17.55 Standing orders of the ODRA Director.

The Director may issue such Standing Orders as necessary for the orderly conduct of business before the ODRA.

Subpart G—Pre-Disputes

§17.57 Dispute resolution process for Pre-disputes.

- (a) All potential disputes arising under contracts or solicitations with the FAA may be resolved with the consent of the parties to the dispute under this subpart.
- (b) Pre-disputes shall be filed with the ODRA pursuant to §17.59.
- (c) The time limitations for the filing of Protests and Contract Disputes established in §§17.15(a) and 17.27(c) will not be extended by efforts to resolve the dispute under this subpart.

§17.59 Filing a Pre-dispute.

- (a) A Pre-dispute must be in writing, affirmatively state that it is a Pre-dispute pursuant to this subpart, and shall contain:
- (1) The party's name, address, telephone and Fax numbers and the name, address, telephone and Fax numbers of the contractor's legal representative(s) (if any);
- (2) The contract or solicitation number and the name of the Contracting Officer:
- (3) A chronological statement of the facts and of the legal grounds for the party's positions regarding the dispute citing to relevant contract or solicitation provisions and documents and attaching copies of those provisions and documents; and
- (4) The signature of a duly authorized legal representative of the initiating party.
- (b) Pre-disputes shall be filed at the following address: ODRA, AGC-70, Federal Aviation Administration, 800 Independence Avenue, SW., Room 323, Washington, DC 20591; Telephone: (202) 267-3290, Fax: (202) 267-3720.
- (c) Upon the filing of a Pre-dispute with the ODRA, the ODRA will contact the opposing party to offer its services pursuant to §17.57. If the opposing party agrees, the ODRA will provide Pre-dispute services. If the opposing party does not agree, the ODRA Pre-dispute file will be closed and no service will be provided.

§ 17.61 Use of alternative dispute resolution.

- (a) Only non-binding, voluntary ADR will be used to attempt to resolve a Pre-dispute pursuant to §17.37.
- (b) ADR conducted under this subpart is subject to the confidentiality requirements of §17.39.

APPENDIX A TO PART 17—ALTERNATIVE DISPUTE RESOLUTION (ADR)

- A. The FAA dispute resolution procedures encourage the parties to protests and contract disputes to use ADR as the primary means to resolve protests and contract disputes, pursuant to the Administrative Dispute Resolution Act of 1996, Public Law 104–320, 5 U.S.C. 570–579, and Department of Transportation and FAA policies to utilize ADR to the maximum extent practicable. Under the procedures presented in this part, the ODRA encourages parties to consider ADR techniques such as case evaluation, mediation, or arbitration.
- B. ADR encompasses a number of processes and techniques for resolving protests or con-

tract disputes. The most commonly used types include:

- (1) Mediation. The neutral or compensated neutral ascertains the needs and interests of both parties and facilitates discussions between or among the parties and an amicable resolution of their differences, seeking approaches to bridge the gaps between the parties" respective positions. The neutral or compensated neutral can meet with the parties separately, conduct joint meetings with the parties" representatives, or employ both methods in appropriate cases.
- (2) Neutral Evaluation. At any stage during the ADR process, as the parties may agree, the neutral or compensated neutral will provide a candid assessment and opinion of the strengths and weaknesses of the parties" positions as to the facts and law, so as to facilitate further discussion and resolution.
- (3) Binding Arbitration. The ODRA, after consultation with the United States Department of Justice in accordance with the provisions of the Administrative Disputes Resolution Act offers true binding arbitration in cases within its jurisdiction. The ODRA's Guidance for the Use of Binding Arbitration may be found on its website at: http://www.faa.gov/go/odra.